# THE SYNTHESIS AND BIOLOGICAL EVALUATION OF NOVEL ANTICANCER AGENTS

A Thesis submitted in part fulfilment of the requirements of the degree of Doctor of Philosophy

Vikki Clark Pearson

Department of Chemistry University of Glasgow Glasgow G12 8QQ

November 1999

© 1999 Vikki C. Pearson

ProQuest Number: 13834100

All rights reserved

INFORMATION TO ALL USERS The quality of this reproduction is dependent upon the quality of the copy submitted.

In the unlikely event that the author did not send a complete manuscript and there are missing pages, these will be noted. Also, if material had to be removed, a note will indicate the deletion.



ProQuest 13834100

Published by ProQuest LLC (2019). Copyright of the Dissertation is held by the Author.

All rights reserved. This work is protected against unauthorized copying under Title 17, United States Code Microform Edition © ProQuest LLC.

> ProQuest LLC. 789 East Eisenhower Parkway P.O. Box 1346 Ann Arbor, MI 48106 – 1346

i.

11772 (com )

Dedicated to my mum and dad



# ACKNOWLEDGEMENTS

First of all I must express my gratitude to the University of Glasgow for funding my PhD studies with a University Scholarship and to Professor David Robins for his supervision throughout the three year period in addition to his meticulously thorough proof reading of my thesis.

I am most appreciative to our biological collaborators - Dr Lloyd Kelland of The Institute of Cancer Research, Surrey and Dr David Gillespie of the Beatson Institute for Cancer Research, Glasgow who provided me with biological test results. Dr Robert Campbell very kindly helped me interpret some of this data and to him I am very grateful. Further thanks also go to Dr Neil Lant for his guidance at the onset of my work and to Robin Jefferson towards the end for his computer expertise in the presentation of this thesis.

Drs Hill and Hartley have been most encouraging and helpful throughout my time here and I wish to extend my thanks to them. The technical services provided within the chemistry department were superb with Tony Ritchie, Victoria Yates, Jim Gall and Kim Wilson all deserving of mention. In particular I must thank Isabel Freer for her efficient running of the Henderson Laboratory and the diligent manner in which she performed my enzyme assays.

My sojourn at Glasgow University has given me the great fortune of meeting and establishing many new friends of all different nationalities. There are just too many to mention but I would particularly like to mention Russell and Graeme of whom fond memories of our time together will always remain.

Outwith the world of Chemistry and the University of Glasgow, Heather, Lee and Nicola can be found. To this trio of very special friends I cannot say thank you enough for standing by me and providing understanding in some of the more black times as well ensuring the highs were very memorable!

And finally, my biggest thank you of all is to my family: mum, dad, Claire and John who have given me their unconditional love and support from the very beginning, so much more than I could have ever expected. Thank you.

# ABBREVIATIONS

ADP	Adenosine diphosphate
ATP	Adenosine triphosphate
BMN	Benzylidenemalononitrile
bp	Boiling point
br.	Broad (NMR spectroscopy)
cat.	Catalyst
CI	Chemical Ionisation
cmpd	Compound
CNS	Central nervous system
d	Day(s)
d	Doublet (NMR spectroscopy)
dec.	Decomposed (mp)
DEPT	Distortionless Enhancement by Polarisation Transfer
DMF	N,N-dimethylformamide
DNA	Deoxyribonucleic acid
El	Electron Impact
ERK	Extracellular signal-regulated kinase
GI50	Concentration required to reduce growth by 50%
GST	Glutathione S-transferase
h	Hour(s)
IARC	International Agency for Research on Cancer
IR	Infrared
JNK	c-Jun N-terminal kinase
lit.	Literature value
LSPD	Long range selective proton decoupling
m	Multiplet (NMR spectroscopy)
m	Medium (IR spectroscopy)
MAP	Mitogen-activated protein
MAPK	Mitogen-activated protein kinase
MEK	MAP/ERK kinase
mp	Melting point
ms	Mass spectrometry
NMR	Nuclear magnetic resonance
PKC	Protein kinase C
q	Quartet (NMR spectroscopy)

quat.	Quaternary (NMR spectroscopy)
Rf	Retention factor
RNA	Ribonucleic acid
rt	Room temperature
S	Strong (IR spectroscopy)
S	Singlet (NMR spectroscopy)
SAPK	Stress-activated protein kinase
SDS	Sodium dodecyl sulfate
ser	Serine
SRB	Sulforhodamine B
t	Triplet (NMR spectroscopy)
THF	Tetrahydrofuran
thr	Threonine
TLC	Thin layer chromatography
tyr	Tyrosine
UV	Ultraviolet
w	Weak (IR spectroscopy)

# CONTENTS

# CHAPTER 1 CANCER - A GENERAL OVERVIEW

1.1 Cancer - an epidemic	1
1.2 The aetiology of cancer	1
1.2.1 Chemicals	1
1.2.2 Radiation	2
1.2.3 Lifestyle	3
1.2.4 Viruses	3
1.2.5 Cancer-causing genes	4
1.2.5.1 Oncogenes (and proto-oncogenes)	4
1.2.5.2 Tumour suppressor genes	4
1.3 The cell cycle	4
1.3.1 Cancer cell division	5
1.4 The treatment of cancer	5
1.4.1 Surgery and radiotherapy	5
1.4.2 Chemotherapy	6
1.4.2.1 Alkylating agents	6
1.4.2.1.1 Nitrogen mustards	6
1.4.2.1.2 Platinum complexes	8
1.4.2.1.3 Nitrosoureas	9
1.4.2.2 Antimetabolites	9
1.4.2.2.1 Fluoropyrimidines	9
1.4.2.2.2 Thiopurines	10
1.4.2.2.3 Antifolates	10
1.4.2.3 Topoisomerase II inhibitors	11
1.4.2.4 Antimitotic agents	12
1.4.2.4.1 Catharanthus (Vinca) alkaloids	13
1.4.2.4.2 Paclitaxel (Taxol)	13
1.4.3 Immunotherapy	14
1.4.3.1 Monoclonal antibodies	14
1.4.3.2 Biological response modifiers	15

# CHAPTER 2 MALIGNANT MELANOMA AND ANTIMELANOMA AGENTS

2.1 Skin cancer - a misunderstood term	16
2.2 Types of malignant melanoma	16
2.3 Malignant melanoma - an epidemic	17

2.4 Risk factors and the development of malignant melanoma	17
2.4.1 Skin type	18
2.4.2 Naevi	18
2.4.3 Early history and genetics	18
2.5 Breslow thickness and the stages of melanoma	19
2.6 Treatment of malignant melanoma	20
2.6.1 Surgery	20
2.6.2 Radiotherapy	20
2.6.3 Chemotherapy	20
2.6.3.1 Single agents	21
2.6.3.2 Combination chemotherapy	21
2.6.4 Biologic therapy	22
2.6.4.1 Interferons	22
2.6.4.2 Other immunotherapies	22
2.7 Preventative medicine?	23
2.8 The melanin pigmentary system	23
2.8.1 Melanin	24
2.8.2 Tyrosinase	25
2.8.3 Melanosome	25
2.8.4 Melanocytes	25
2.8.5 Keratinocytes	25
2.8.6 Epidermal melanin unit	26
2.9 Strategy for specific drug design	26
2.9.1 Tyrosinase - the key target	26
2.9.2 The journey of discovery to parent drug	27
2.9.3 Phenolic prodrug mechanism of action	28
2.9.4 The research story so far	28
2.9.5 Proposed work	29

# CHAPTER 3 CELL SIGNALLING AND MAP KINASE INHIBITORS

3.1 Protein kinases and cell signalling pathways	31
3.1.1 Protein kinase superfamily	31
3.1.2 Mitogen-activated ser/thr kinase pathways	32
3.2 MAP kinase family	33
3.2.1 Pioneering work	33
3.2.2 MAP kinases in yeast	33
3.2.3 MAP kinase subfamilies	34
3.3 MAP kinase pathways	34

3.3.1 Ras/ERK MAP kinase components	35
3.3.1.1 MEK	36
3.3.1.2 Raf	36
3.3.1.3 Ras	36
3.3.1.4 ERKs	37
3.3.2 JNK/SAPK MAPK pathway	37
3.4 MAPK pathways and cancer	38
3.5 MAPK inhibition	39
3.6 Model for MAPK inhibitors in this work	39

# CHAPTER 4 SYNTHESIS AND BIOLOGICAL EVALUATION OF ANTIMELANOMA AGENTS

4.1 Phenolic thioether target compounds	41
4.1.1 Oxazolines	41
4.1.1.1 Synthesis of oxazolines	42
4.1.1.2 Reactivity of 2-oxazolines	43
4.1.2 Synthesis of the phenolic thioether target compounds	45
4.1.2.1 The Wehrmeister reaction	45
4.1.2.2 Overall route to general phenolic thioether target compounds	46
4.1.2.2.1 Step 1: Oxazoline ring-opening	46
4.1.2.2.2 Step 2: N-Alkylation of secondary amides	47
4.1.2.2.3 Step 3: Methyl ether deprotection	51
4.1.3 Biological evaluation of phenolic thioether compounds	55
4.1.3.1 In vitro 96 hour sulforhodamine growth inhibition assay	55
4.1.3.1.1 Results	56
4.1.3.1.2 General observations	57
4.1.3.1.3 Conclusions	57
4.1.3.2 In vitro mushroom Tyrosinase spectrophotomeric assay	58
4.1.3.2.1 Results and discussion	59
4.1.3.3 Correlation of assay results	59
4.1.3.3.1 Conclusions on biological test results	60
4.2 Amidine salts target compounds	60
4.2.1 Amines and 2-oxazoline ring-opening reactions	60
4.2.2 Synthesis of amidine salts	61
4.2.3 Biological evaluation of amidine salts	66
4.2.3.1 Results and discussion	66
4.2.3.2 Conclusions	68
4.3 Tyrphostin target compounds	68
4.3.1 The Knoevenagel condensation reaction	68

4.3.1.1 Mechanism for Knoevenagel reaction	69
4.3.2 Synthesis of tyrphostin target compounds	70
4.3.2.1 Synthesis results	71
4.3.3 Biological evaluation of tyrphostins	73
4.3.3.1 Results and discussion	73
4.3.3.2 Conclusions	75
4.4 Recent developments	75
4.5 Conclusions and future plans	76

# CHAPTER 5 SYNTHESIS AND BIOLOGICAL EVALUATION OF MAP KINASE INHIBITORS

5.1 Background	78
5.2 2-Iminolactone target compounds	78
5.2.1 2-Iminolactone ring system	78
5.2.2 Structure elucidation of 2-iminolactones	79
5.2.3 Synthesis of 2-iminolactone target compounds	80
5.2.3.1 Strategy for synthesis	80
5.2.3.2 Synthesis	80
5.2.4 Biological evaluation of 2-iminolactone target compounds	81
5.2.4.1 In vitro JNK MAP kinase assay	81
5.2.4.2 Results	82
5.2.4.3 Trends	83
5.3 Tyrphostin target compounds	84
5.3.1 Background	84
5.3.2 Synthesis of tyrphostin target compounds	84
5.3.2.1 Synthesis	84
5.3.2.2 NMR studies into stereoselective Knoevenagel products	86
5.3.3 Biological evaluation of tyrphostin target compounds	87
5.3.3.1 Results	87
5.3.3.2 Trends	88
5.4 3-Coumarin target compounds	89
5.4.1 Target compound design strategy	89
5.4.2 Coumarins	89
5.4.2.1 Naturally occurring coumarins	89
5.4.2.2 Properties of coumarins	90
5.4.2.3 Syntheses of coumarins	91
5.4.3 Synthesis of 3-coumarin target compounds	92
5.4.4 Biological evaluation of 3-coumarin target compounds	94
5.4.4.1 Results	94

5.5 Overall results from biological evaluation of target compounds	95
5.5.1 Correlation of results	95
5.5.2 Conclusions	95
5.6 Thiophene isostere target compounds	95
5.6.1 Synthetic proposal	95
5.6.2 Some biologically active thiophene systems	96
5.6.3 Thiophene tyrphostin target compounds	97
5.6.3.1 3-Methoxythiophene	97
5.6.3.2 Formylation of 3-methoxythiophene	97
5.6.3.3 Synthesis of thiophene tyrphostin target compounds	99
5.6.4 Thiophene iminolactone target compounds	100
5.6.4.1 Two synthetic routes proposed	100
5.6.4.2 Attempted synthesis of iminolactones using route 1	101
5.6.4.2.1 Synthesis of 2-formyl-3-hydroxythiophene	101
5.6.4.2.2 Attempted synthesis of iminolactones	102
5.6.4.3 Attempted synthesis of iminolactones using route 2	103
5.6.5 Biological evaluation of thiophene tyrphostin target compounds	103
5.6.5.1 Results	103
5.6.5.2 Conclusions	104

# CHAPTER 6 EXPERIMENTAL

6.1 General	105
6.2 Experimental for Chapter 4	106
6.2.1 General preparative methods 1 and 2	106
6.2.2 Experimental data	106
6.3 Experimental for Chapter 5	135
6.3.1 General preparative methods 3 to 6	135
6.3.2 Experimental data	136

# REFERENCES

155

# SUMMARY

Two separate cancer topics are included in our studies of the synthesis and biological evaluation of novel anticancer agents. These topics are synthesis of antimelanoma agents to combat malignant melanoma and synthesis of MAP kinase inhibitors designed to interfere with cell signalling pathways.

In the chemotherapeutic approach towards malignant melanoma treatment a selective approach can be adopted in drug design. Melanin is synthesised within specialised cells called melanocytes and malignant melanoma results when the control of melanocytic cell division is disturbed leading to excessive melanin production. The target strategy behind malignant melanoma drug design exploits specifically Tyrosinase, an enzyme crucial to melanin synthesis and unique to the melanocyte. Tyrosinase catalyses the conversion of L-tyrosine (i) into dopaquinone (ii).



Design strategies concentrate on phenolic prodrugs modelled on the natural substrate (i) of the oxidase enzyme. Tyrosinase catalyses the conversion of the phenolic prodrug into the corresponding *o*-quinone. These *o*-quinones are the mediators of cytotoxicity believed to impair the melanocytes' metabolism by combining with sulfhydryl-containing entities crucial to the cell cycle such as DNA polymerase. Alternatively the *o*-quinones may participate in a cyclic redox process resulting in the production of active oxygen species. The parent drug in the antimelanoma field, (iii), has been found to possess significant activity. An extensive programme of research was performed by Neil Lant at the University of Glasgow into the structure activity relationships of related compounds.



Our programme of research involved the synthesis of three sets of target compounds modelled on the lead compound designed as substrates for Tyrosinase. These tertiary amides (iv), amidine salts (v) and tyrphostins (vi) were successfully prepared and biologically evaluated for *in vitro* antimelanoma activity and relative substrate activity for Tyrosinase.



The lipophilic tertiary amides (iv) proved to be the most successful as potential antimelanoma agents with the *N*-benzyl methyl analogue in particular exhibiting good levels of cytotoxicity and selectivity for Tyrosinase-containing melanoma cell lines with GI50 values in the range 5-30  $\mu$ M. Furthermore this compound was a Tyrosinase substrate. Although the amidine salts indicated some levels of cytotoxicity and selectivity their cytotoxicity is believed to be the result of a non-Tyrosinase mediated mechanism of action. In general this biological behaviour exhibited by the amidines (v) was mirrored by the tyrphostins (vi). In conclusion we believe that the combination of the lipophilic nature of the tertiary amides and the close structural resemblance to the parent compound (iii) provided the series (iv) with better antimelanoma activity than the other two series (v) and (vi).

The second area of cancer chemotherapy research investigated is concerned with cell signalling pathways and inhibition of mitogen-activated protein (MAP) kinases which will prevent division of cancer cells. In our studies into MAP kinase inhibitors we employed 2-iminolactone derivative (vii) as our lead which had been identified from a preliminary screen within the Robins research group. This compound is modelled on serine, an amino acid residue phosphorylated by the MAP kinase enzyme.



Our programme of research first investigated the optimum position of the methoxy substituent on the benzene ring of the lead compound with the series (viii) prepared. The corresponding open chain derivatives (ix) and the coumarins (x) were also prepared.



The *in vitro* inhibitory assay results for the 2-iminolactones (viii) and tyrphostins (ix) indicated worthwhile candidates for a more detailed study into MAP kinase inhibitory behaviour. However, the coumarins (x) did not exhibit inhibitory behaviour. In response to the biological study results the preparation of thiophene bioisosteric replacements (xi) and (xii) of the 2-iminolactone (viii) and tyrphostins (ix) was proposed. Only the thiophene tyrphostins (xii) were successfully prepared and the MAPK inhibitory activity shown by the thiophene tyrphostins appeared to support the strategy behind their design.



# **CHAPTER 1**

# **CANCER - A GENERAL OVERVIEW**

# **1.1 CANCER - AN EPIDEMIC**

The World Health Organisation reported that an estimated 10 million cases of cancer occurred world-wide in 1998. Furthermore they predicted a continuation of this onslaught with numbers rising to 14.7 million over the next 20 years.<sup>1</sup> The current situation is that the proportion of fatalities from cancer in the population is increasing simply because fewer people are dying from other illnesses.<sup>2</sup>

Many factors combine to ensure the prevalence of this epidemic which is especially high in the industrialised nations. Among the causative factors are increased life expectancy, growing urbanisation, industrialisation and changes in lifestyle. Although the incidence in the number of cancer cases (and deaths) could be significantly lowered through changes in factors such as lifestyle and diet, preventative measures for many cancers remain either unknown or not applicable.<sup>2</sup>

In these cancer-stricken times, comfort must be taken from the technological advances and scientific progress which continue to be made by the present science community. These developments in turn enable a greater understanding of epidemiological diseases such as cancer and ultimately should facilitate the progress towards prevention, treatment and cure for most, if not all, cancers by sometime in the 21st century.

## **1.2 THE AETIOLOGY OF CANCER**

The term carcinogen is a very loosely used one and by definition encompasses any agent which confers an increase in cancer risk. This is a vast topic and under this heading of aetiology of cancer some of the main causative factors of cancer will be presented.

#### 1.2.1 CHEMICALS

1

The cancer-causing potential of chemicals was first recognised over 200 years ago by the English physician Pott who linked soot to the incidence of scrotal cancer in chimney sweeps.<sup>3</sup> The carcinogen involved, a polycyclic aromatic hydrocarbon, was not identified until much later. This exemplifies the workplace as one source of chemical carcinogens, with other sources including pollutants in air and water, foodstuffs and natural products.<sup>4</sup> A monograph series published by the International Agency for Research on Cancer (IARC)<sup>5</sup> is one of many publications that details chemicals with carcinogenic potential for employers, workers and the public in general to consult.

The polycyclic aromatic hydrocarbons like those found in the chimney soot are far more abundant today as air pollutants. With the onset of industrialisation these chemicals are prevalent in exhaust fumes and factory emissions.<sup>4</sup> A very well publicised chemical carcinogen of modern times is asbestos,<sup>6</sup> with pneumoconiosis or asbestosis affecting workers in that area. Furthermore, it is believed that exposure to this chemical increases the individual's risk of lung cancer and mesothelioma.<sup>6</sup>

These are just a few representatives of the vast number of chemical carcinogens known. However, many chemical cancer causing agents found in some sources are present in very low levels such that the attributable risk is considered to be negligible, e.g. some food colourants.

#### **1.2.2 RADIATION**

Following Roentgen's discovery of X-rays in 1895 the first reports of radiation carcinogenesis emerged. The cancer appeared on the hands of pioneering radiologists and later cases of leukaemia were to emerge.<sup>7</sup> The carcinogenic nature of ionising and ultraviolet (UV) radiation has been well documented since these early reports. Radiation cancer risk studies were carried out on individuals including patients irradiated for medical treatment and the Hiroshima and Nagasaki atomic bomb survivors.<sup>8</sup> These studies confirmed the radiation exposure/cancer risk correlation. Some of the cancers reported with an increased risk of development following exposure include leukaemia and cancer of the breast, lung and stomach.<sup>9</sup>

UV light radiation is a well known carcinogen.<sup>10</sup> The correlation between malignant melanoma and UV light is an established phenomenon with overexposure to this form of radiation thought to increase the cancer risk.<sup>11</sup> The

use of UV artificial light sources such as sun lamps and beds confirms these findings.<sup>12</sup> The UV light/melanoma topic is discussed in greater detail in Chapter 2.

### **1.2.3 LIFESTYLE**

Of all the causative factors in cancer the one responsible for the most cancer deaths is cigarette smoking.<sup>13</sup> So great are the number of deaths linked to smoking that cigarette companies are legally bound to advertise the fatal implications of this habit on their packaging. As well as lung cancer, other tobacco-related cancers include cancer of the oral cavity, pharynx, urinary bladder and pancreas.<sup>13</sup>

The American Institute for Cancer Research is devoted to cancer prevention through changes in diet and nutrition. This research body reports reductions in some cancers of 60 to 70% by changes in lifestyle such as diet, weight and physical activity.<sup>1</sup> The pioneering work in this area by Tannenbaum<sup>14</sup> demonstrated correlations between animal diets and risk of breast cancer with increased carcinogenicity found in animals consuming a diet high in fat and calories compared to those adhering to normal diets.

#### 1.2.4 VIRUSES

Cancer-causing viruses can be divided according to their genetic makeup such that DNA- and RNA-containing tumour viruses exist.<sup>15</sup> Retroviruses are the oncogenic viruses with an RNA genome exclusively. The first reported human retrovirus, human T-lymphotropic virus-type 1 (HTLV-1) infects mature T-lymphocytes which results in adult T-cell leukaemia-lymphoma (ATL). ATL incurs high mortality rates and is geographically restricted with the highest incidence found in Japan and the Caribbean.<sup>16</sup> Consider the second group of oncogenic viruses. Examples of these DNA-containing tumour viruses are more plentiful, e.g. the Epstein-Barr virus (EBV). This infects B lymphoma cells producing lymphoproliferative diseases including Burkitt's lymphoma and infectious mononucleosis.<sup>17</sup> Other DNA-containing tumour viruses such as human papilloma virus (HPV) and herpes simplex virus-type 2 (HSV-2) both produce cervical cancer.

#### **1.2.5 CANCER-CAUSING GENES**

Genes are well established as a causative factor of many cancers. The two main types of these genes are oncogenes and tumour suppressor genes.

#### 1.2.5.1 Oncogenes (and proto-oncogenes)<sup>18</sup>

Oncogenes are the main cancer-causing genes. When expressed these genes encode protein products which bring about changes to a normal cell characteristic of malignancy. Proto-oncogenes are genes found in every cell in the human body which can be converted into oncogenes, e.g. by mutation or deletion which then enables them to assume an oncogenic function. In normal cells proto-oncogenes encode proteins which play important roles in cellular processes such as proliferation and growth.<sup>19</sup> Therefore the net effect of the proto-oncogene/oncogene transformation is that proteins are encoded with aberrant cellular functions and the transformed cell proliferates in an uncontrolled manner, characteristic of a cancer cell.<sup>20</sup> Proto-oncogenes and their role in malignant transformations of normal cell are discussed in greater detail in section 3.4.

#### 1.2.5.2 Tumour suppressor genes

Tumour suppressor genes play a regulatory role in normal cell growth. These genes are activated on demand to halt the cell growth process. However, when these genes are lost or inactivated, their inhibitory regulation of cell growth is thus removed whereupon malignancy results.<sup>21</sup> Wilms' tumour gene is a common suppressor gene. Deletions or mutations to this gene on the short arm of chromosome 11 are attributed to Wilms' tumour, the most prevalent childhood kidney cancer.<sup>22</sup> The *CDKN2* tumour suppressor gene is the most important familial malignant melanoma gene.<sup>23</sup> This gene and its relationship with malignant melanoma is discussed further in section 2.4.3

# 1.3 THE CELL CYCLE<sup>24</sup>

Cell division plays a role of immeasurable importance in the life processes of various organisms. Just consider for example the unicellular organism Amoeba which reproduces by cell division or that multicellular organisms require cell division for growth and repair.<sup>25</sup> The cell cycle comprises of an ordered sequence of definitive events within the dividing cell. This cycle is made up of the mitotic (M) phase and interphase with 90% of cycle time devoted to interphase. In G<sub>1</sub>, G<sub>2</sub> and S, the three sub phases of interphase, the cell grows (G<sub>1</sub> and G<sub>2</sub> sub phases) and chromosomes are copied (only in S subphase). On reaching the mitotic phase of the cycle the cell is divided whereupon two identical daughter cells are produced.<sup>26</sup>

The crucial point for the cell cycle occurs within  $G_1$  phase at the restriction point. A trigger at this point is required for procession into the S phase. However, the cell can depart from the cycle at this point into the  $G_0$  phase, the phase in which most human cells are found. The proliferative nature of cells vary such that some cells, for example marrow cells rapidly divide, whereas other cells like those of the nerves or muscle, never divide. However, certain cells including lung and liver cells, have the potential to enter the cell cycle when required which is usually only for repair purposes.<sup>26</sup>

#### **1.3.1 CANCER CELL DIVISION**

Cancer cell proliferation is uncontrolled with the transformed cell failing to respond to the regulatory mechanisms for normal growth control.<sup>27</sup> The malignant transformation of normal cells to cancer cells is under genetic control with the genes which are controlling cell proliferation being active beyond control and apparently unable to be switched off. The mechanism which maintains the genes in this active state holds the key to fully understanding cancerous cell production and the genetics behind its control.<sup>27</sup>

# **1.4 THE TREATMENT OF CANCER**

#### **1.4.1 SURGERY AND RADIOTHERAPY**

Cancer in its many forms can be treated with a variety of therapeutic agents which include surgery, radiotherapy, chemotherapy and immunotherapy.

The use of surgery in cancer treatment is relatively diverse and very much dependent on the nature of the cancer, e.g. surgery is fundamental for use in biopsies and localised excisions but limited in the successful treatment of many advanced neoplastic cancers.<sup>28</sup> Common to most cancer therapies, surgery is regularly supplemented with other approaches.

Radiotherapy is a potentially very effective treatment of cancer, hampered only by its ability to damage normal cells. Irradiation sources include X-rays, γ-rays and radioactive isotopes of elements such as radium and cobalt.<sup>29</sup> This therapeutic agent is common in adjuvant treatment for post-surgical patients to limit the risk of recurrence. In the treatment of Hodgkin's disease, radiotherapy and chemotherapy (see section 1.4.2) in combination or as single agents provide one of the most successful treatments of a cancer regardless of the stage of the disease.<sup>30</sup> Computerised tomography<sup>31</sup> and NMR imaging<sup>32</sup> advances have enabled improvements in surgery and radiotherapy by providing a more defined image of the cancer. These imaging modems enable a precise treatment of infected tissues with normal tissues affected as little as possible.

\$

#### 1.4.2 CHEMOTHERAPY<sup>33, 34</sup>

The vast range and number of cancer chemotherapeutic agents in clinical use today alone means that this topic cannot be discussed here in great detail. However, a representative discussion of the major classes of anticancer drugs will follow describing alkylating agents, antimetabolites, topoisomerase II inhibitors and antimitotic agents.

## 1.4.2.1 Alkylating agents

This class of chemotherapeutic agents react chemically with the cancer cells' own DNA. On rendering the DNA structurally and functionally redundant, cell death occurs. The alkylating agents can be classified into nitrogen mustards, platinum complexes and nitrosoureas.

#### 1.4.2.1.1 Nitrogen mustards<sup>35</sup>

The tertiary amine methyl bis(2-chloroethyl)amine (mechlorethamine) **1** was the first clinical cancer chemotherapeutic agent used and still finds employment today as part of a drug combination regimen to treat Hodgkin's disease.<sup>36</sup> Chlorambucil **2** and melphalan **3**, both structural analogues of mechlorethamine are widely used antitumour agents in combination or as single agents. Notably melphalan is used as an adjuvant to surgery in primary breast cancer treatment.<sup>37</sup>



The mechanism of cytotoxicity for these three nitrogen mustards is the same with the bis(2-chloroethyl)amino moiety providing the structural feature required.<sup>35</sup> The nitrogen mustards function as alkylating agents by crosslinking the double strands of DNA. Cell death ensues as a crosslinked DNA species cannot unwind and replicate to complete the cell cycle. A simplified illustration for this cytotoxic mechanism of the action of nitrogen mustards using mechlorethamine **1** is shown in Scheme 1.



On forming the aziridine species **4**, the nitrogen mustard is set up for nucleophilic attack. For cytotoxicity leading to cell death DNA provides the source of nucleophilicity. Nucleotide bases on the DNA, e.g. guanine provide the nucleophilic site (N-7 position) for attack of the aziridine. The monoadduct product **5** can result in the mispairing of bases or strand breakage of the DNA; however the cell can invariably repair this damage. On the other hand, the monoadduct can be further involved in a repeat crosslinking process with a final

adduct **6** linking either the same DNA strand (intrastrand crosslink) or as is more commonly found linking complementary DNA strands (interstrand crosslink).<sup>35</sup>

The leading alkylating agent, cyclophosphamide 7 is not itself active, requiring metabolic activation before it can exert cytotoxicity against the cancer. Amongst the cyclophosphamide metabolites produced are phosphoramide mustard 8 and acrolein 9 and these are believed to be the cytotoxic species with the phosphoramide mustard 8 behaving as a DNA crosslinking species. Cyclophosphamide exhibits broad spectrum anticancer activity which is reflected in its wide use in the treatment of cancers such as acute lymphocytic leukaemia, breast carcinoma and Hodgkin's disease.<sup>39</sup>



1.4.2.1.2 Platinum complexes

Of the platinum complexes cisplatin **10** exhibits the best antitumour effects with activity against several human cancers when used alone or in combination drug therapy.<sup>40</sup> Structure activity studies identified the *cis*-arrangement of ligands in cisplatin to be imperative for activity.<sup>40</sup> The cytotoxicity of this platinum complex is believed to arise from DNA interactions whereby the main adducts form from intrastrand DNA crosslinks.<sup>41</sup> Carboplatin, iproplatin and tetraplatin are among the most potent of 2000 or so analogues of cisplatin synthesised. However, they do not improve on cisplatin itself as they fail to reduce levels of toxicity or resistance of tumours found with repeated cisplatin drug use.<sup>40</sup>

#### 1.4.2.1.3 Nitrosoureas

Random screens performed by the National Cancer Institute in 1959 prompted the first work into nitrosoureas as potential chemotherapeutic agents.<sup>42</sup> Later 1,3-bis(2-chloroethyl)-1-nitrosourea (BCNU) **11** emerged as one of the first nitrosoureas with anticancer potential, followed by lomustine (CCNU) **12** and its 4-methyl homologue, semustine (MeCCNU) **13**.<sup>43</sup> The replacement of the chloroethyl moiety in BCNU **11** with a cyclohexyl ring in CCNU **12** and MeCCNU **13** was found to improve lipophilicity thus greatly assisting absorption. This enhanced lipophilicity enabled penetration of the blood-brain barrier with central nervous system tumours able to be reached.<sup>44</sup> Although all three nitrosoureas are employed in drug combination chemotherapy, each of them are found to exhibit significant activity in the treatment of Hodgkin's disease<sup>44</sup> and brain tumours. DNA is the major target of the nitrosoureas with evidence of interstrand crosslinking and single strand breaks reported.<sup>43</sup>



#### 1.4.2.2 Antimetabolites

Antimetabolites target the metabolic pathways of the cancer cells. These agents usually take the form of synthetic analogues of the normal metabolites with the prime objective being to halt the uncontrolled cell growth in the cancer patient.<sup>45</sup> In effect, antimetabolites assume the role of an imposter in the metabolic pathways of cancer cells. The most successful antimetabolite drugs are found to affect the pathways leading to the production of DNA-related components.<sup>45</sup> The main antimetabolites include fluoropyrimidines, thiopurines and antifolates.

#### 1.4.2.2.1 Fluoropyrimidines

The lead fluoropyrimidine, 5-fluorouracil (5--FU) **14** was designed as a pyrimidine base mimic. The cytotoxicity of 5-FU<sup>†</sup> results from DNA-directed action of the drug due to incorporation of 5-FU into DNA. This fluoropyrimidine as a single chemotherapeutic agent is most commonly used in the treatment of

solid tumours, in particular breast and gastrointestinal tract carcinomas. However, it is employed to a greater extent in combination drug therapy.<sup>45</sup> For example, 5-FU in a three-drug regimen with doxorubicin (see section 1.4.2.3) and cyclophosphamide is used to treat advanced breast cancer.<sup>46</sup> Derivatisation of fluoropyrimidines has been found to be biologically very useful. One such example is the prodrug ftorafur **15** which exhibits antitumour activity against a range of solid tumours.<sup>47</sup>



#### 1.4.2.2.2 Thiopurines

Of the unnatural purines investigated for anticancer activity, 6mercaptopurine **16** and 6-thioguanine **17** proved to be among the most promising.<sup>48</sup> However, 6-mercaptopurine **16** proved to be the most potent of the two antimetabolites exhibiting activity against acute childhood leukaemia.<sup>49</sup> The tumour response with 6-mercaptopurine **16** has been correlated with conversion of the drug into the corresponding ribonucleotide.<sup>48</sup>



#### 1.4.2.2.3 Antifolates

The folates are coenzymes involved in many biochemical transformations. Dihydrofolate reductase (DHFR) is among the many folateutilising enzymes and plays a key role in pathways leading to the formation of nucleic acids.<sup>50</sup> Methotrexate (MTX), **18** a classical antifolate, potently inhibits DHFR resulting in cytotoxicity. Used in the treatment of choriocarcinoma and acute lymphocytic leukaemia,<sup>51</sup> MTX **18** is also employed in various combination drug regimens. However major limitations in MTX drug use arise via toxic side effects and the development of resistance to the drug by tumour cells.<sup>51</sup> These are problems commonly faced by many antitumour agents.



#### 1.4.2.3 Topoisomerase II inhibitors

DNA topoisomerase II is a homodimeric protein responsible for the control of double stranded DNA topology. When required these enzymes function by first breaking the paired DNA strands and then later reforming the breaks e.g. during transcription whereby the DNA returns to its original strand-paired state.<sup>52</sup> This characteristic topoisomerase II action has seen these enzymes implicated in facilitating the cytotoxicity of intercalating drugs. These agents bind tightly to DNA but this alone does not correlate with the cytotoxicity observed.<sup>53</sup> A complex mechanism is proposed whereby the intercalators inhibit the breakage-reunion reaction of the topoisomerase II.<sup>54</sup> Cell death arises from lethal breaks in the double stranded DNA.<sup>54</sup>

The anthracycline antibiotics are DNA intercalating agents believed to target topoisomerase II enzymes to elicit their anticancer activity. Isolated from a *Streptomyces* species, the anthracyclines include Daunomycin (daunorubicin) **19** and Adriamycin (doxorubicin) **20** which were the first anthracycline antibiotics to be used in tumour chemotherapy. Notably Daunomycin **19** predominates in the treatment of acute leukaemia. However the side effects of toxicity and alopaecia are standard.<sup>55</sup> The functional diversity of the anthracyclines is of particular interest because in addition to inhibiting topoisomerase II, these chemotherapeutic agents chelate transition metal ions and generate free radicals within the tumour cells.<sup>55</sup>



The synthesis of anthracycline analogues has provided little improvement on anticancer activity. However, the synthetic anthraquinone mitoxantrone **21** although not strictly an anthracycline analogue is structurally quite similar. While it demonstrates similar cytotoxicity,<sup>56</sup> mitoxantrone **21** reduced overall toxicity suggesting perhaps that the toxicity in the anthracyclines may be attributable to the varied functionality they exhibit. Metastatic breast cancer, leukaemias and renal cell carcinoma are among the cancers with which mitoxantrone has produced responses.<sup>56</sup>



#### 1.4.2.4 Antimitotic agents

As the name infers this class of compounds act during the mitotic phase of the cell cycle. In normal cells spindle apparatus operates in this mitotic phase by controlling the division of chromosomal material, a process crucial to successful cell division. The spindle is constructed from microtubule proteins which co-exist in a balanced state with tubulin.<sup>57</sup> Antimitotic agents disturb this delicate microtubule/tubulin balance and by preventing spindle formation halt cell growth.<sup>57</sup>

## 1.4.2.4.1 Catharanthus (Vinca) alkaloids



Vinblastine **22** and vincristine **23** isolated from the Madagasian periwinkle plant *Catharanthus roseus* are amongst the best known Vinca alkaloids.<sup>58</sup> These antimitotic agents function by binding to dimeric tubulin which perturbs the balanced microtubule/tubulin state. Inhibition of spindle construction and function results, followed by cell death.<sup>59</sup> Vinblastine **22** and vincristine **23** are used to treat rapidly growing cancers such as leukaemia but neurotoxic side effects are common.<sup>59</sup> This toxicity along with multidrug resistance has hampered the rapid progression with further natural or synthetic Vinca alkaloids.<sup>58</sup> However, vindesine **24** is a more recently identified antimitotic agent used to treat leukaemia.<sup>60</sup>



## 1.4.2.4.2 Paclitaxel (Taxol)

Unlike the Vinca alkaloids, the antimitotic agent paclitaxel (Taxol) **25** disturbs the delicate microtubule balance by catalysing microtubule formation.<sup>61</sup> The equilibrium becomes heavily unbalanced in favour of the microtubule

protein and hence cell death ensues due to incompletion of the mitotic phase of the cell cycle. The isolation of Taxol from the stem bark of the Pacific yew *Taxus brevifolia* revealed an antimitotic agent with potent anticancer properties.<sup>62</sup> This discovery prompted the quest for a synthetic source of this very scarce natural product. However, it was over 20 years before the first total synthesis emerged from the groups of Nicolaou<sup>63</sup> and Holton<sup>64</sup> closely followed by Danishefsky.<sup>65</sup> From the endeavours of these workers and others, chemistry crucial for the semi-synthetic production of Taxol was revealed. Initially approved for ovarian cancer treatment,<sup>66</sup> this potent drug exhibits significant activity against breast, lung and head and neck cancers.<sup>67</sup> The literature continues to be saturated with reported uses of Taxol. Whether on its own or in combination chemotherapy, this exceptionally promising agent is employed in treatment of the advanced metastatic forms of many cancers.

#### 1.4.3 IMMUNOTHERAPY<sup>68</sup>

The immune system removes any agent or agents it regards as foreign to the human body by eliciting a specific immunological response. The potential for an immunological role in cancer therapy requires tumour cells to have target molecules recognised by antibodies or cells involved in the immune process.<sup>69</sup> Current approaches to immunotherapy include monoclonal antibodies and biological response modifiers.

#### 1.4.3.1 Monoclonal antibodies

The use of monoclonal antibodies in cancer treatment is very limited. These antibodies are employed to recognise antigens specifically associated with the tumour cells.<sup>70</sup> Some antigenic tumours include melanoma ovarian and colon cancers. Problems which limit this form of immunotherapy can arise from immunological responses removing the antibodies from the system.<sup>70</sup> However, monoclonal antibody-directed therapy enjoys a far greater potential in cancer therapy. In effect, this therapy links the antibody to a drug thus directing the drug to the tumour cell surface.<sup>71</sup> Shortages of specific antibodies and insufficient quantities of drug delivered have produced problems.<sup>71</sup> However, promising results with experimental leukaemias or ascites tumour have been observed.<sup>72</sup>

#### 1.4.3.2 Biological response modifiers<sup>73</sup>

The biological response modifiers which are mostly polypeptides elicit their anticancer behaviour through regulating the immunological responses of the cancer patient.<sup>73</sup> The most promising biological response modifiers are the interferons. Interferons are usually produced by the body in response to viral infection and display inhibition of cell proliferation.<sup>73</sup> Gene cloning techniques have been employed to produce these immunotherapeutic agents and among the most successful are the interferons IFN- $\alpha$  and interleukin-2 (IL-2). IFN- $\alpha$  is used to treat malignancies including advanced lymphoma and renal-cell cancers. Interleukin (IL-2) is particularly effective against melanoma and kidney cancer.<sup>74</sup> However, the problem with these interferons and other biological response modifiers is that the mechanism of action remains unclear.

# **CHAPTER 2**

# MALIGNANT MELANOMA AND ANTIMELANOMA AGENTS

#### 2.1 SKIN CANCER - A MISUNDERSTOOD TERM

Skin cancer is a familiar term to the population in general, however, its true nature is not well understood. Unknown to many, skin cancer or cutaneous malignancy comprises three forms: basal cell carcinoma, squamous cell carcinoma and malignant melanoma.<sup>11</sup> Despite its comparative rarity malignant melanoma is the potential killer of the three forms.<sup>11</sup> However non-melanoma skin cancer (basal cell carcinoma and squamous cell carcinoma) is one of the most common cancers to affect men and women.<sup>2</sup> With mortality rates very low and tumours only rarely known to metastasise, sufferers of non-malignant melanoma can be treated in a very low key manner in general practice or as hospital outpatients.

The skin cancer discussed in depth from this point on is the comparatively rare but potentially lethal form, namely malignant melanoma.

# 2.2 TYPES OF MALIGNANT MELANOMA

There are four main types of malignant melanoma, namely superficial spreading, nodular, lentigo and acral melanomas.<sup>75</sup> The superficial spreading and nodular melanomas generally occur in younger people, whereas lentigo malignant melanoma is most likely to affect the faces of the elderly. The acral-lentiginous melanoma is a malignancy of the skin which affects the palms and soles.<sup>75</sup>

Of the four types, the superficial spreading and nodular melanomas are the most common and aggressive types. Between them they constitute approximately 90% of all malignant melanoma cases and most recent epidemiological studies concentrate only on these types.<sup>76</sup>

# 2.3 MALIGNANT MELANOMA - AN EPIDEMIC

Considering that malignant melanoma is the cause of only 1% of all new cancer cases,<sup>77</sup> why then is there so much concern associated with this particular cancer? With this in mind, consider the following very sobering aspects of the disease which should illustrate the reasoning behind such widespread concern.

- At present the incidence of malignant melanoma continues to escalate among the fair-skinned populations of the world.<sup>76</sup>
- The major aetiological factor attributed to the cancer is known, namely overexposure of skin to strong sunlight (ultraviolet light with wavelength 290-320 nm, UV-B).<sup>10,11</sup>
- This cancer can be cured but only if recognised in its early stages of development before the onset of metastasis.<sup>78</sup> Otherwise no curative therapy for advanced forms exists.
- The age at onset is earlier than in most other cancers. Over the period 1986-1995 it was the third most common cancer in the 15-34 age group.<sup>79</sup>

So in summary, unlike most other cancers, melanoma is potentially curable and the major cause known. It would therefore appear that through early identification of the cancer and simple education with regard to sun exposure the incidence and early age onset of the disease could be reduced.

# 2.4 RISK FACTORS AND THE DEVELOPMENT OF MALIGNANT MELANOMA

As already mentioned the main aetiological factor associated with skin cancer is exposure to solar radiation.<sup>10,11</sup> However, the risk of developing malignant melanoma can be further increased by a number of contributing factors or risk factors. These principal risk factors can be illustrated through discussion of skin type, naevi and genetics and family history.

#### 2.4.1 SKIN TYPE

Susceptibility to malignant melanoma is considered to be related to skin pigmentation,<sup>77</sup> i.e. those phenotypes with naturally occurring brown or black skin are at a far lower risk of developing malignant melanoma than fair-skinned individuals who have a tendency to burn or freckle rather than tan. It has also been proposed that episodes of severe sunburn and general overexposure to the sun can cause malignant melanoma <sup>80</sup>

## 2.4.2 NAEVI

Benign pigmented melanocytic naevi or "moles" as they are more commonly known, can be present from around birth (congenital naevi) or acquired later in life (common or acquired naevi). Both types of naevi are considered to be risk factors for malignant melanoma.<sup>81</sup>

Dysplastic naevi are acquired naevi. The risk of malignant melanoma has been reported as proportional to the number of these naevi<sup>80</sup> with large numbers proving to increase the significance of the risk factor. However, as solar exposure increases the number of naevi it remains unclear whether UV radiation promotes conversion of naevi into melanoma or acts as some sort of initiator.<sup>77</sup>

The other type of mole, congenital naevi, have been used as a powerful tool in public awareness campaigns to promote self-recognition of the early signs of the cancer.<sup>82</sup> Warning signs of malignant melanoma manifest themselves in featural changes of these moles, from changes in mole size, outline and colour to bleeding and itching.<sup>11</sup>

## 2.4.3 EARLY HISTORY AND GENETICS

Case studies identified the link between family history and melanoma risk with an individual's risk of developing malignant melanoma doubled if they have at least one first-degree relative with the cancer.<sup>83</sup> Statistically this represents approximately 2% of melanoma sufferers.<sup>84</sup> Furthermore, it would appear that genetic susceptibility and sun exposure are likely to act multiplicatively on melanoma risk.<sup>83</sup>

The search for the hereditary genetic component of malignant melanoma is an ongoing one. While 5-10% of malignant melanoma cases occur in a hereditary form,<sup>85</sup> a number of familial cancer genes have been cloned and characterised, most of which prove to be involved in sporadic cancers as well as hereditary cancers.<sup>86</sup> The most important familial melanoma gene reported to date is the p16<sup>INK4</sup>(*CDKN2*) MTS1 gene found on chromosome 9p21.<sup>87</sup> *CDKN2* possesses the hall marks of a tumour suppressor gene and is lost or mutated in many cell lines including melanoma.<sup>85,88</sup> Such *CDKN2* mutations have been reported in approximately 20-30% of all familial melanoma cases.<sup>89</sup>

# 2.5 BRESLOW THICKNESS AND THE STAGES OF MELANOMA

Malignant melanoma originates within the melanocyte where under conditions of uncontrolled melanocytic cell division excessive production of melanin occurs. This results in malignant melanoma appearing as dark naevi or lesions on the skin. However, the most feared aspect of melanoma is metastasis, the spread of the cancer, as metastasis can spread the cancer to distant sites within the body including the brain, lungs and gastrointestine.<sup>78</sup> It is of no surprise then that chances of survival are far greater when the malignancy is treated in the early stages of development, before metastasis.<sup>78</sup>

Of paramount importance in prognosis is the vertical thickness of the tumour known as the Breslow thickness.<sup>90</sup> Quite simply "the thinner the better" best describes tumour prognosis for recovery. Breslow thickness along with the anatomical level of local invasion are used to characterise malignant melanoma into four stages of development<sup>91</sup> as detailed below.

## STAGE I

- Tumour less than 1.5 mm thick and/or invades papillary-reticular dermal interface.
- No regional lymph node or distant metastasis.

## STAGE II

- Tumour between I.5 and 4 mm thick and/or invades reticular dermis.
- No regional lymph node or distant metastasis.

# STAGE III

- Tumour more than 4 mm thick with invasions as far as subcutaneous tissue.
- Metastasis in any regional node(s).

• No distant metastasis.

#### STAGE IV

• Mestasis in regional lymph node(s) and distant metastasis in skin tissue, subcutaneous tissue or lymph node(s) beyond the regional lymph nodes.

Having detected the cancer's stage of development, the appropriate treatment can be selected. However, treatment choice is greatly diminished for advanced metastatic tumours in the later stages of advancing metastasis.

# 2.6 TREATMENT OF MALIGNANT MELANOMA

The therapeutic forms of malignant melanoma treatment include surgery, radiotherapy, chemotherapy and biological therapy. As already implied, the applicability of each treatment is very much dependent on the stage of development of the cancer. Each of the four treatments will be discussed in turn.

# 2.6.1 SURGERY

Surgery is the most effective treatment of the early stages of melanoma.<sup>92</sup> Furthermore, studies reported that surgical margins need be no greater than 2 cm for the excision of melanomas of intermediate thickness (0.75 to 4.0 mm).<sup>93</sup> This significantly reduces the need for skin grafting with most patients treated on an outpatient basis. Surgery faces a far more limited role in the treatment of advanced melanoma. However, isolated metastases found in the lungs, gastrointestinal tract, bone or occasionally the brain, may be palliated by resection with some improvements in survival resulting.<sup>94</sup>

## 2.6.2 RADIOTHERAPY

Radiation therapy does not play a curative role in treatment of melanoma patients. However, this treatment is not redundant as radiotherapy provides symptomatic relief for sufferers with metastases including those affecting the central nervous system and bone.<sup>95</sup>

## 2.6.3 CHEMOTHERAPY

The chemotherapeutic approach to melanoma treatment can be divided into the use of drugs as single agents or as part of a combination regimen.

#### 2.6.3.1 Single agents

From the early 1970s dacarbazine (DTIC) **26** has remained the drug of choice for single agent treatment of metastatic melanoma. DTIC **26** displays response rates around 20%.<sup>96</sup> Before the discovery of DTIC **26**, some of the single agents in use included hydroxyurea, procarbazine and alkylating agents such as cyclophosphamide.<sup>97</sup> Their use was discontinued due to low response rates. Later cisplastin<sup>98</sup> and the Vinca alkaloid family (vindesine<sup>99</sup> and vinblastin<sup>100</sup>) demonstrated antitumour activity; in particular response rates of 10 to 15% were exhibited for cisplatin treatment.<sup>98</sup> The potential of these single agents was recognised by incorporating these drugs in combination chemotherapy regimens. To date, the nitrosourea family continues to produce encouraging single agents for melanoma treatment. Earlier nitrosourea examples include carmustine (BCNU), lomustine (CCNU) and semustine (MeCCNU)<sup>96</sup> which reported antitumour potency analogous to DTIC. However severe toxicological problems hampered further use of nitrosoureas such as CCNU.<sup>96</sup>



Fotemustine **27** is a relatively new nitrosourea with response rates reported in the region of DTIC.<sup>101</sup> Such was the antitumour potential of fotemustine **27**, it was incorporated immediately into combination therapeutic regimens with DTIC **26** and further studies for it as a single agent should be imminent. Many of these chemotherapeutic agents, their uses and mechanisms of action are discussed in section 1.4.2.

#### 2.6.3.2 Combination chemotherapy

In an attempt to improve on the antitumour activity of single agents, combination chemotherapy employs two and three combinations of drugs to treat metastatic melanoma. However, results have proven to be quite disappointing with the response rates observed for DTIC **26** never surpassed.
Combination chemotherapy regimens involving DTIC and the nitrosoureas BCNU, CCNU and MeCCNU displayed some potential but not comparable to the results from DTIC alone.<sup>102</sup> Similarly two and three drug combinations of vinblastine, vindesine and cisplatin with DTIC were to produce analogous observations in their results.<sup>103</sup> Encouraging results were obtained very recently in a study using DTIC and fotemustine. Although response rates were mediocre (11%), minimal side effects and no lung toxicity were observed. This provides a palliative treatment for advanced metastic melanoma.<sup>104</sup>

To date however, DTIC maintains the lead single chemotherapeutic agent for malignant melanoma and as yet no conclusive evidence has been reported to demonstrate combination chemotherapy as a superior mode of drug treatment.

## 2.6.4 BIOLOGIC THERAPY

#### 2.6.4.1 Interferons

The first evidence of effective adjuvant therapy for metastatic malignant melanoma treatment emerged recently using interferons.<sup>105</sup> These immunotherapeutic agents provide a novel source of anticancer agents (see section 1.4.3.2). The Eastern Co-operative Oncology Group (ECOG) trial 1684 reported using interferon alfa-2b (IFN- $\alpha$ 2b) at maximally tolerated doses for one year.<sup>106</sup> Overall survival was improved by approximately 1 year with estimated overall 5-year survival rates of 46% (compared to 37% for those who received no adjuvant therapy). Furthermore, the toxic side effects were tolerated by the majority of patients.<sup>106</sup>

#### 2.6.4.2 Other immunotherapies

One target in the development of immunological cancer vaccines are the ganglioside antigens found expressed on the cell surface of human malignant melanoma.<sup>107</sup> The ganglioside GM2 antigen vaccine (GM2 vaccine) induces antibody responses with a high percentage of patients and on combining the GM2 vaccine with *Bacillus calmette-guerin* (BCG), in the GM2/BCG vaccine, improvements in overall recovery and relapse times occurred.<sup>108</sup> To investigate these findings further, GM2 vaccine and Interferon IFN combination trials are expected.

Another approach in specific tumour immunotherapy arises from the discovery of T-cell recognised tumour associated antigens (TAA).<sup>109</sup> This has enabled the development of T-cell antigens vaccines which have been used in clinical trials for the past few years.<sup>110</sup>

To summarise the treatments of malignant melanoma discussed: although surgery enables treatment of early stage melanoma with relative ease and minimal disruption to the patient, advanced metastatic melanoma remains resistant to most therapies. However the reported encouraging responses from some chemotherapy and especially the emerging immunotherapies can surely provide hope for sufferers and provide concrete foundations for further work in this cancer field.

## 2.7 PREVENTATIVE MEDICINE?

When we consider the overwhelming contribution sun exposure plays in the incidence of skin cancer, never has prevention been such an attractive therapy. In an attempt to change habits with regard to sun exposure public awareness campaigns have been implemented world-wide from Australia to the UK.<sup>111</sup> These campaigns have a common theme whereby tanned skin is no longer to be considered a sign of beauty, wealth or status. Re-education tactics include promoting the use of sunscreens, avoidance of midday sun and special care of young children's skin whilst in the sun.<sup>112</sup> To enable the earliest treatment of this cancer, prompt identification through self-examination of moles is also encouraged (see section 2.4.2).

Sadly however, on reflection, the white skinned individual's urge for the perfect tan has had alarming consequences on a scale which could never have been predicted. It can only be hoped then that in this trend-following world, styles will revert back in time to the Victorian era when white almost porcelain-like skin was desired to portray the image of upper class status, wealth and social standing!

## 2.8 THE MELANIN PIGMENTARY SYSTEM

Malignant melanoma results from the aberration of melanocytic cell division which leads to excessive melanin production.<sup>113</sup> To understand the biological process of pigmentation itself<sup>114</sup> the functional components of the melanin pigmentary system should be examined. The more important

components include Tyrosinase, the melanosome, keratinocytes, melanocytes and melanin itself.

## 2.8.1 MELANIN

This brown-black pigment of the skin is synthesised within the melanocyte on the melanosome (a sub-cellular organelle). The virtual insolubility of melanin made elucidation of the biosynthetic pathway very difficult.<sup>115</sup> Scheme 2 depicts a very simplified version of this pathway. Two types of pigment, eumelanins **28** and phaeomelanins **29** (coloured brown-black and reddish respectively) are both seen to originate from tyrosine **30**, the amino acid starting material of the complex pathway.<sup>116</sup> The photoprotective role played by melanin is exemplified when one considers the ease with which a light-skinned individual burns compared to their naturally dark-skinned counterpart, e.g. Negro.<sup>115</sup>



#### 2.8.2 TYROSINASE

This aerobic oxidase enzyme is unique to the melanocyte cell where it plays a crucial role in melanisation.<sup>114</sup> The enzyme catalyses the conversion of tyrosine **30** into dopaquinone **31**, a reaction believed to proceed via the catechol L-dopa **32**.<sup>118</sup> However, recent studies indicate that L-dopa arises from the disproportionation of dopaquinone and acts as an activator of the Tyrosinase enzyme.<sup>119</sup> This also explains the unusual kinetic properties found with Tyrosinase oxidation of tyrosine and other phenols.<sup>119</sup> Dopaquinone **31** then reacts further to yield eumelanin **28** and on combining with sulfur-rich cysteine **33**, phaeomelanin **29** is produced (via cysteinyldopa **34** and subsequent intermediates) (Scheme 2).<sup>114</sup> The Tyrosinase activity is subject to regulatory control including negative feedback inhibition by melanin and control from intracellular cyclic AMP.<sup>120</sup> Perhaps of greater relevance to this discussion is the increase in tyrosinase activity caused by ultraviolet light.<sup>121</sup>

#### 2.8.3 MELANOSOME

The synthesis of melanin occurs within this subcellular organelle.<sup>116</sup> Four stages of this melanisation take place.<sup>122</sup> In stages I and II the melanosome changes from a spherical to an oval shape. The development of melanin synthesis proceeds from a partial state in stage III to completion by stage IV. The few melanosomes seen in pale-skinned individuals are found to be in stages I and II only. However, in Negroes for example, mostly stage IV melanosomes are present. The resultant effects of overexposure to UV light in the paler Caucasians can be observed with stage III melanosomes appearing following exposure.<sup>122</sup>

#### 2.8.4 MELANOCYTES<sup>113</sup>

Within these specialised cells melanosomes and the melanocyte enzyme Tyrosinase can be found.<sup>113</sup> Melanocytes are found in the mucous membranes, nervous system and eye (ureal tract and retinal epithelium) as well as the skin. Malignancy can arise in all these sites except the retinal pigment epithelium.<sup>113</sup> No correlation between melanocyte number and skin colour is thought to exist. However, exposure to UV light can increase the number of melanocytes.<sup>116</sup>

#### 2.8.5 KERATINOCYTES

The transfer of melanin from the melanocyte to the keratinocyte is essential in the pigmentation process.<sup>122</sup> Within the keratinocyte, while the larger melanosomes function individually, the smaller melanosomes congregate to maintain their functional purpose.<sup>122</sup>

#### 2.8.6 EPIDERMAL MELANIN UNIT

This functional unit is formed when the components of the melanin pigmentary system combine and interact functionally. The epidermal melanin unit is made up of one melanocyte surrounded by 20 to 36 keratinocytes.<sup>116,123</sup> The transfer of melanin from the melanocyte to the keratinocytes is believed to be crucial for pigmentation. Additional major factors in determining skin colour include the stage of melanisation, and the size and number of melanosomes.<sup>116,123</sup>

## 2.9 STRATEGY FOR SPECIFIC DRUG DESIGN

## 2.9.1 TYROSINASE - THE KEY TARGET<sup>124</sup>

The target strategy for malignant melanoma drug design is to exploit specifically Tyrosinase, an enzyme crucial to melanin synthesis. As Tyrosinase is unique to the melanocytes this provides a facilitative modem for distinguishing the melanocyte from non-melanocytic cells.<sup>124</sup>

Phenolic prodrugs modelled on tyrosine **30** (see Scheme 2), the natural substrate of the enzyme can be accepted by Tyrosinase. However, catecholic compounds were removed promptly from drug design programmes as toxicity arose from oxidation via a non-enzymatic route.<sup>125</sup> The lead compound to emerge from this area of malignant melanoma drug design which specifically targets Tyrosinase was compound **35**.<sup>126,127</sup>



#### 2.9.2 THE JOURNEY OF DISCOVERY TO PARENT DRUG 35

The ground breaking progress in the search for new melanocytic agents was made by Miura and co-workers who prepared a series of phenolic thioether compounds.<sup>128</sup> The novel introduction of sulfur aimed to increase lipophilicity. In this early work the most potent compound **36** was found to possess significant *in vivo* and *in vitro* antimelanoma activity.<sup>128</sup> However Padgette and co-workers reported<sup>129</sup> that phenylaminoethyl sulfides including compound **36** were substrates for monoamine oxidase (MAO) and the action of this enzyme produced the corresponding aldehyde. Further to these findings it was later revealed that the potent melanocytic activity of compound **36** was completely lost on inhibition of MAO.<sup>130</sup>



To circumvent this catabolic enzyme's action, the preliminary lead compound **36** was derivatised and it was from this work that the *N*-acetyl derivative **35** emerged displaying marked improvements over previous compounds.<sup>126</sup> This compound displayed *in vivo* melanocytic behaviour which included remarkably high depigmentation of black mice hair follicles while *in vitro* inhibition of B16F10 murine melanoma cell lines was equally as successful.<sup>127</sup>

Alena and co-workers furthered this study<sup>127</sup> by combining lead compound **35** with buthionine sulfoximine (BSO) **37** whereupon enhanced *in vitro* and *in vivo* antimelanoma effects of the lead compound were found. BSO facilitates this improved activity by depleting melanocytes of glutathione, a component involved in the pathway of phaeomelanin synthesis.



#### 2.9.3 PHENOLIC PRODRUG MECHANISM OF ACTION<sup>131</sup>

Within the melanocytic cells the targeted enzyme Tyrosinase catalyses the conversion of inactive phenol prodrug **38** into the corresponding *o*-quinone **39**. These active oxygenated species mediate cytotoxicity and impair the melanocytes' metabolism by combining with sulfhydryl containing entities **40** such as DNA polymerase or membrane ion pumps which are crucial to the cell cycle. Alternatively the *o*-quinone may participate in a cyclic redox process whereupon active oxygen species **41** are produced.



#### 2.9.4 THE RESEARCH STORY SO FAR

The significant *in vivo* antimelanoma activity exhibited by compound **35** inspired the beginning of a comprehensive drug design and synthesis programme within the Robins research group. Preliminary investigative work

was performed by McKeown<sup>133</sup> followed immediately by the extensive research of Lant.<sup>134</sup> These workers modified systematically the structure of lead compound **35** in an attempt to reveal the pharmacophoric moieties of the drug. The detailed studies of McKeown and Lant provided a wealth of compounds which can be represented in the summary below (Diagram 1).



#### 2.9.5 PROPOSED WORK

A major structural feature of the parent drug which remained untouched throughout the before-mentioned investigations was the secondary amide group functionality. One programme of structural modifications proposed for this project involved the production of a series of tertiary amide derivatives of the parent drug. The series **42** would also incorporate variations in the amide alkyl group some of which had previously showed enhanced activity.<sup>134</sup> Through this modification of two structural features the series of compounds **42** were anticipated at the very least to improve the lipophilic nature of the drug, remove the hydrogen bonding possibilities of the amide group and furthermore enhance the antimelanoma activity.



This programme of research also included two additional structural attacks on the parent compound. The first area of this additional work involved the synthesis of a series of amidine salts **43**. These compounds introduced a novel structural feature to the antimelanoma agents with this amidine group. The effect of this more water-soluble moiety on activity was hoped to be beneficial with enhanced biological activity anticipated. The third area of work was concerned with the development of a range of tyrphostins **44**. These phenolic compounds have already shown interesting antiproliferative activity<sup>136</sup> and possess similarity to the structure of the substrates of Tyrosinase.



Following the completion of each of the three programmes of research, the *in vitro* melanocytotoxicity of the target compounds and their potential as Tyrosinase substrates would be investigated. The results from these biological studies will then enable the test compounds' potential as antimelanoma agents to be assessed. Concurrently the hypothesis behind the design of the three compounds series' will also be explored. The synthesis and biological evaluation of these potential antimelanoma agents is discussed in Chapter 4.

# **CHAPTER 3**

# CELL SIGNALLING AND MAP KINASE INHIBITORS

#### 3.1 PROTEIN KINASES AND CELL SIGNALLING PATHWAYS

The successful transduction of signals from the cell surface to the nucleus is crucial to cellular function. These signals are transferred via a complicated sequence of events within pathways termed cell signalling pathways.<sup>137</sup> Protein kinases play a vital regulatory role within these transduction pathways whereby they phosphorylate target proteins at specific sites in the pathways. The importance of protein kinases and their protein phosphorylating function to the cell is exemplified by the number of protein kinase genes present in human genetic makeup. It is estimated that as many as 2000 protein kinase genes are possessed by man and the protein kinases are considered one of the largest families of proteins and genes.<sup>138</sup>

#### **3.1.1 PROTEIN KINASE SUPERFAMILY**

Over the past 20 years the discovery of new protein kinases has progressed at such an incredibly fast rate that these protein kinases are now described as a superfamily.<sup>139,140</sup> The rate of discovery was a direct result of implementing developing molecular cloning techniques and the realisation that many oncogenes encode protein kinases.<sup>139</sup> The members of the superfamily are related through structure and function. There is a common structural feature present in the catalytic kinase region of the protein where approximately 250 to 350 amino acids are found. The three dimensional catalytic core results from the folding of the 12 subdomains of the protein.<sup>140</sup>

Consider the phosphorylation function common to the superfamily of kinases (Scheme 4). This involves the enzymatic-catalysed transfer of the  $\gamma$ -phosphate of ATP to the alcohol (serine/threonine) or phenol (tyrosine) group of the protein **45** whereupon the phosphate monoester **46** and ADP are formed.<sup>140</sup>



Although the superfamily proteins share similar three dimensional core structures and phosphorylate via a common mechanism, two main subdivisions of the family exist: the protein serine/threonine (ser/thr) kinases and the protein tyrosine (tyr) kinases.<sup>139</sup> A method for distinguishing the two protein groups from one another exists in examination of the major autophosphorylation site in the catalytic region. Discovery of a tyrosine residue in the presence of acidic amino acids identifies a protein tyrosine kinase.<sup>139</sup>

The detailed discussion from this point will concentrate on the protein serine/threonine kinases with particular emphasis on one member of this family, the mitogen-activated protein (MAP) kinases.

## 3.1.2 MITOGEN-ACTIVATED SER/THR KINASE PATHWAYS

An area of ongoing research and great interest involves the cell signalling pathways in which mitogen-activated (ser/thr) kinases participate. Insulin, growth factors and cytokines, the mediators of these pathways,<sup>142,143</sup> bind to specific receptors on the cell surface. These receptors are transmembrane tyrosine kinases or are associated with intracellular tyrosine kinases.<sup>143</sup> Binding produces tyrosine phosphorylation of the receptors which triggers the intracellular signalling pathway. In the cascade of events which follows many ser/thr proteins kinases are activated and a veritable flood of

ser/thr phosphorylation events ensues.<sup>143</sup> Mitogen-activated protein (MAP) kinases are a family of ser/thr kinases involved in such pathways.

# **3.2 MAP KINASE FAMILY**

## **3.2.1 PIONEERING WORK**

The pioneering work in the MAP kinase field of research was carried out by Sturgill and Ray<sup>144</sup> who were investigating insulin-promoted ser/thr phosphorylation of polypeptides. They employed a protein phosphatase-1 regulatory protein, Inhibitor-2 (I-2), as a substrate to probe for kinase activity in insulin-treated 3T3-LI cells. A ser/thr phosphorylated polypeptide extracted from the I-2 isolate was identified as microtubule-associated protein-2 (MAP-2).<sup>144</sup> The MAP-2 kinase then isolated from the insulin-treated 3T3-LI cells was identified as the insulin-promoted ser/thr phosphorylating enzyme.<sup>144</sup>

These early observations of Sturgill and Ray generated further interest with two observations. Firstly via a collaboration with Erikson and Maller<sup>145</sup> they demonstrated phosphorylation and activation of Rsk enzymes S6 kinase II by MAP-2 kinase and secondly that MAP-2 kinase activation required both tyrosine and threonine residues on the MAP-2 kinase polypeptide to be phosphorylated.<sup>145</sup>

This pioneering work discovered MAP-2 kinase, which was later renamed mitogen-activated protein (MAP) kinase.<sup>146</sup> MAP kinase is now recognised to play a vital role in cell signalling by converting tyrosine phosphorylation into serine/threonine phosphorylation essential for regulation of pathways within the cell.<sup>147</sup>

## 3.2.2 MAP KINASES IN YEAST

The role of MAP kinases in yeast cell signalling pathways has been well explored with many pathways elucidated. It is believed that the distinct similarities in mammalian and yeast MAP kinases should enable the application of the understanding of yeast pathways to those in mammals.<sup>148</sup>

Yeast MAP kinases are known to participate in at least four distinct pathways. For example, in the mating pathway of *Saccharomyces cerevisiae*, two gene products FUS3<sup>149</sup> and KSSI<sup>150</sup> are kinases which function at G<sub>1</sub> in

the yeast cell cycle thereby arresting growth. Such are the similarities in mammalian MAP kinases and yeast kinases that sequences encoding newly identified mammalian kinases are compared to those of known yeast kinases.<sup>148</sup>

#### 3.2.3 MAP KINASE SUBFAMILIES

Boulton and co-workers dominated the discovery of extracellular signalregulated kinases (ERKs), an insulin-activated MAP kinase subfamily.<sup>151</sup> ERK-1 was the first ERK family member to be purified and cloned. This 44kDa insulin-activated MAP kinase was found to possess remarkable homology (56%) to the FUS3 and KSS1 yeast genes. Using the same molecular cloning techniques two other ERK subfamily members were discovered, ERK-2 and ERK-3.<sup>152</sup> ERK-2 was found to correspond to the original kinase isolated by Sturgill and Ray (MAP-2 kinase). ERK-2 possessed the same identity with the yeast genes (56%) as ERK-1. However, ERK-3 was found to possess homology to a lower extent (37%) thereby implicating ERK-3 as belonging to a separate subfamily.<sup>152</sup>

A second MAP kinase subfamily was uncovered by Kyriakis and Arruch.<sup>153</sup> The p54 SAP kinase was purified from rat liver following cycloheximide injection. This MAP kinase subfamily also exhibits homology with FUS3 and KSSI (42%). The p54 SAP and ERK subfamilies are all highly expressed in the mammalian brain. This along with the homology the subfamilies exhibit towards the MAP kinases involved in the arrest of yeast cell growth indicates the potential role played by MAP kinases in post mitotic cells. The MAP kinases discussed here can be divided into 3 main subfamilies of ERK-I and ERK-2 in the first subfamily and the other two subfamilies composed of ERK-3 and the p54 SAP kinases.<sup>154</sup> However, such is the pace of discovery in this area that novel subfamilies will continue to emerge.

## **3.3 MAP KINASE PATHWAYS**

As is the case for many signalling pathways, the prime concern for the MAP kinase pathways is to transport the extracellular signal successfully to the predetermined destination within the cell which more often than most is the nucleus of the cell.<sup>148</sup> The cellular instructions contained within the signal must be conserved throughout the biochemical cascade of events. In MAP kinase pathways where the signal instructions are directed towards transcription of

specific genes, expression of these genes controls a variety of cell functions which include proliferation.<sup>155</sup>

Fundamental to the MAP kinase (MAPK) pathways is a functional triad of serine/threonine protein kinases: MAPK is activated through threonine/tyrosine phosphorylation by MAPKK, a dual-specificity protein kinase which is in turn activated via serine phosphorylation by MAPKKK, the third member of the MAPK pathway core.<sup>147</sup>

Four distinct MAP kinase pathways have been identified in yeast and three in mammals but progress is such that more are destined to be uncovered. Perhaps the best way to discuss the mammalian MAP kinase pathways is through an example using the most well-known ras/ERK MAPK pathway (see Diagram 2).

## 3.3.1 RAS/ERK MAP KINASE COMPONENTS

Within the known ras/ERK MAPK pathway the components corresponding to the functional triad are ERK (MAPK), MEK (MAPKK) and raf (MAPKKK).<sup>147</sup> Each of these components will now be discussed as well as ras which functions as the link between the extracellular signal and the MAPK cascade (Diagram 2).



#### 3.3.1.1 MEK

The entity controlling activation of MAP kinase (later known to be a MAP kinase kinase) was first discovered by Ann and co-workers.<sup>157</sup> They revealed that this activator could reactivate ERK-1 and ERK-2 MAP kinases previously inactivated by treatment with phosphatase. Further to these findings Gomez and Cohen<sup>158</sup> reported that MAP kinase kinase (MKK) required serine/threonine phosphorylation to function and later the ERK-specific MKK gene was identified and named *mek* (MAP/ERK kinase).<sup>159</sup> The dual specific nature of MEK unfolded from the discovery that phosphorylation of both tyrosine and threonine residues is required for activation of the ERK MAP kinases (ERK-1 and ERK-2).<sup>160</sup>

#### 3.3.1.2 Raf

The immediate upstream activator of MEK in the ras/ERK pathway is raf. This kinase has been shown to activate MEK by phosphorylation of two adjacent serine/threonine (ser/thr) residues, an activation mechanism consistent to MAPKKs alike.<sup>161</sup> Raf was identified as the upstream kinase of MEK and has been likened to protein kinase C (PKC), another ser/thr kinase.<sup>162</sup> Although structural and operational similarities exist between the two kinases, the operational aspects for PKC are understood to a far greater extent. Raf is believed to be regulated by multiple mechanisms of control.<sup>162</sup>

#### 3.3.1.3 Ras

Ras is a small (21 kDa) guanine nucleotide binding protein.<sup>163</sup> This Gprotein plays a pivotal role in the ras/ERK pathways by linking receptor tyrosine kinase activation to the intracellular MAP kinase pathway (see Diagram 2). Ras is positioned immediately upstream of raf in this pathway and is only active in a GTP-bound state as guanine nucleotide binding controls the activity of the Gprotein.<sup>164</sup> However, raf is not directly activated by binding to the GTP-bound ras. It is proposed that raf bound to ras positions raf at the membrane, whereupon an additional as yet undefined event follows resulting in raf kinase activation.<sup>163</sup>

#### 3.3.1.4 ERKs

MAPK is the effector of the MAPK pathway and many MAP kinases have the potential to regulate gene expression by targeting transcription factors in the final phosphorylation of the pathway.

In the ras/ERK pathway the transcription factor target is TCF (ternary complex factor) in particular the SAP-1 and Elk-1 family members.<sup>165</sup> Found within the nucleus the ERK-1 and ERK-2 MAP kinases activate these TCFs by phosphorylating ser/thr residues on the SAP-1 and Elk-1. This phosphorylation is reportedly crucial to stimulation of the c-fos gene transcription process which follows.<sup>165</sup> The ras/ERK MAPK pathway represents how a signal produced from extracellular mitogen-activation of a tyrosine kinase receptor is then involved in a ser/thr cascade of intracellular phosphorylation events to deliver the signal to its destination in the nucleus where the signal's instructions can be implemented.<sup>155</sup>

#### 3.3.2 JNK/SAPK MAPK PATHWAY

The predeterminant for the activation of the ERKs (ERK-1 and -2) and p54 SAP MAP kinases is dual ser/thr and tyr phosphorylation.<sup>166</sup> However, the modulator for this process is quite different for each of the two MAPK subfamilies. The ERKs' activator MEK was reported as unable to activate bacterially expressed p54 and in a variety of cell lines, the mitogenic activators of the ERKs which act via ras were functionally redundant with the p54 MAPK.<sup>166</sup> However, in response to cellular stress the p54 kinases were strongly activated and hence the name stress-activated protein kinases (SAPK) was introduced. Other SAPK activators include heat-shock, ultraviolet light and tumour necrosis factor (TNf- $\alpha$ ). It is believed that this activation is not coupled to ras and the MAPK pathway.<sup>166</sup>

The SAPKs target the transcription factor c-jun and phosphorylate ser63 and ser73 in the amino-terminal activation domain. SAPKs stimuli were shown to stimulate phosphorylation and activation of c-jun and the SAPKs are believed to be the main c-jun kinases (JNK).<sup>166</sup>

In summary, this JNK/SAPK pathway represents a distinct MAPK cell signalling pathway. Activated in response to stress stimuli the pathway targets

proliferation.<sup>166,167</sup> However, as yet not all of the upstream components of the cascade have been identified but of interest are the similarities found in yeast reported with the stress-activated HOG1 MAP kinase pathway.<sup>167</sup>

## 3.4 MAPK PATHWAYS AND CANCER

A great many genes have been implicated in the malignant transformation of cells leading to the development of cancer.<sup>20</sup> A large number of these genes are oncogenes or cancer-causing genes as they are sometimes called. These oncogenes facilitate the aberration of normal cell behaviour by disturbing the regulatory systems central to the cell, e.g. the cell signalling pathways.<sup>20</sup> More than 20 oncogenes have been identified and these arise from mutation, amplification or altered expression of normal genes called protooncogenes found within the cell.<sup>19</sup> The protein products encoded by the protooncogenes are components of cell proliferation and differentiation processes such as growth factors, growth factor receptors, signal transduction machinery and transcription factors.<sup>20,168</sup> Therefore, in theory aberration can occur potentially at many points along the pathway.

For the MAPK pathways oncogenic components identified include the protein products encoded by the ras, raf, jun and fos oncogenes.<sup>19</sup> These transformed genes facilitate the malignant transformations of cells which employ the deregulated MAPK signalling pathways.

The G-protein ras is a key player in MAPK pathways especially the ras/ERK MAPK pathway. The ras oncogenes were discovered following lengthy research into the existence of dominant oncogenes in human tumours.<sup>169</sup> The first such gene transfer assays were performed by Harvey<sup>170</sup> and Kirsten<sup>171</sup> on strains of rat sarcoma viruses. In human cancers ras oncogenes are the most prevalent family found and about 10% of the most common human cancers contain these transformed ras genes.<sup>172</sup> The genetic transformation occurs via a single point mutation within the phosphate binding region, whereupon the mutant ras protein is locked in the activated GTP-bound state. Loss of G-protein control results in continued signal transduction and ultimately uncontrolled cell growth.<sup>172</sup>

A further oncogenic MAPK pathway component is mutated raf protein.<sup>173</sup> In normal raf the N-terminal domain of the protein functions to suppress kinase activity.<sup>174</sup> Transformation of the gene involves the deletion of this domain, whereupon high levels of deregulated kinase activity are exhibited by the oncogene raf.<sup>173</sup>

This very brief outline, depicts the powerful role played by genetics in cell function regulated by enzymes such as MAPK. Identification of the components in such signal transduction pathways and the genes encoding these functional entities is the key to a thorough understanding of cell proliferation in normal and malignant states. Such a wealth of information is pivotal to the development of targets for anti cancer treatment.

# **3.5 MAPK INHIBITION**

For the protein kinase superfamily the protein tyrosine kinases dominate the area of research devoted to inhibitors. During the past 5 years there has been a veritable explosion in this area with drastic improvements in the potency and specificity of protein tyrosine kinase inhibitors reported.<sup>175</sup> However, the situation for protein serine/threonine kinase inhibitors is not quite as aweinspiring. While some members of this family such as protein kinase C (PKC) have been studied in depth with resultant potent and selective inhibitors reported,<sup>176,177</sup> the MAP kinase family remains untouched with respect to successful inhibition of members of this enzyme group. Encouraging reports however have involved inhibition of other member of the MAPK pathway such as MEK, the upstream kinase activator of MAP kinase.<sup>177,178</sup>

At present progress in the area of cell signalling is so rapid that revelations regarding MAPK inhibitors must be imminent. With biological probes developing and understanding of cellular pathways improving, interest in this field of research continues apace.

# 3.6 MODEL FOR MAPK INHIBITORS IN THIS WORK

Very limited work has been carried out within the Robins research group in the area of MAPK inhibitors. However, a preliminary biological screen<sup>179</sup> of a series of compounds prepared provided encouraging results with a lead compound identified possessing potential for further work in the development of novel MAP kinase inhibitors.

This compound **47** was modelled on serine **48**, an amino acid phosphorylated on MAPK substrates by the enzyme's action. The structure is a

conformationally restricted form of serine incorporating an electron rich benzene ring. Furthermore, it has been reported that such compounds exist in an iminolactone form **49** rather than the open form of **47**<sup>180</sup> and require careful handling due to the sensitive nature of these compounds.<sup>181,182</sup>



The research proposal to develop the lead compound **49** within this project involved preparation of three sets of compounds.

- 1 A series of iminolactones **50** derived from **49** would investigate ring substitution position.
- 2 A series of compounds fixed in the open form **51** might enhance the serine-like moiety within the structure further.



3 The iminolactones **50** would be employed as starting materials for the facile conversion into the corresponding lactones **52**. These compounds are coumarins, a family of compounds known to possess a variety of interesting biological activities.<sup>183</sup>

On completing these synthetic targets, the three sets of compounds would be biologically evaluated for MAP kinase inhibitory potential. The synthesis and biological test results are discussed in Chapter 5.

# **CHAPTER** 4

# SYNTHESIS AND BIOLOGICAL EVALUATION OF ANTIMELANOMA AGENTS

## **4.1 PHENOLIC THIOETHER TARGET COMPOUNDS**

#### 4.1.1 OXAZOLINES<sup>184</sup>

The century-old 2-oxazoline ring system<sup>185</sup> **53** has evolved well beyond its established origin as a masked carboxylic acid moiety and the stability of this heterocyclic compound is such that it remains resistant to most reagents with the exception of mineral and Lewis acids.<sup>186</sup> Indeed, the 2-oxazolines **53** provide synthetic precursors for various other heterocycles<sup>187</sup> including oxazolidines **54** and oxazoles **55**.



The functionality of the 2-oxazoline and its derivatives enables these compounds to penetrate into diverse fields of interest from pharmaceuticals and agriculture to protective coatings and plasticizers.<sup>187</sup> However, to the synthetic organic chemist the 2-oxazoline ring provides a system with potential for modifications at the 2-, 4- and 5-positions. With these levels of structural variety possible the resultant chiral and achiral 2-oxazolines possess a whole host of synthetic uses<sup>184,188</sup> which include applications as protecting groups, analytical probes and monomers in polymerisation processes. This synthetic utility of 2-oxazolines extends to the preparation of conformationally constrained peptides which are of increasing importance to the medicinal chemist. Moreover, optically active oxazolines enjoy participation in asymmetric syntheses and play the role of recoverable ligands in asymmetric catalytic processes.<sup>184</sup>

The variety of the structural modifications possible on the 2-oxazoline ring system is reflected in the wealth of synthetic routes reported alone for these compounds.<sup>184</sup> However, as the 2-oxazolines involved in our work are a simple form featuring substitution at the 2-position only, the synthesis of these 2-oxazolines is very straightforward. Therefore the synthesis of 2-oxazolines detailed in section 4.1.1.1 is not to be considered representative of the synthetic area as a whole.

#### 4.1.1.1 Synthesis of oxazolines

The 2-oxazolines **56** were prepared in a single step catalytic procedure<sup>189</sup> from ethanolamine **57** and the appropriate nitrile **58**. Following fractional distillation of the solvent-free reaction mixture, the products were obtained in moderate yields (see scheme 5). This cadmium acetate-catalysed synthesis was found to be of particular use in our work with its application extended to the preparation of oxazines required for application in section 4.2. For the oxazine series (**56**, n=2) rather than ethanolamine, **3**-amino-1-propanol was used which maintained the continued use of cheap, readily available materials. The structures of these known compounds **59** to **64** were confirmed by the spectroscopic data collected. This information together with the boiling points recorded were found to be in agreement with the findings reported in the literature.

R-CN + <b>58</b>	НО <sup>^^</sup> ( 57	NH <sub>2</sub> Cd	(OAc) <sub>2</sub> .2H <sub>2</sub> C 130 °C, 1.5 - 29 - 49%	) (cat.) 7 d	(∖)n O_N R 56			
	n	R	Product	Yield				
	1	Et <i>cyclo</i> C <sub>6</sub> H <sub>11</sub>	59 60	34% 42%				
	2	Me Et <i>cyclo</i> C <sub>6</sub> H <sub>11</sub> Ph	61 62 63 64	49% 43% 37% 29%				
Scheme 5								

A mechanism is proposed<sup>134</sup> for this catalytic preparation of 2oxazolines and oxazoles for a general case in scheme 6. Notice that although cadmium acetate was our catalyst of choice, soluble heavy-metal salts including zinc acetate and zinc chloride are also recommended.<sup>189</sup>



#### 4.1.1.2 Reactivity of 2-oxazolines

For our purposes the 2-oxazolines were employed as an electrophilic source poised for ring opening attack by selected nucleophilic sources. The chemical nature of this heterocyclic species is such that the 2- and 5-ring positions are potentially most susceptible to nucleophilic attack. Fazio confirmed this with spectroscopic investigations he performed on the neutral **59** and cationic **65** forms of 2-ethyl-2-oxazoline.<sup>190</sup> The oxazolinium cation formed by protonation in ring opening reactions was created in the study through Lewis acid coordination. In table 1 the <sup>13</sup>C NMR values found for the two ethyl-2-oxazoline species can be seen. The deshielding of the 2- and 5-ring carbons is very clear and identifies the electrophilicity of these two positions. This

therefore confirms the 2- and 5-ring oxazoline positions as the potential sites for attack by an incoming nucleophile.



Consider the 2-oxazoline ring opening reaction presented in a generalised form below.



The oxazolidine **54** and carboxamide **66** are the products of opening the oxazoline ring at the 2- and 5-positions, respectively.<sup>190</sup> The pathway taken, and therefore the product obtained, are controlled by the nucleophilic source involved. Later we illustrate this effect with the use of aryl thiols and primary aryl amine salts - two very different nucleophiles.

#### 4.1.2 SYNTHESIS OF THE PHENOLIC THIOETHER TARGET COMPOUNDS

The strategy behind our melanoma drug design together with the three programmes of research we proposed are outlined in section 2.9. The first of the three programmes of research involved the preparation of a series of phenolic thioether compounds which incorporated a tertiary amide into the alkyl side arm. However, before embarking on the synthesis of this first set of target compounds, the simple parent compounds **67** from which they originate had to be prepared. The parent compounds were to be included in the biological studies performed on the target compounds to provide base values for comparison purposes. In fact, one of these compounds, **35**, is the lead drug in the antimelanoma field on which our compounds are modelled.

#### 4.1.2.1 The Wehrmeister reaction

The Wehrmeister reaction<sup>191</sup> was employed to prepare the two parent compounds **35** and **68** and this reaction is an example of the exclusive opening of the 2-oxazoline ring system at the 5-position (discussed in section 4.1.1.2). Under neat heating at reflux and catalytic conditions, *p*-hydroxybenzenethiol **69** and 2-oxazoline **53** were combined to afford the carboxamide **67**. Notice the nucleophilic supremacy demonstrated by the thiol over the phenol within the *p*-hydroxythiophenol reagent **69**. The Wehrmeister reaction will play a pivotal role in the three step synthesis of the phenolic thioether target compounds in section 4.1.2.2.



The reaction was found to proceed smoothly with both products obtained in high yields. In the preparation of compound **68**, the 2-cyclohexyl-2-oxazoline **60** was provided by the cadmium acetate catalysed reaction<sup>189</sup> presented in section 4.1.1.2 while commercially available oxazoline **70** was used to prepare compound **35**. The sharp melting points and spectroscopic data of **35** and **68** were in agreement with the literature findings for these known compounds.

#### 4.1.2.2 Overall route to general phenolic thioether target compounds

The series of phenolic thioether compounds **42** were of a structurally similar nature which enabled the devising of one scheme for the preparation of all nine target compounds. In the three step synthesis, the first step provided the basic starting compounds **71** which would be transformed over the following two steps. Each step will be discussed in turn.



#### 4.1.2.2.1 Step1: Oxazoline ring-opening

The first step was crucial to the synthesis as the products **71** incorporate the basic structural skeleton required for the final target compounds. The oxazoline ring-opening Wehrmeister reaction<sup>191</sup> was employed with the

nucleophilic source provided by 4-methoxybenezenethiol **72**. As expected the reaction furnished the carboxamide product **71**.



The reaction produced compounds **73**, **74** and **75** in high yields, with sharp melting points and analytical purity. Spectral confirmation of the products' formation centred around the amide functionality with IR observations including characteristic absorption frequencies in the 1630-1645 cm<sup>-1</sup> region for the carbonyl bond and the NH stretching vibrations at *ca.* 3300 cm<sup>-1</sup>. The <sup>13</sup>C NMR spectra supported these findings with the carbonyl carbon signal at *ca.*  $\delta$  170. As expected the *p*-disubstituted benzene ring produced <sup>1</sup>H NMR signals characteristic of an AA'BB' system at *ca.*  $\delta$  6.8 and 7.3. The corresponding signals on <sup>13</sup>C NMR spectra resonated at *ca.*  $\delta$  115 and 134.

#### 4.1.2.2.2 Step 2: N-Alkylation of secondary amides

The straightforward conversion<sup>192</sup> of secondary amide **71** into the corresponding tertiary amide **77** was facilitated using sodium hydride and three alkylating sources; methyl iodide, ethyl bromide and benzyl bromide. As a result, nine products were obtained from the three starting compounds. Preparative attempts using isopropyl bromide as an alkylator proved to be unsuccessful even under forcing conditions. This result was attributed to the hindered nature of the secondary alkyl halide. On completion of the successful reactions the sodium halide side product was removed by filtration from the

reaction mixture and the solution worked-up in a standard manner<sup>192</sup> to yield the crude amide. Following purification by column chromatography, the nine compounds were isolated as oils in a range of yields (mean value 60%) and the structures were confirmed by full spectroscopic characterisation. A discussion of the most pertinent features of these spectroscopic results will follow.



NMR spectroscopy provided the data of greatest interest. However, the IR spectra are deserving of a mention as the successful conversion of secondary into tertiary amide was confirmed by the absence of any absorption frequencies for the amide NH bonds. The interest in the NMR spectra arose from the doubling of signals observed on both the <sup>1</sup>H and <sup>13</sup>C NMR spectra thus indicating that the tertiary amides exist in rotameric forms. Integration of peak heights on the <sup>1</sup>H NMR spectra enabled the rotameric ratio for each compound to be established. A range of ratios for the minor:major rotamer signals from 1:1.2 to 1:1.5 for the nine compounds was calculated. The <sup>1</sup>H and <sup>13</sup>C NMR spectra of compound **81** were chosen to exemplify this NMR phenomenon exhibited by the tertiary amide series (see diagrams 3 and 4).





From the <sup>1</sup>H NMR spectrum of compound **81** (diagram 3) a ratio of 1:1.2 for the minor:major rotamers was established. The signals for the cyclohexyl moiety were evident in the upfield region  $\delta$  0.90 to 2.34. In particular the methine proton produced a pair of triplet of triplets at  $\delta$  2.00 and 2.34. Notice the coupling constant values of 11.6 and 3.3 Hz for each signal consistent with *cis*- and *trans*-vicinal coupling expected for such a six-membered saturated ring system. This <sup>1</sup>H NMR spectrum demonstrates pairing of signals with beautiful clarity such that even the pairs of rotameric signals for the AA'BB' system in the downfield region  $\delta$  6.76-7.33 show distinct separation. The <sup>13</sup>C NMR spectrum (diagram 4) maintains this illustrative pleasure by exhibiting a paired signal for the deshielded carbonyl carbon at  $\delta$  176.4 and 176.5, a feature not preserved throughout the nine compound series.

The three phenyl compounds **84**, **85** and **86** were found to produce broad pairs of signals in their <sup>1</sup>H NMR spectra. This broadening of signals may be due to an electronic effect with perhaps conjugation of the phenyl ring with the amide causing the change in amide bond character.

#### 4.1.2.2.3 Step 3: Methyl ether deprotection

The final step in the three step synthesis of the phenolic thioether target compounds **42** employed a boron tribromide-mediated reaction to deprotect the methyl phenyl ethers **77** (see scheme 12). Preliminary investigations into reagents for this synthetic transformation included the chlorotrimethyl silane/sodium iodide reagent<sup>193</sup> and boron trichloride.<sup>194</sup> The former proved rather disappointing with very poor yields obtained and although boron trichloride was a successful reagent its bromide counterpart proved to be the better boron-based Lewis acid for the ether cleavage.

QМе				он
$\begin{vmatrix} & & \\ & $	i. `R	BBr <sub>3</sub> (1M, 4 DCM, –78 ° ii. –78 °C→ 43 - 72%	eq.) C rt	$ \begin{array}{c}                                     $
R	R <sup>1</sup>	Substrate	Product	Yield
Me Me Me	Me Et Bn	78 79 80	87 88 89	77% 60% 76%
<i>cyclo</i> C <sub>6</sub> H <sub>11</sub> <i>cyclo</i> C <sub>6</sub> H <sub>11</sub> <i>cyclo</i> C <sub>6</sub> H <sub>11</sub>	Me Et Bn	81 82 83	90 91 92	68% 43% 53%
Ph Ph Ph	Me Et Bn	84 85 86	93 94 95	88% 53% 72%
		Scheme	12	

For all nine compounds the low temperature (-78 °C) deprotection proceeded in a satisfactory manner. Following an ice/water-mediated quench of the reaction, standard work-up conditions<sup>194</sup> were employed. Eight of the nine compounds required purification by column chromatography and all of the purified products were obtained in moderate to high yields with a mean value of 66%. The phenolic compounds provided sharp melting points and were fully characterised.

The successful deprotection of most phenyl ethers is a transformation easily followed by spectroscopic analysis and this was found to be the case here with the phenol group providing the most important feature in the IR spectra. Characteristic OH absorption frequencies were found in the 3400-2500 cm<sup>-1</sup> region and at *ca.* 1270 cm<sup>-1</sup> for the respective stretching and bending OH vibrational modes. Trade mark phenolic behaviour also prevailed in the UV spectra with the main absorption band common to all nine compounds found at *ca.*  $\lambda$  253 nm ( $\epsilon$  6000-10120). This UV activity was in agreement with the findings reported by Lant<sup>134</sup> for compounds containing the same chromophore.

NMR spectra produced the rotameric observations expected for these tertiary amides. The NMR of phenol **90** are included here (diagrams 5 and 6).





These spectra provide another demonstration of very clearly paired signals. Consider the <sup>1</sup>H NMR spectrum (diagram 5); the successful deprotection is quite apparent through the absence of the methyl ether signal at  $\delta$  3.6-3.8 while the phenol produces a pair of broad singlets in the characteristic downfield region  $\delta$  7.8-9.1. However, it should be noted that such paired phenol signals are not a feature conserved throughout the series of nine target compounds. The <sup>13</sup>C NMR spectrum (diagram 6) **90** was found to demonstrate the same evidence for the deprotection through the absence of the methyl ether signal at *ca*.  $\delta$  55. A continuation of this exploration into the NMR behaviour of these rotameric compounds found changes in the rotamer ratios occurred on deprotecting the phenol. For example, a 1.3 fold increase in the ratio of minor to major rotamers is found on deprotecting **81** to afford **90**.

# 4.1.3 BIOLOGICAL EVALUATION OF PHENOLIC THIOETHER COMPOUNDS

#### 4.1.3.1 In vitro 96 hour sulforhodamine growth inhibition assay

The first of two investigations into the biological activity of the phenolic thioether compounds was performed by Dr Lloyd Kelland of The Institute of Cancer Research, Surrey.<sup>195</sup> In this *in vitro* evaluation the test compounds were incubated in a 96 hour sulforhodamine B (SRB) growth inhibition assay where the inhibitory activities of the compounds were assessed against six melanoma cell lines (B008, B0010, G361, HT144, SKMeL24 and SKMeL2) and an ovarian cancer cell line (SKOV-3) which acted as the non-melanoma control. The SKMeL24 is a specialised melanoma cell line which lacks the biochemical machinery required to convert the phenolic prodrugs into the corresponding ortho-quinones. This non melanotic cell line provides an indication of cytotoxicity produced by a mechanism of action independent of Tyrosinase. Since we designed our compounds to target Tyrosinase this specific cell line was an important feature of the assay. The cytotoxicity of the compound was assessed from the size of the GI<sub>50</sub> value quoted for the assay. The GI<sub>50</sub> value is the concentration of drug which causes 50% inhibition of cell growth. Ideally we required a test compound to demonstrate cytotoxicity and selectivity for melanoma cell lines containing Tyrosinase. To do so we wanted low GI50 values for the five Tyrosinase-containing melanoma cell lines together with high values for the non melanotic and ovarian cancer cell lines.

#### 4.1.3.1.1 Results

As we discussed earlier in section 4.1.2 the parent N-H compounds **35**, **68** and **96** were included in the assay to provide comparison results. The assay results for all the test compounds can be seen in the following three tables which house separately the results for the methyl, cyclohexyl and phenyl analogues prepared. Compound **96** was kindly supplied by Lant.<sup>134</sup>



Table 2 GI<sub>50</sub> values ( $\mu$ M) for compounds **42** (**R** = **Me**) in 96 h SRB assay

		Cell line						
Cmpd	R1	B008	B0010	G361	HT144	SKMeL2	SKMeL24	SKOV-3
35	н	>100	>100	>100	>100	>100	>100	>100
87	Me	78	>100	>100	>100	>100	>100	>100
88	Et	33.5	>100	70	92	92	>100	>100
89	Bn	5	30	12	17	17.5	>100	>100

Table 3 GI\_{50} values ( $\mu$ M) for compounds **42** (**R** = *cyclo***C**<sub>6</sub>**H**<sub>11</sub>) in 96 h SRB assay

		Cell line							
Cmpd	R <sup>1</sup>	B008	B0010	G361	HT144	SKMeL2	SKMeL24	SKOV-3	
68	Н	29	>100	77	>100	61	>100	>100	
90	Me	7.8	41	90	13.5	14.5	>100	51	
91	Et	14.5	74	39	53	49	>100	>100	
92	Bn	14	24.5	14	20	32	>100	53	

		Cell line							
Cmpd	R1	B008	B0010	G361	HT144	SKMeL2	SKMeL24	SKOV-3	
96	Н	>100	>100	>100	52	>100	>100	>100	
93	Ме	34	110*	69	93	105*	>100	>100	
94	Et	24	85	43	63	73	>100	>100	
95	Bn	16	34	15.5	30	37	>100	56.5	

Table 4 GI<sub>50</sub> values ( $\mu$ M) for compounds 42 (**R** = **Ph**) in 96 h SRB assay

\* Obtained by extrapolation

results of particular interest

#### 4.1.3.1.2 General observations

In general most of the compounds exhibited improved cytotoxicity over the lead compound **35**. Attention must be drawn to compounds **89**, **91** and **94**, three potentially exciting new compounds. In particular the *N*-benzylmethyl analogue **89** exhibited better selectivity and cytotoxicity than our group has seen before. This compound produced low GI<sub>50</sub> values against the melanotic cell lines and this together with a high GI<sub>50</sub> value for the SKMeL24 cell line demonstrated the good selectivity possessed by compound **89** for the Tyrosinase-containing cell lines. This significant improvement in cytotoxicity and selectivity was facilitated by the simple action of converting the parent drug **35** into the corresponding *N*-benzyl compound **89**. Compounds **91** and **94** also demonstrated similiar selectivity but the cytotoxicity levels were not as high. Selectivity problems were found with cyclohexyl compounds **90** and **92** as well as phenyl compound **95**. Because these three compounds showed low GI<sub>50</sub> values against the SKOV-3 cell line as well as the melanotic ones, selectivity appeared to be lower.

#### 4.1.3.1.3 Conclusions

The lead compound **89** from this assay cannot be disputed. It exhibited good levels of cytotoxicity and some desirable selectivity. This level of biological activity was followed by **91** and **94** but with less potency.

In general *N*-alkylation of the simple parent compounds was found to improve cytotoxicity greatly and in most cases selectivity for the melanoma cell lines. This improved cytotoxicity could be attributed to the lipophilic character of the test compounds. These tertiary amides would be expected to be more
lipophilic than the corresponding secondary amides and as such the lipophilicity may increase drug absorption and therefore distribution in the tissues. However, lipophilicity alone cannot be used to explain the improvement in both cytotoxicity and selectivity observed because these attributes were not common to all the tertiary amide compounds tested. It would appear that some sort of balance is required between the size of the two substituents (R and R<sup>1</sup>) in the compounds found to produce the desirable antimelanoma activity in vitro. Consider the lead compound 89, an N-benzylmethyl derivative. Selectivity was lost when the other N-benzyl compounds 92 and 95 were created. Could this reduced selectivity be attributed to too much steric bulk offsetting the proposed delicate balance between the size of the two substituents required for the success found with compound 89. In addition it was found on replacing the Nbenzyl group in compound 89 with N-ethyl that an overall loss in cytotoxicity and selectivity occurred. Therefore we would propose that the suggested balance between the two substituents could have been disturbed again but this time as a result of too little steric bulk.

#### 4.1.3.2 In vitro mushroom Tyrosinase spectrophotomeric assay

To widen the biological evaluation of our potential antimelanoma compounds a spectrophotomeric enzyme assay<sup>196</sup> was employed to investigate the relative substrate activity of these compounds for the enzyme Tyrosinase. The assay was performed by Miss Isabel Freer of the Tissue Culture Unit, University of Glasgow. As Tyrosinase of mammalian origin was not available to us, commercially available mushroom Tyrosinase was used instead. Like mammalian Tyrosinase, the mushroom enzyme is a copper-containing aerobic oxidase which catalyses the conversion of L-Tyrosine into dopaquinone.<sup>197</sup> However it must be remembered that any observations made and conclusions that may be drawn from the assay are to be used only as a guide to how the test compounds might behave with mammalian Tyrosinase. Nonetheless, we hoped the results would compliment the *in vitro* SRB inhibition assay results (section 4.1.3.1).

The mushroom Tyrosinase assay used UV spectroscopy at wavelength 280 nm to follow the production of *ortho*-quinone from the phenolic test compound. The spectrophotomeric assay measurements enabled us to determine the relative substrate activity of the test compound by comparing the relative activity of the unnatural substrate when it replaced the natural substrate in the assay at the same concentration. Obviously L-tyrosine, the natural

substrate of Tyrosinase, would produce a value of 100% relative substrate activity. The assay results can be seen in table 5.

#### 4.1.3.2.1 Results and discussion



Table 5 Relative substrate activities (%) of compound	42
for mushroom Tyrosinase	

			F	1	
		Н	Ме	Et	Bn
R = Me	Compound	35	87	88	89
% Relative su	bstrate activity	4.5	7.9	17.5	20.7
R = cyclohexyl	Compound	68	90	91	92
% Relative su	bstrate activity	37.8	7.1	16.8	0
R = Ph	Compound	96	93	94	95
% Relative su	bstrate activity	6.4	17.8	28.3	0

results of particular interest

From the results the best substrate appears to be **68** and this compound had showed modest cytotoxity against some cell lines in the SRB inhibitory assay. The *N*-ethyl compounds **88**, **91** and **94** were the next best substrates in the assay. However, **93** and **89** were also found to produce values in this range. These Tyrosinase assay results can be used as an indication that the compounds are substrates for Tyrosinase and may exhibit their cytotoxicity by Tyrosinase-mediated oxidation. Surprising results were found for the *N*-benzyl compounds **92** and **95** with 0% relative substrate activities recorded in the assay, an indication that these compounds may be poor substrates for the enzyme.

#### 4.1.3.3 Correlation of assay results

The results appeared to compliment the *in vitro* assay observations. As was already discussed (section 4.1.3.1) the most successful candidate from the *in vitro* assay was compound **89** followed by **91** and **94**. These compounds

showed good cytotoxicity and selectivity for Tyrosinase-containing cell lines. The Tyrosinase assay results complimented these findings with **89**, **91** and **94** being among the test compounds which showed good values for relative substrate activity, an indication of their potential as Tyrosinase substrates. The inhibitory assay findings had also revealed **90**, **92** and **95** to be cytotoxic but what is more important is that they lacked selectivity. Perhaps this lack of selectivity can be explained by the Tyrosinase assay results which found both **92** and **95** as poor potential substrates for the enzyme with zero values for relative substrate activity recorded.

## 4.1.3.3.1 Conclusions on biological test results

Armed with the results of the two assays we propose that compounds **89**, **91** and **94** are cytotoxic and selective for the Tyrosinase-containing melanoma cell lines and that the cytotoxicity might be the result of Tyrosinase-mediated oxidation of those phenolic test compounds. Furthermore we propose that the lack of selectivity exhibited by compounds **90**, **92** and **95** could be the result of cytotoxicity produced from a non-Tyrosinase mediated mechanism of action.

# 4.2 AMIDINE SALTS TARGET COMPOUNDS

The second programme of research we proposed for the antimelanoma field involved the preparation of a series of amidine salts. This would introduce a novel structural feature to the phenolic thioether-based compounds already prepared. As outlined in section 2.9.5, we anticipated that the amidine moiety might confer some hydrophilic character to these compounds, an effect which we hoped would be beneficial in the biological assay results. Moreover, in preparing these compounds the ring opening of 2-oxazolines and oxazines by aryl amine salts would be investigated, a reaction very much dependent on the nucleophilic reagent employed (see section 4.1.1.2).

# 4.2.1 AMINES AND 2-OXAZOLINE RING-OPENING REACTIONS

In sections 4.1.2.1 and 4.1.2.2 we illustrated the reaction between aryl thiols and 2-oxazolines, which is an exclusive ring opening reaction at the oxazoline 5-position. However, on substituting the aryl thiol with an amine or amine salt far less clear-cut results are obtained.

Fazio<sup>190</sup> identified carboxamide products from the reaction between 2oxazolines and a selection of secondary aliphatic and aromatic amines. Although primary aromatic amines provided a mixture of products, these too were afforded from oxazoline ring opening at the 5-position.<sup>190</sup> On the other hand, Magosch<sup>198</sup> found that 1,2- and 1,3-alkylene diamines provided oxazolidine products characteristic of 2-position attack on oxazolines. These findings were also mirrored by Fazio using other primary alkyl amines.<sup>190</sup> However the reaction pathway taken changed on introducing an oxazoline substituted at the 2-position with the bulky tertiary butyl group. As a consequence ring opening at the 5-position was forced to occur and the carboxamide product was obtained.<sup>190</sup>

Our work followed the investigations of Kormendy *et al*<sup>199</sup> into the reaction of aryl amine salts **97** and 2-position opening of oxazoline ring systems **76** (see scheme 13). In this amine addition reaction the oxazolidine **98** can afford the corresponding amidine **99**. However, purely on the premise of IR data Kormendy *et al*<sup>199</sup> concluded that they had isolated the oxazolidine **98**. On the basis of investigative work performed by Lant<sup>134</sup> it would appear Kormendy *et al* had misassigned the product and so we employed the reported experimental protocol to prepare a series of amidine salts from aryl amine salts and a selection of 2-oxazolines and oxazines.



#### 4.2.2 SYNTHESIS OF AMIDINE SALTS

The synthesis of the amidine salt target compounds **43** was a straightforward, single step procedure performed at room temperature (scheme 14). The experimental conditions employed to couple the 4-aminophenol hydrochloride **100** to a series of 2-oxazoline and oxazines **56** were those

reported by Kormendy *et al.*<sup>199</sup> The oxazine reagents (**56**, n=2) were introduced to extend the alkyl side chain length by one carbon in the product. The synthesis of these compounds and the 2-oxazolines is detailed in section 4.1.1.1 using a cadmium acetate catalysed procedure<sup>189</sup> to combine the appropriate nitrile and aminoalcohol under neat reflux conditions. The alkyl groups in the phenolic thioether series were also introduced into the amidine series through the use of oxazolines and oxazines substituted with these alkyl groups.



In this series only the amidine salts prepared from the methyl and ethyl substituted ring systems were isolated. The four products (**101** to **104**) were easily isolated from the reaction mixture through dropwise addition of chilled diethyl ether. The filtered products were recrystallised affording the purified products in high yields (mean value of 70%) with sharp melting points. Combustion data confirmed the molecular formulae for the hydrochloride salts with elemental analyses recorded within 0.1% of the theoretical values. Meanwhile EIMS for the four compounds confirmed the molecular ion for the free base and a common base peak (m/z 109) was identified as the *p*-hydroxyanilinium fragment ion.

However, following numerous attempts the preparation of the amidine salts containing cyclohexyl and phenyl moieties proved to be unsuccessful. These findings were attributed to attack at the 2-position of the oxazoline being blocked by the bulk of the substituent positioned there. This is analogous to Fazio's report<sup>190</sup> on the reaction of primary alkyl amines and 2-oxazolines which suffered from a similar handicap from bulk preventing attack at the oxazoline's 2-position (see section 4.2.1). Nevertheless, the four target compounds isolated provided interesting NMR observations which indicated that the amidine salt was present as geometric isomers.



In theory four geometric isomers of the amidine salt are possible. Consider the example shown for amidine **101**. Using simple molecular modelling the more thermodynamically favourable conformation for the geometric isomers can be visualised. In the conformation where the aromatic and amidine moieties are arranged coplanar, the steric interference would be greatest for **105** and **106** and therefore we propose that the product is composed of geometric isomers **107** and **108**.

On the <sup>1</sup>H and <sup>13</sup>C NMR spectra, pairs of peaks were observed for the two isomeric product forms and the NMR spectra of the simplest amidine salt **101** are included (diagrams 7 and 8).





Integrating the peak heights of the signals on the <sup>1</sup>H NMR spectrum (diagram 7) revealed the two geometric isomers to be present in a 1:1 ratio. In fact this was found to be the case for all four compounds prepared. The <sup>1</sup>H NMR spectra is simplified by the absence of OH and NH signals which the NMR solvent D<sub>2</sub>O exchanged (HOD peak is very obvious at  $\delta$  4.69). The methyl group can be clearly seen as a pair of sharp singlets at  $\delta$  1.96 and 2.29. The two methylene groups were observed as two pairs of triplets in the  $\delta$  3.2-3.8 region with vicinal coupling behaviour exhibited by the protons in each of the two groups. The <sup>13</sup>C NMR spectrum (diagram 8) was also beautifully clear with all the isomeric pairs of signals readily identified. The pair of signals for the methylene carbon alpha to the hydroxyl group resonated downfield at  $\delta$  58.9 and 60.1 and can be compared to the paired signals for the less deshielded methylene carbon alpha to the amidine moiety which was found to resonate at  $\delta$  44.7 and 46.0.

The IR spectra yielded evidence for the more prominent features of the amidine salts with strong absorption bands at *ca.* 1650 cm<sup>-1</sup> observed for the C=N stretching vibration and characteristic NH stretching bands in the 3200-2800 cm<sup>-1</sup> region. The phenolic OH absorption bands for bending vibrations were sharp at *ca.* 1270 cm<sup>-1</sup> while the stretching vibration found some overlap with the NH absorption bands in the *ca.* 3400 cm<sup>-1</sup> region.

These phenolic compounds were anticipated to possess UV activity and all four compounds recorded main absorption bands at *ca*.  $\lambda$  224 nm ( $\epsilon$  16700-9600). The *p*-substitution of the phenol with an amidine group appears to have extended the conjugation of the phenol chromophore and therefore the amidine moiety is considered to be behaving as a complementary auxochrome. The UV activity of the amidine salts can be compared to phenol ( $\lambda_{max}$  211 nm,  $\epsilon$  6200<sup>200</sup>) which demonstrates this observed UV behaviour.

#### 4.2.3 BIOLOGICAL EVALUATION OF AMIDINE SALTS

#### 4.2.3.1 Results and discussion

The amidine salts were evaluated biologically using the same 96 hour sulforhodamine (SRB) inhibitory assay used to investigate the activity of the phenolic thioether compounds in section 4.1.3.1. The results from this *in vitro* assessment of the cytotoxicity and selectivity of test compounds **43** are shown in tables 6 and 7.



Table 6 GI<sub>50</sub> values ( $\mu$ M) for compounds 43 (n = 1) in 96 h SRB assay

			Cell line						
Cmpd	R	B008	B0010	G361	HT144	SKMeL2	SKMeL24	SKOV-3	
101	Ме	28	45	51	42	60	110*	>100	
102	Et	15.5	32.5	43	22	58	>100	>100	

\* Obtained by extrapolation

results of particular interest

Table 7 GI<sub>50</sub> values ( $\mu$ M) for compounds **43** (**n** = **2**) in 96 h SRB assay

			Cell line					
Cmpd	R	B008	B0010	G361	HT144	SKMeL2	SKMeL24	SKOV-3
103	Ме	>100	>100	>100	>100	>100	>100	>100
104	Et	>100	66	>100	78	>100	>100	>100

From the test results it would appear that the two carbon chain feature in the amidine salts (table 6) improves the inhibitory activity of these test compounds. In particular the ethyl analogue **102** proved to be the most potent and selective for the Tyrosinase-containing cell lines. Although the corresponding methyl compound **101** was found to be inhibitory, the modest level of toxicity it showed against the non melanotic SKMeL24 cell line does question the selectivity of the compound. Compounds **103** and **104** which contain a three carbon chain (table 7) did not demonstrate any notable activity against the panel of cell lines used in the study. Nevertheless, all four amidine salts were further investigated by their inclusion in the mushroom Tyrosinase assay detailed in section 4.1.3.2. The results from this study into the compounds' potential as Tyrosinase substrates are shown in table 8.

		I	7		
		Me Et			
n = 1	101	102			
% Relative	e substrate activity	0.1	0		
n = 2	Compound	103	104		
% Relative	0	1.3			

Table 8 Relative substrate activities (%) of compounds 43for mushroom Tyrosinase

The trend previously observed for the phenolic thioether compounds between the cell line testing and enzymatic assay results was not detected for the amidine salts. Extremely low values of less than 1% for relative substrate activity were found for all the amidines including the two compounds **101** and **102** that had shown inhibitory activity in the cell line tests. The results shown in table 8 indicate that the amidine test compounds are poor substrates for Tyrosinase.

## 4.2.3.2 Conclusions

We can conclude that while the *in vitro* cell tests indicated that the amidine **102** is cytotoxic and selective towards cell lines containing Tyrosinase, the enzyme assay indicated that the cytotoxic nature of the compound is the effect of a non-Tyrosinase mediated mechanism of action. Perhaps then these results reflect the hydrophilic character of the amidine salts, an indication that these compounds may not be absorbed as readily into the tissues and as such are unable to exert their antimelanoma activity. Furthermore we believe that the amidine salts may be structurally too dissimilar from the original lead compound.

# **4.3 TYRPHOSTIN TARGET COMPOUND SERIES**

# 4.3.1 THE KNOEVENAGEL CONDENSATION REACTION<sup>22</sup>

The Knoevenagel condensation is a fundamental carbon-carbon bond forming reaction regularly used by synthetic chemists. Early investigations into this type of reaction by Knoevenagel in 1894<sup>202</sup> were followed two years later with the reaction shown in scheme 15.<sup>203</sup> This piperidine-catalysed production of the benzylidene-1,3-dicarbonyl **109** from benzaldehyde **110** and ethyl acetoacetate **111** remains a good illustration of the traditional Knoevenagel condensation.



The limited reactivity of ketones in the reaction restricts the source of carbonyl compounds to aldehydes.<sup>201</sup> However the range of aldehydes in existence is far from restricted. Similarly, a large range of active methylene compounds are available with the presence of two electron withdrawing moieties flanking the methylene group being the only structural requisite.<sup>201</sup> This great variety in the two reactant sources therefore creates a potentially abundant supply of Knoevenagel products, many of which assume the role of crucial intermediates in synthetic transformations.

The homogeneous conditions of this condensation reaction most commonly employ as a catalyst a weak base such as an amine or ammonium salt or Lewis acids such as ZnCl2<sup>204</sup> and TiCl4/base.<sup>205</sup> However, more recently the nature of the Knoevenagel catalyst has been changing, a timely reminder of the economic and environmental pressures to which modern synthetic chemists must respond. The desire for high yielding reactions under milder conditions with no side products has prompted the application of heterogeneous catalysis to the Knoevenagel reaction. Through the use of inorganic solid supports low cost, recyclable catalytic sources are available with some recent examples including rare-earth exchanged NaY zeolite<sup>206</sup> and caesium/sodium cation-exchanged mesoporous MCM-41.<sup>207</sup> As well as the use of these synthetic zeolites, aluminium oxide<sup>208</sup> and xonotlite or xonotlite doped with potassium *tert*-butoxide<sup>209</sup> are also established heterogeneous catalytic sources while the inorganic solid potassium fluoride<sup>210</sup> is an alternative base. These heterogeneous catalysts are useful to enable industrial application of the reaction.

#### 4.3.1.1 Mechanism for Knoevenagel reaction

Two mechanistic pathways are proposed for the Knoevenagel reaction and it is the base employed which decides the mechanism to be taken. The Knoevenagel mechanism<sup>211</sup> proposed by the reaction's founder occurs under the more traditional conditions when a primary or secondary amine **112** is employed. Knoevenagel proposed the formation first of an imine **113** which on reacting with the active methylene compound **114** affords an intermediate **115**. Following elimination on **115**, the desired alkene product **116** is obtained. In the presence of a tertiary amine the Hann-Lapworth mechanism<sup>212</sup> is proposed. This is a shorter mechanism and assumes the formation of  $\beta$ -hydroxy compound **117** directly from the two reactants **114** and **110** followed by dehydration to yield the Knoevenagel product **116** as before.



Scheme 16 details the Knoevenagel product of benzaldehyde **110** and 2-cyanoacetamide **114**. As depicted here and in accordance with the literature findings consistently reported for such a reaction, the geometric isomer formed is assumed to be in the *E*-form. This is the less sterically hindered and more thermodynamically stable form.

#### **4.3.2 SYNTHESIS OF TYRPHOSTIN TARGET COMPOUNDS**

The third and final programme of research we proposed for the antimelanoma area in section 2.9.5 involved the preparation of a tyrphostin series **44**. In a manner similar to the amidine series we would move away from the phenolic thioether structure common to the previous antimelanoma

compounds. Although the third compound series **44** would maintain the phenolic character we planned to introduce a *p*-substituent with alkene character. In fact these tyrphostins compounds **44** are the Knoevenagel products of phenolic benzaldehydes and malononitrile derivatives.



The typhostins 44 were included in the target antimelanoma compounds as they had already been found to possess antiproliferative activity.<sup>136,213</sup> Gazit made this discovery in compounds which incorporated the benzylidene malononitrile (BMN) nucleus 118. He coined the name "tyrphostin" for these compounds, a name which describes many compounds even those which deviate from the BMN structure. Moreover. it was the hydroxybenzylidenemalononitriles which were found to possess the most distinctive activity.<sup>213</sup> Therefore, we planned to investigate whether our tyrphostin series 44 would exhibit selectivity for melanoma by behaving as unnatural substrates for the target enzyme Tyrosinase.

#### 4.3.2.1 Synthesis results

The tyrphostins **44** were prepared under traditional Knoevenagel conditions of homogeneity with piperidine providing the catalytic source (see Scheme 17).<sup>213</sup> Piperidine is one of the more popular bases employed for a simple Knoevenagel transformation.<sup>201</sup> Three groups of tyrphostins were prepared, each derived from the aldehyde **119** employed with four malononitrile derived compounds **120** providing the active methylene reactants. In total twelve target compounds **121** to **132** were prepared. Notably, the reactions with *p*-hydroxybenzaldehyde and vanillin favoured room temperature reaction conditions whereas for the preparation of the catechol compounds reflux conditions were preferred (scheme 17). Following completion of the reaction, product precipitation was achieved (only if required) by the addition of water to the cooled ethanolic reaction mixture.

Х_ НО <sup>^</sup>	119	CHO R Pip + CN CN 120	peridine (ca DH, rt or re 19 - 94%	t.), X flux HO 4	CN R CN
	Х	R	Product	Conditions	Yield
	H H H H	$\begin{array}{c} CN\\ CO_2Et\\ CONH_2\\ C(NH_2) = C(CN)_2 \end{array}$	121 122 123 124	rt rt rt rt	81% 58% 61% 19%
	OH OH OH OH	$\begin{array}{c} CN\\ CO_2Et\\ CONH_2\\ C(NH_2){=}C(CN)_2 \end{array}$	125 126 127 128	reflux reflux reflux reflux	59% 43% 30% 25%
	OMe OMe OMe OMe	$\begin{array}{c} CN\\ CO_2Et\\ CONH_2\\ C(NH_2){=}C(CN)_2\end{array}$	129 130 131 132	rt rt rt rt	81% 94% 64% 29%
			Scheme 17		

The recrystallised products were brightly coloured compounds with sharp melting points. The three malononitrile dimer reaction products **124**, **128** and **132** recorded the lowest yields of the series at less than 30%. However, the remaining products were obtained in moderate to high yields with a mean yield of 64%.

The eight compounds from the phenol and catechol tyrphostin series **121** to **128** were known compounds and their melting points and spectroscopic data were in agreement with those reported in the literature. On the other hand, the vanillin-derived series **129** to **132** required full characterisation which confirmed their structures. The prominent nitrile, alkene and phenol features of these compounds were easily identified by IR spectroscopy. The series of bands in the 1500-1600 cm<sup>-1</sup> region were characteristic of the alkene moieties while the strong nitrile and phenol stretching frequencies could be observed at *ca.* 2200 cm<sup>-1</sup> and *ca.* 3400 cm<sup>-1</sup>, respectively. The NMR spectra for the four compounds were straightforward with successful reactions confirmed on <sup>1</sup>H NMR spectra by the absence of the vanillin aldehyde proton at *ca.*  $\delta$  10 and the appearance of the alkene proton singlet at *ca.*  $\delta$  8.2. Similarly on <sup>13</sup>C NMR

spectra the nitrile group common to all four compounds afforded a signal at *ca*.  $\delta$  117. However, it was the UV spectra for these highly conjugated compounds which produced the results of greatest interest. The alkene moiety creates an electronic compliment for the phenol within these tyrphostins. This delocalised, electron-withdrawing group behaves as an auxochrome by creating a shift in the absorption maximum of the phenol chromophore from  $\lambda_{max}$  211 nm<sup>200</sup> to *ca*.  $\lambda_{max}$  365 nm in the target compounds. This red shift towards longer wavelength is also visually observed by the bright colouring characteristic of the compounds.

#### 4.3.3 BIOLOGICAL EVALUATION OF TYRPHOSTINS

#### 4.3.3.1 Results and discussion

The tyrphostins were biologically evaluated in the same way as the phenolic thioether compounds and amidine salts. The results from the *in vitro* SRB inhibitory cell line testing of the tyrphostins are shown in tables 9, 10 and 11.



Table 9 GI<sub>50</sub> values ( $\mu$ M) for compounds 44 (X = H) in 96 h SRB assay

			Cell line					
Cmpd	R	B008	B0010	G361	HT144	SKMeL2	SKMeL24	SKOV-3
121	CN	47	27.5	33	38	49	>100	71
122	CO <sub>2</sub> Et	59	45	54	54	54	>100	95
123	CONH <sub>2</sub>	81	26.5	53	53	50	>100	>100
124	dimer	>100	>100	>100	>100	>100	>100	>100

results of interest

dimer =  $C(NH_2)=C(CN)_2$ 

Table 10 GI<sub>50</sub> values ( $\mu$ M) for compounds 44 (X = OH) in 96 h SRB assay

			Cell line							
Cmpd	R	B008	B0010	G361	HT144	SKMeL2	SKMeL24	SKOV-3		
125	CN	43.5	20	20.5	30	53	>100	>100		
126	CO <sub>2</sub> Et	>100	>100	88	100	>100	>100	>100		
127	CONH <sub>2</sub>	>100	>100	>100	>100	>100	>100	>100		
128	dimer	41	23	21	29	56	>100	>100		

dimer =  $C(NH_2)=C(CN)_2$ 

Table 11 GI<sub>50</sub> values ( $\mu$ M) for compounds 44 (X = OMe) in 96 h SRB assay

			Cell line					
Cmpd	R	B008	B0010	G361	HT144	SKMeL2	SKMeL24	SKOV-3
129	CN	>100	82	85	82	>100	>100	>100
130	CO <sub>2</sub> Et	>100	>100	>100	>100	>100	>100	>100
131	CONH <sub>2</sub>	>100	>100	>100	>100	>100	>100	>100
132	dimer	>100	>100	>100	>100	>100	>100	>100

dimer =  $C(NH_2)=C(CN)_2$ 

Overall no real patterns of inhibitory activity were found for the tyrphostins with only compounds **123**, **125** and **128** found to exhibit modest cytotoxic and selective behaviour towards the Tyrosinase cell lines. Compounds **121** and **122** were found to be moderately potent but with less selectivity as cytotoxicity was shown against the SKOV-3 ovarian cancer cell line. Furthermore very low levels of cytotoxicity were demonstrated against most cell lines by the four tyrphostins derived from vanillin, **129** to **132** (table 11). Despite these generally disappointing results, to complete the biological evaluation process all the tyrphostins were included in the mushroom Tyrosinase enzymatic assay. The results from the Tyrosinase assay are shown in table 12.

			R	
	CN	CONH <sub>2</sub>	CO <sub>2</sub> Et	dimer
X = H Compound	121	122	123	124
% Relative substrate activity	6.1	0	2.0	0.3
X = OH Compound	125	126	127	128
% Relative substrate activity	5.0	14.1	6.0	12.1
X = OMe Compound	129	130	131	132
% Relative substrate activity	2.5	1.4	1.1	0

Table 12 Relative substrate activities (%) of compounds 44 for mushroom Tyrosinase

Of the three compounds, **123**, **125** and **128**, which demonstrated moderate inhibitory activity in the cell line tests, only compound **128** produced a significant result in the Tyrosinase assay. This was only a modest value for relative substrate activity and suggests there would be some Tyrosinase-mediated cytotoxicity by the compound. All of the other tyrphostins apart from **126** exhibited low values for relative substrate activity.

## 4.3.3.2 Conclusions

When considering compounds for their potential antimelanoma activity the tyrphostins would not be a good choice. Perhaps the mushroom Tyrosinase assay provided the most useful information regarding the biological activity of these compounds. Indeed the cytotoxicity exhibited by these compounds could be attributed to a non-Tyrosinase mediated mechanism of action. The results from the biological evaluation of the tyrphostins would also appear to confirm the findings from the amidines' test results which led us to propose that moving too far away from the structure of the phenolic thioether parent compound **35** appeared to reduce the biological characteristics required for specific antimelanoma activity.

# **4.4 RECENT DEVELOPMENTS**

Following their early discoveries into the antimelanoma potency of **35**,<sup>127</sup> Jimbow and co-workers recently proposed to manipulate the lipophilic character of the drug in an attempt to improve its activity.<sup>214</sup> This compound change was achieved by preparing the propionyl derivative of the lead compound. In recent

studies this new phenolic thioether compound **133** was found to produce very interesting *in vitro* and *in vivo* results.<sup>214,215</sup>



From the *in vitro* studies compound **133** was found to be both a better substrate for Tyrosinase<sup>215</sup> and more melanocytotoxic than **35**.<sup>214</sup> Furthermore, the *in vivo* investigations revealed that the new compound significantly inhibits melanoma tumour growth and increases the life span of the melanoma-bearing study group of C57 black mice.<sup>214</sup> *In vivo* melanocytotoxicity of the drug was confirmed by its depigmentation of mice hair follicles with potency comparable to **35**.<sup>214</sup> These *in vivo* findings established **133** as a new phenolic thioether lead compound for melanogenesis-based antimelanoma agents.

These recent findings by Jimbow and co-workers were quite surprising to us as Lant prepared this propionyl derivative **133** during his research programme of work and following the inclusion of **133** in an *in vitro* SRB inhibition growth study (see table 13) this compound was not considered to possess significant antimelanoma activity.

Cell line							
B008	B0010	G361	HT144	SKMeL2	SKMeL24	SKOV-3	
82	>100	>100	>100	>100	>100	>100	

Table 13 GI50 values (µM) for compound **133** in 96 h SRB assay

The results from the assay shown in table 13 indicate that compound **133** exhibits only modest levels of cytotoxicity and selectivity.

# 4.5 CONCLUSIONS AND FUTURE PLANS

The biological evaluation of the target compounds prepared enabled us to assess their potential as antimelanoma agents. The most obvious conclusion

must be that the phenolic thioether compound series afforded the candidates with the most potential. However, both the tyrphostin and amidine salt series did produce a range of cytotoxicty values with some selectivity observed. Indications of their non-Tyrosinase mediated cytotoxicty were not too surprising and accordingly we proposed that the structures of these two series may be too far removed from the parent compound **35** to enable them to exhibit specific antimelanoma activity. Furthermore, we believed the lipophilicity of the tertiary amide compounds together with a certain combination of alkyl groups may contribute to the activity exhibited by many members of that series.

The phenolic thioether compounds proved particularly successful in the *in vitro* studies with levels of potency and selectivity for melanotic cell lines observed. We believe that the more attractive candidates from the series merit further biological evaluation. Hence we suggest that the collaboration with Professor Rona MacKie, Department of Dermatology<sup>216</sup> should be reestablished. This collaboration previously provided an *in vivo* evaluation of a selection of compounds prepared by Lant.<sup>134</sup> Moreover *in vivo* assessment of the melanocytotoxicity and antimelanoma activity of these compounds is imperative to complete their biological evaluation and should be carried out with compounds **89**, **91** and **94**.

# **CHAPTER 5**

# SYNTHESIS AND BIOLOGICAL EVALUATION OF MAP KINASE INHIBITORS

## 5.1 BACKGROUND

The inhibition of MAP kinases in cell signalling pathways is the second area of cancer research which our studies embrace. In Chapter 3 this area was discussed in some detail together with the proposed research work (section 3.6). There we introduced the lead compound **49**. This 2-iminolactone or 2-iminochromene derivative as it is also known, provided a model for our synthetic investigations into compounds with MAP kinase inhibitory potential.



# **5.2 2-IMINOLACTONE TARGET COMPOUNDS**

#### 5.2.1 2-IMINOLACTONE RING SYSTEM

The 2-iminolactone ring system enjoys positions of importance in both biological and chemical fields. Originally used in the design of conformationally constrained mimics of active compounds such as **134**, the bicyclic iminolactone compounds **135** and many of their derivatives are now established as a source of biologically active compounds in their own right.<sup>217,218</sup> Behaviour which they exhibit includes significant antitumour activity.<sup>218</sup>



Furthermore, the functionality of these 2-iminolactone compounds enables them to provide the structural base for the synthesis of polycyclic compounds such as substituted benzopyranopyridines and benzopyranopyridopyrimidines.<sup>219</sup>

#### **5.2.2 STRUCTURE ELUCIDATION OF 2-IMINOLACTONES**

Schiemenz implemented infrared-based product characterisation to formulate 2-iminolactones **136** and **137** correctly as the products obtained from the Knoevenagel condensation of salicylaldehyde **138** and malononitrile derivative **120** (scheme 18).<sup>180</sup> Previously the products were assigned the isomeric open chain structure **139** and **140**, which is the traditional type of Knoevenagel product.<sup>182</sup> IR spectroscopy was crucial to confirming the misassignment of the products **136** and **137**. In particular the characteristic nitrile stretch at *ca.* 2260 cm<sup>-1</sup> for the amide **140** in the open chain form should have been obvious on the IR spectrum. However, this was not present and bands attributed to NH groups were observed in the 3500-3300 cm<sup>-1</sup> region thus confirming the cyclic iminolactone structure **137**.



Iminolactone formation also provides information for the stereospecific nature of this carbon-carbon bond forming reaction as the nitrile group must be *cis* to the phenyl group in **140** to enable cyclisation affording the iminolactone **137**. This *trans*-stereospecific nature of the Knoevenagel condensation reaction is discussed later in section 5.3.2.2.

#### 5.2.3 SYNTHESIS OF 2-IMINOLACTONE TARGET COMPOUNDS

#### 5.2.3.1 Strategy for synthesis

Our first synthetic proposal as outlined in section 3.6 involved an investigation into the optimum position of the methoxy substituent on the benzene ring of the lead compound **49**. In order to do this a series of iminolactone compounds **50** featuring methoxy substitution on the 5-, 6-, 7- and 8-benzene ring positions were prepared as well as the unsubstituted compounds. Although the lead compound **49** featured a nitrile group, in keeping with the reported finding on the biological activity of iminolactones (see section 5.2.1) we also planned to prepare the corresponding amide derivatives.



#### 5.2.3.2 Synthesis

The *o*-hydroxyaldehyde **141** and malononitrile-derived compound **120** were condensed under the mild reaction conditions of room temperature and a catalytic amount of piperidine<sup>182</sup> (scheme 19). For each reaction the 2-iminolactone product **135** was found to precipitate readily from the alcohol reaction mixture. However, the literature reports that these isolated crude products cannot be recrystallised unchanged.<sup>181,182</sup> Therefore in order to obtain analytical purity the compounds were triturated in cold alcohol. Further handling difficulties were encountered due to the powerful sternutative and irritant properties of these bicyclic compounds.

x 4 3	2_ОН	CN pipe	EtOH, rt,	x - C - NH
x 5 6	1 +	R	pridine (cat.),	R
141	СНО	120	62 - 86%	135
_	Х	R	Product	Yield
	H	CN	136	62%
	H	CONH <sub>2</sub>	137	75%
	3-OMe	CN	142	84%
	3-OMe	CONH <sub>2</sub>	143	86%
	4-OMe	CN	144	86%
	5-OMe	CN	49	62%
	5-OMe	CONH <sub>2</sub>	145	74%
	6-OMe	CN	146	67%
	6-OMe	CONH <sub>2</sub>	147	74%
		Sch	eme 19	

The products were obtained in good yields (mean value 74%) and their structures were confirmed from the spectroscopic data collected. Compounds **136**, **137**, **143** and **145** were identified as known compounds and the data recorded for these four compounds were found to be in agreement with the findings reported in the literature. As already mentioned in section 5.2.2 perhaps the most important spectroscopic features of the iminolactones are found on the IR spectra as the characteristic nitrile band at *ca.* 2300 cm<sup>-1</sup> was absent for the amide iminolactones and this together with the NH stretching vibrations in the 3400-3200 cm<sup>-1</sup> region, confirmed the cyclic iminolactone structure. From the <sup>1</sup>H NMR spectra of the iminolactones the imine proton signal was observed at *ca.*  $\delta$  8.4 while on the <sup>13</sup>C NMR spectra the nitrile group carbons of the 3-cyanonitriles were found to resonate at *ca.*  $\delta$  116.

# 5.2.4 BIOLOGICAL EVALUATION OF 2-IMINOLACTONE TARGET COMPOUNDS

#### 5.2.4.1 In vitro JNK MAP kinase assay

The 2-iminolactones prepared were assessed for their MAP kinase inhibition by Dr David Gillespie at the Beatson Institute for Cancer Research,

Glasgow.<sup>220</sup> This investigative assay was designed as a screen to identify the best compounds as candidates for a more detailed study. The MAP kinase enzyme employed in the assay was the c-Jun N-terminal kinase (JNK), a component of the stress-activated protein kinase (SAPK) pathway which regulates c-Jun transcriptional activity (The JNK/SAPK MAPK pathway was discussed in section 3.3.2.). The principles behind this assay<sup>221</sup> involved the use of a recombinant GST-c-Jun protein (purified from E.coli) bound to Sepharose beads. This behaved as a combined affinity matrix and substrate for the JNK MAP kinases. Cell extracts were prepared from primary avian fibroblasts treated with anisomycin to provide the source of activated endogenous JNK to be incubated with the GST-c-Jun beads. The assay followed standard experimental protocol<sup>221</sup> and reactions were performed for each 2-iminolactone test compound at concentrations of 500, 50 and 5 micromolar. SDS gel electrophoresis was used to resolve the bead-bound reaction mixtures and following electroblotting, the labelling of the GST-c-Jun from each reaction was visualised by autoradiography.

#### 5.2.4.2 Results

Although the results from the assay were determined by visually analysing the autoradiography plates, this qualitative approach still enabled iminolactones that exhibited inhibitory activity to be identified. Test compounds which failed to show any inhibitory activity were easily identified from the autoradiograph by a stain of intensity comparable to that exhibited by the experimental control. The autoradiography results for the 2-iminolactones **135** are represented in table 14.



			Cmpd conc. (µM)		
Х	R	Cmpd	5	50	500
н	CN	136			$\square$
н	CONH <sub>2</sub>	137			
8-OMe	CN	142			
8-OMe	CONH <sub>2</sub>	143			
7-OMe	CN	144			
6-OMe	CN	49			$\square$
6-OMe	CONH <sub>2</sub>	145			$\square$
5-OMe	CN	146			$\Box$
5-OMe	CONH <sub>2</sub>	147			
Experiment control					

Table 14 JNK MAPK inhibitory assay results for 2-iminolactones 135



clearly inhibitory activity modest inhibitory activity no inhibitory activity

At first glance it is clear that at the lower concentrations of 5 and 50  $\mu$ M none of the 2-iminolactones exhibited inhibitory activity. As the next concentration is 500  $\mu$ M we cannot assess the inhibitory behaviour within the 50 and 500  $\mu$ M range. The model compound **49** showed clear inhibitory activity at 500  $\mu$ M concentration as did the corresponding amide iminolactone 145. Compound **146** was the third iminolactone with methoxy ring substitution to show such inhibitory activity. The simplest iminolactones 136 and 137 both exhibited activity; however it was the 3-cyanonitrile 136 which exhibited better activity. The modest inhibitory activity of **137** was also shown by **144**. The only compounds which did not show inhibitory activity at any concentration were 142 and 143, the two 8-methoxy substituted iminolactones and iminolactone 147.

#### 5.2.4.3 Trends

As we would have anticipated the model compound 49 exhibited clear inhibitory activity as did the amide iminolactone 145. An additional two compounds, **136** and **146**, were also clearly seen to inhibit. Like the lead compound **49**, these two iminolactones were also cyanonitriles. Finally, the two compounds, **137** and **144**, which were found to possess less potent inhibitory activity were amide and nitrile iminolactones which featured no substitution on the benzene ring and methoxy group at the 5-position respectively.

# **5.3 TYRPHOSTIN TARGET COMPOUNDS**

# 5.3.1 BACKGROUND

Following the proposal of an investigative study into the cyclic 2iminolactones it was considered to be a good idea to explore the synthesis of the corresponding open chain derivatives. These compounds would provide analogues of the iminolactones. Prepared in exactly the same manner as their cyclic counterparts this series would employ *o*-methoxybenzaldehyde (anisaldehyde) rather than *o*-hydroxybenzaldehyde (salicylaldehyde) to ensure no cyclisation onto the nitrile group could occur. This second series of potential MAP kinase inhibitors are tyrphostins **148** which can be prepared easily under Knoevenagel reaction conditions. A phenolic series of tyrphostins was prepared during the antimelanoma research programme and so in sections 4.3.1 and 4.3.2 the history, mechanistics and alternative reaction conditions of the Knoevenagel condensations have already been presented.

# **5.3.2 SYNTHESIS OF TYRPHOSTIN TARGET COMPOUNDS**

# 5.3.2.1 Synthesis

The open chain olefinic compounds **148** were prepared under piperidinecatalysed Knoevenagel reaction conditions.<sup>213</sup> The anisaldehyde **149** was combined with malononitrile or 2-cyanoacetamide **120** to afford a series of amide and nitrile tyrphostins **150** to **159** (scheme 20). Room temperature reaction conditions were sufficient for most of the compounds with only **151** and **159** requiring conditions of reflux. Unlike their cyclic iminolactone counterparts these open chain compounds were far easier to handle and following isolation of the products from the reaction mixture the tyrphostins were recrystallised from a selection of alcohols to afford the purified products in good yields (mean value 67%).

$X = \frac{4}{5} = \frac{3}{6} = \frac{2}{6} OMe + CN$		EtOH, piperidine (cat.), rt or reflux X-{ 43 - 87%		OMe CN R		
	149	120			148	
	Х	R	Product	Conditions	Yield	
	H H	CN CONH₂	150 151	rt reflux	43% 46%	
	3-OMe 3-OMe	CN CONH <sub>2</sub>	152 153	rt rt	68% 70%	
	4-OMe 4-OMe	CN CONH₂	154 155	rt rt	80% 80%	
	5-OMe 5-OMe	CN CONH₂	156 157	rt rt	57% 81%	
	6-OMe 6-OMe	CN CONH <sub>2</sub>	158 159	rt reflux	87% 61%	
		- ,	Scheme 2	0		

Infrared spectroscopy confirmed the open chain structure of these compounds. Unlike the iminolactones the derivatives prepared from 2cyanoacetamide produced the nitrile stretch at *ca.* 2220 cm<sup>-1</sup> on the IR spectra. Furthermore there were no bands observed in the 3400-3200 cm<sup>-1</sup> range common to iminolactone NH stretching vibrations. The success of the condensation reaction was also observed on the NMR spectra. On the <sup>1</sup>H NMR spectra of the typhostins the olefinic proton produced a singlet at *ca*.  $\delta$  8.4 which replaced the starting material aldehyde proton signal at *ca*.  $\delta$  10.5. Similarly on the <sup>13</sup>C NMR spectra the olefinic methylene carbon of the products was found to resonate at *ca*.  $\delta$  150 (compared to *ca*.  $\delta$  190 for the aldehyde carbonyl carbon). The NMR spectra obtained for the amide target compounds indicated that the product was present as a single geometric isomer and on considering the thermodynamics of the geometric isomers it is assumed that the amide group is positioned trans to the phenyl group. This stereospecific nature of the Knoevenagel reaction was briefly mentioned in section 4.3.1.1. In fact the stereochemistry of the Knoevenagel products has been reported in the literature

and in particular **151** along with several other compounds relevant to our work have been investigated.

## 5.3.2.2 NMR Studies into stereoselective Knoevenagel reaction products

NMR spectroscopy has been used by many groups to confirm the stereochemistry of Knoevenagel condensation products.<sup>222,223</sup> We followed these examples in order to confirm the *trans* stereochemistry we had assigned to tyrphostins prepared in sections 4.3.2.1 and 5.3.2.1. A <sup>13</sup>C NMR long-range selective proton decoupling (LSPD) experiment was performed on compound **151** chosen to represent the tyrphostins prepared during the course of our research. In the coupled <sup>13</sup>C NMR experiment the cyano and amide carbons were each found to produce a doublet with coupling constants of *J* 14.3 Hz and *J* 6.2 Hz respectively (see diagram 9).



The LSPD experiment which followed irradiated the olefinic proton in **151** whereupon the coupling behaviour exhibited by the two carbons mentioned was removed. This identified three-bond heteronuclear coupling between the cyano and amide carbons with the olefinic proton and the *J* values were in agreement with the reported values for this compound ( ${}^{3}J_{CN-H}$  14.2 Hz and  ${}^{3}J_{CO-H}$  6.6 Hz).<sup>222</sup> Indeed, these long distance heteronuclear coupling constants recorded for **151** confirmed the *trans*-stereochemistry assigned to this compound. Furthermore, the cyano carbon *J* value of 14.3 Hz is characteristic of *trans* positioned coupling nuclei and similarly the *cis*-arranged olefinic proton and amide group were confirmed from the smaller *J* value of 6.2 Hz. Confirmation for assigning *trans* stereochemistry to compound **151** on the basis of *J* values

was obtained from considering the three-bond coupling constants for the cyano group and olefinic proton in cinnamonitrile **160** (see diagram 9). In this compound, where the olefinic proton and cyano carbon are arranged *cis*,  ${}^{3}J_{CN-}$  H 8.7 Hz was recorded. This provides a *J* value for *cis*-coupled nuclei to compare with the values recorded for the coupling nuclei of interest within **151**.<sup>223</sup>

Other <sup>13</sup>C NMR LSPD experiments as well as X-ray crystallographic studies have been used to elucidate the *trans* stereochemistry found in various 3-, 4- and 5-phenolic substituted tyrphostins with amide and ester groups present in the olefinic side arm.<sup>223</sup> Many of these compounds possess the same or very similar structures to the tyrphostin target compounds which we prepared during our MAP kinase inhibitory and antimelanoma work. These reports<sup>222,223</sup> together with the NMR study performed on compound **151** are good evidence for the *trans* stereochemistry we assigned to our products.

## **5.3.3 BIOLOGICAL EVALUATION OF TYRPHOSTIN TARGET COMPOUNDS**

#### 5.3.3.1 Results

The tyrphostin target compounds were biologically evaluated using the same JNK MAPK inhibitory assay employed to assess the inhibitory potential of the 2-iminolactone compounds in section 5.2.4. The autoradiography results are represented in table 15.



			Cmpd conc. (µM)		
Х	R	Cmpd	5	50	500
Н	CN	150			
Н	CONH <sub>2</sub>	151			
3-OMe	CN	152		$\Box$	$\square$
3-OMe	CONH <sub>2</sub>	153			
4-OMe	CN	154			
4-OMe	CONH <sub>2</sub>	155			
5-OMe	CN	156			
5-OMe	CONH <sub>2</sub>	157			
6-OMe	CN	158			
6-OMe	CONH <sub>2</sub>	159			
Experiment control			•		

Table 15 JNK MAPK inhibitory assay results for typhostins 148



C clearly inhibitory activity M modest inhibitory activity

no inhibitory activity

As was found with the 2-iminolactones, no inhibitory behaviour was exhibited by the tyrphostins at the lower concentrations of 5 and 50 µM except for compound **152** which showed almost complete inhibitory activity at 50  $\mu$ M. This identified the malononitrile-derived typhostin **152** as the most active of the series mirroring the inhibitory activity of the best iminolactones (see section 5.2.4.2). The other five active typhostins displayed inhibitory activity of only modest proportions. Among these compounds were **150** and **151**, the two compounds possessing the simplest benzene ring substitution and the cyanoacetamide-derived tyrphostin 153. The 2,5- and 2,6-dimethoxy substituted typhostins (**157** and **159**) were the other two JNK MAPK inhibitory compounds identified.

#### 5.3.3.2 Trends

Compound **152** was identified as the lead tyrphostin in the JNK MAPK inhibitory assay. Although a further five typhostins exhibited inhibitory behaviour it was of more modest potency. The main pattern observed among these less active compounds was that four of the five compounds were cyanoacetamide-derived tyrphostins. These active compounds featured 2-, 2,3-, 2,5- and 2,6-patterns of methoxy substitution on their benzene rings. Compound **150** was the only malononitrile-derived tyrphostin among the less potent JNK MAPK inhibitory compounds.

# **5.4 3-COUMARIN TARGET COMPOUNDS**

# 5.4.1 TARGET COMPOUND DESIGN STRATEGY

The 3-substituted coumarins **161** were the third series modelled on the lead compound **49**. These target compounds were included in the design of potential MAP kinase inhibitors for the following reasons: their structural closeness to the 2-iminolactone ring system; their ease of preparation from the corresponding 2-iminolactone; and the diverse activities that the coumarin family as a whole are known to possess.

$$\begin{array}{c} 0 \\ X \\ R \\ R \\ X = H, OMe \end{array}$$

## 5.4.2 COUMARINS<sup>183</sup>

## 5.4.2.1 Naturally occurring coumarins

Coumarin is the commonly used name for compounds which are derived from or incorporate the skeleton **162**. This aryl lactone **162** was first isolated from tonka beans in 1820<sup>224</sup> and synthesised some 50 years later by Perkin.<sup>225</sup>



Most naturally occurring coumarins have been isolated from higher plants of the Umbelliferae and Rutaceae families and although their role in plant metabolism is not fully understood, it is believed that coumarins may behave as growth regulators. Out of the nearly 800 natural coumarins identified, only 35 are found to lack an oxygen function at C-7. As a result therefore 7-hydroxycoumarin **163**, also commonly referred to as umbelliferone, is regularly found to be the basic framework of complex coumarins.

#### 5.4.2.2 Properties of coumarins

Among the many coumarins of natural origins some notable examples are dicoumarol (3,3'-methylenebis(4-hydroxycoumarin)) **164**, the blood anticoagulant<sup>226</sup> and the antibiotics novobiocin **165** and coumermycin A<sub>1</sub> **166**.<sup>227</sup> Notice that dicoumarol **164** does not feature any substituents on the benzene ring, in particular the most commonly observed oxygenation at C-7. However, the two antibiotics incorporate the umbelliferone structural base with the main structural difference between them arising from the three methylpyrrolyl units in coumermycin A<sub>1</sub> **166**.



The fluorescence possessed by many natural coumarins affords extensive uses for these compounds and their synthetic counterparts. This important physical property allows 7-hydroxy and 7-aminocoumarins which contain an aryl or heteroaryl group at C-3 to be employed as laser dyes and fluorescent brightening agents in detergents, paper and textiles which mask yellowing of white materials.<sup>228,229</sup> Further exploitation is possible with some of

the structurally more simple coumarin derivatives such as 7-hydroxy-4methylcoumarin which is used in fluorimetric enzyme assays.<sup>229</sup>

As can be seen the coumarins, whether of natural or synthetic origins, enjoy a whole host of applications based on the compounds' wealth of biological and physical properties. The coumarins mentioned here and their applications described represent a small cross-section of a far larger area of interest.

#### 5.4.2.3 Syntheses of coumarins

As already mentioned in section 5.4.2.1, Perkin in 1868 was the first to synthesise coumarin **162**. This classical reaction combined salicylaldehyde **138** with acetic anhydride **167** and anhydrous sodium acetate **168**. The intermediate sodium salt **169** spontaneously cyclised to the final product **162** (scheme 21).<sup>225</sup>



This synthesis is limited to those coumarins which possess no substitution in the pyrone ring. Indeed the need for structurally different coumarins for both chemical and biological applications is reflected in the ongoing investigations into synthetic approaches to these compounds.

Substitution in the C-7 position is important to this family of compounds (see section 5.4.2.1) and Wittig<sup>230</sup> and Pechmann<sup>231</sup> chemistry in particular find common applications in the preparation of these coumarins. In order to prepare 3- and 4-substituted coumarins the requisite substituents should be in place before construction of the pyrone ring. Accordingly the Reformatsky reaction<sup>232</sup> is often used for preparing such coumarins with substituents positioned on carbon-4 of the coumarin pyrone ring. However, as we designed 3-substituted coumarins in our research programme our synthetic interest lies

more towards the preparation of these compounds. The majority of syntheses reported for such compounds (**170**) can be generalised as Knoevenagel-based one pot procedures. These tend to employ substituted salicylaldehyde **171** and active methylene compounds **172** derived from malononitrile under catalytic conditions of homo- or hetero-geneity (see scheme 22).<sup>232,233</sup> In fact these syntheses where the intermediate compound **173** is not isolated are almost identical to those outlined in section 4.3.1 for the Knoevenagel preparation of tyrphostins. The pressures of trying to attain environmentally and ecologically friendly reaction conditions for the preparation of these 3-coumarins have seen the use of inorganic solid support catalysts<sup>233</sup> and solid-phase syntheses<sup>234</sup> to mention two synthetic conditions applied.



#### 5.4.3 SYNTHESIS OF 3-COUMARIN TARGET COMPOUNDS

The 3-substituted coumarins **161** were prepared in a facile acid hydrolysis reaction which employed the freshly-prepared 2-iminolactones **135** (see section 5.2.3) as starting materials (scheme 23). On warming the aqueous solution of dilute hydrochloric acid and 2-iminolactone **135**, the corresponding lactone **161** was quickly afforded. Unlike the 2-iminolactone starting materials, the products **174** to **182** provided no handling difficulties and the 3-coumarins were recrystallised and obtained in high yields (mean value 84%).

$X_{6}^{7}$ $X_{6}^{1}$ $X_{6$	NH 2 3 R	dil. HCl, 60 °C 59 - 96%	x	
Substrate	х	R	Product	Yield
136	H	CN	174	86%
137	H	CONH <sub>2</sub>	175	89%
142	8-OMe	CN	176	89%
143	8-OMe	CONH <sub>2</sub>	177	59%
144	7-OMe	CN	178	91%
49	6-OMe	CN	179	66%
145	6-OMe	CONH <sub>2</sub>	180	96%
146	5-OMe	CN	181	96%
147	5-OMe	CONH <sub>2</sub>	182	82%
		Scheme 23		

The structural similarity between the iminolactones and lactones was most obvious from comparing the compounds' NMR spectra where the only difference observed was the absence of the imine proton signal at *ca.*  $\delta$  8.3 on the 3-coumarins' <sup>1</sup>H NMR spectra. IR spectroscopy also supported the success of the iminolactone hydrolysis with the NH stretching bands of the starting material common to the 3400-3200 cm<sup>-1</sup> region being replaced by the carbonyl absorption at *ca.* 1700 cm<sup>-1</sup> region on the IR spectra of the coumarins. This observation was more prominent for the 3-cyanonitrile coumarins as the corresponding amide compounds' carbonyl stretch masked the new lactone IR band. Further data of interest were the melting points of the 3-coumarins which were found to be consistently higher than those of the corresponding 2-iminolactones. The coumarin melting points were generally greater than 220 °C compared to *ca.* 140 to 180 °C for the 2-iminolactones.
#### **5.4.4 BIOLOGICAL EVALUATION OF 3-COUMARIN TARGET COMPOUNDS**

#### 5.4.4.1 Results

The 3-coumarins, the final set of target compounds, were assessed for their inhibitory potential in the JNK MAPK assay using the same in vitro conditions employed to investigate the 2-iminolactones and tyrphostins. In an identical manner as before the autoradiography results obtained are represented in tabular form (table 16).



			Cm	od conc. (	μΜ)
X	R	Cmpd	5	50	500
н	CN	174			
н	CONH <sub>2</sub>	175			
8-OMe	CN	176			
8-OMe	CONH <sub>2</sub>	177			
7-OMe	CN	178			
6-OMe	CN	179			
6-OMe	CONH <sub>2</sub>	180			
5-OMe	CN	181			
5-OMe	CONH <sub>2</sub>	182			
Ехр	eriment co	ntrol			

Tables 16 JNK MAPK inhibitory assay results for coumarins 161

clearly inhibitory activity modest inhibitory activity no inhibitory activity

At the 5, 50 and 500 µM compound concentrations none of the lactones were found to exhibit inhibitory activity in the assay. Therefore we would conclude that the 3-substituted coumarins did not provide candidates for further investigations into their JNK MAPK inhibitory activity.

# 5.5 OVERALL RESULTS FROM BIOLOGICAL EVALUATION OF TARGET COMPOUNDS

#### **5.5.1 CORRELATION OF RESULTS**

The results from this investigative inhibitory assay allowed us to conclude that the 2-iminolactones provided the most number of inhibitory compounds while the structurally very similar 3-substituted coumarins did not afford inhibitory compounds. However, the tyrphostins which were designed as analogues of the iminolactones provided one potent inhibitory compound and a very simplistic structural relationship was tentatively assigned between the 2-iminolactones and their open chain derivatives. This structural relationship was identified for three of the potent inhibitory iminolactones **136**, **49** and **146** which were 3-cyanonitrile iminolactones featuring no substitution, 6-methoxy and 5-methoxy substitution patterns on the benzene ring of these bicyclic compounds. Of the tyrphostins modelled on these three iminolactones it was the cyanoacetamide derivatives **151**, **157** and **159** which were found to possess inhibitory activity.

#### 5.5.2 CONCLUSIONS

As mentioned in section 5.2.4 the *in vitro* biological evaluation of the target compounds was designed as a screen to identify candidates which showed significant MAP kinase inhibitory activity. Dr Gillespie proposed to investigate the active compounds in a more detailed study. From the results shown in sections 5.2.4, 5.3.3 and 5.4.4 the compounds for further evaluation would be those 2-iminolactones and tyrphostins found to exhibit JNK MAPK inhibitory activity at 500  $\mu$ M concentration.

### **5.6 THIOPHENE ISOSTERE TARGET COMPOUNDS**

#### 5.6.1 SYNTHETIC PROPOSAL

Following the results from the biological evaluation of our target compounds where the most active compounds were electron rich aromatic derivatives, an extension of the original research programme was undertaken. This design strategy involved the preparation of thiophene isosteric replacements of the 2-iminolactones and tyrphostins which were found to exhibit MAP kinase inhibitory activity (see section 5.5.2). The target compounds **183** and **184** featured 2,3-ring substitution patterns on the bicyclic iminolactones **183** and their open chain derivatives **184**. This approach where electron-rich heterocyclic compounds are prepared as analogues of aromatic compounds to investigate structure-activity relationships is an approach commonly employed in medicinal chemistry research.



#### **5.6.2 SOME BIOLOGICALLY ACTIVE THIOPHENE SYSTEMS**

Thiophene derivatives are pharmacologically active compounds with applications which include central nervous system (CNS) and cardiovascular drugs. They are also employed in metabolic and infectious disease therapies. When we consider that infectious diseases include bacterial, fungal and viral infections the scope of thiophene derivatives in biological applications is very wide indeed.<sup>235</sup>

The use of thiophenes in biologically active compounds can be illustrated with the following examples. Consider taclamine **185** and QM-7184 **186**, both pyschotropic agents.<sup>236</sup> While taclamine is used to treat anxiety its thiophene isostere QM-7184 provides drug therapy for schizophrenia.<sup>236</sup> This demonstrates the change in pharmacological activity produced on preparing the simple thiophene isostere of compound **185**.



Quite separate from the use of thiophene as a benzene isostere in drug design is the importance of the substitution patterns on the thiophene ring of active compounds. This can be exemplified with two terthienyl isomers **187** and **188** used in infectious disease therapy. The  $\alpha$ -terthienyl **187** exhibits antifungal and nematicidal behaviour while isomer **188** is a cytotoxic agent.<sup>235,237</sup>



#### 5.6.3 THIOPHENE TYRPHOSTIN TARGET COMPOUNDS

#### 5.6.3.1 3-Methoxythiophene

The two thiophene tyrphostin target compounds were prepared before the iminolactones. The differences and resultant difficulties in the syntheses of these thiophene tyrphostins compared to their benzene counterparts became apparent immediately. The synthesis of the benzene-based tyrphostins described in section 5.3.2.1 used cheap, commercially-available anisaldehyde as starting material in the Knoevenagel condensation reaction. However, for the thiophene isosteric compounds the corresponding starting material possessing the aldehyde and methoxy groups in adjacent positions was not available. Therefore, the 2,3-disubstituted thiophene had to be synthesised.

The preparation and reactivity of 3-methoxythiophene systems have been reported in the literature<sup>238,239</sup> and it is possible to prepare 3methoxythiophene from 3-bromothiophene in a cupric oxide-catalysed reaction with sodium methylate in methanol.<sup>239</sup> However on consideration of the economics we preferred to employ commercially available 3-methoxythiophene in our synthesis.

#### 5.6.3.2 Formylation of 3-methoxythiophene

The key to the carbon-2 formylation of 3-methoxythiophene **189** is the metalation of this compound or more specifically *ortho*-lithiation.<sup>238,239</sup> The

metalation of 3-methoxythiophene **189** with *n*-butyllithium was found to occur exclusively at position 2 (scheme 24). Aldehyde **190** formation in the 2-position substantiated the formation of the 2-thienyllithium species **191** which was treated with *N*,*N*-dimethylformamide at 0 °C to afford the final product **190**. On warming the reaction mixture slowly to room temperature and quenching with water, the mixture was worked up in a standard manner. Following purification by column chromatography, the recrystallised product was obtained in 58% yield.



The combustion data confirmed the elemental composition of the formylated 3-methoxythiophene **190** and high resolution mass spectroscopy identified the compound's accurate mass. However, it was NMR spectroscopy that provided the conclusive evidence for the aldehyde group introduction into the 2-position. The <sup>1</sup>H NMR spectrum confirmed the presence of two thiophene ring protons in the downfield region ( $\delta$  6.8-7.6) in the form of two doublets and the large *J* value of 5.2 Hz was consistent with *ortho* coupled protons. The <sup>13</sup>C NMR spectrum confirmed two aromatic CH carbons (at  $\delta$  116.2 and 135.6) together with the deshielded aldehyde carbon at  $\delta$  181.7. IR spectroscopy further confirmed the successful formylation with a strong absorption at 1643 cm<sup>-1</sup> characteristic of an aryl carbonyl stretch.

The metalation of 3-*t*-butoxythiophenes has also been reported in the literature<sup>240</sup> and the *o*-lithiations of these compounds along with their 3-methoxy counterparts we have dealt with demonstrates a very powerful method for selectively introducing 2-position substituents such as acetyl, carbethoxy and alkyl groups to 3-alkoxythiophene compounds. In fact the important role played by the methoxy group in the thiophene 3-position is reflected in the metalation of 3-methylthiophene where lithiation is directed to the 5-position,<sup>238</sup> a site electronically favoured by the sulfur and the least hindered position.

#### 5.6.3.3 Synthesis of thiophene tyrphostin target compounds

Preparation of the thiophene compounds **184** was not as straightforward as the previous syntheses of tyrphostins (see section 4.3.2 and 5.3.2). The reaction conditions employed for the compounds **192** and **193** were different (see scheme 25).



The thiophene compounds 184 were the condensation products of the freshly prepared 2-formyl-3-methoxythiophene 190 (section 5.6.3.2) and active methylene compound 120 (malononitrile or 2-cyanoacetamide) (see scheme 25). The synthesis of **192** which employed 2-cyanoacetamide proceeded smoothly under Knoevenagel reaction conditions of neat piperidine catalysis and ethanol at room temperature. The product 192 was recrystallised from isopropanol to provide the target compound in 72% yield. However, the second thiophene tyrphostin 193 was not as easy to prepare. Standard piperidinecatalysed Knoevenagel reaction conditions were not conducive to obtaining the product in an acceptable yield and therefore different reaction conditions where the thiophene aldehyde and malononitrile were condensed under reflux conditions were employed. These catalyst-free reaction conditions were reported by Henrio and co-workers<sup>241</sup> who performed the condensation reaction with 2-formyl-3-t-butoxythiophene and malononitrile. The recrystallised product 193 was obtained in 68% yield and the yellow needles were found to be particularly light sensitive requiring storage in the dark.

The success of the syntheses of the two tyrphostins **192** and **193** was most obvious from their NMR spectra. The <sup>1</sup>H NMR spectra identified the products from the olefinic proton signal at *ca*.  $\delta$  8.1 and the absence of the aldehyde proton signal at  $\delta$  9.91 of the starting material. Similarly on the <sup>13</sup>C NMR spectra the aldehyde's carbon signal which resonated downfield at  $\delta$  181.7 was absent and replaced with the nitrile carbon signal at *ca*.  $\delta$  144 for both **192** and **193** and the carbonyl carbon at  $\delta$  164.6 for compound **192**. NMR spectroscopy confirmed the synthesis of a single geometric isomer for the amide **192** which we assigned *trans* stereochemistry on thermodynamically favoured grounds and in accordance with the stereoselective nature of Knoevenagel condensation products which was discussed in detail in sections 4.3.1.1 and 5.3.2.2. On the IR spectra the success of the reaction could also be observed with the presence of a characteristic nitrile stretch at *ca*. 2200 cm<sup>-1</sup> for both condensation products as well as the carbonyl vibrational band at 1704 cm<sup>-1</sup> for the amide **192**.

#### 5.6.4 THIOPHENE IMINOLACTONE TARGET COMPOUNDS

#### 5.6.4.1 Two synthetic routes proposed

Following the successful synthesis of the thiophene tyrphostin target compounds in section 5.6.3.3, the corresponding iminolactones **183** were to be prepared. We designed and investigated two different synthetic routes towards this target (scheme 26). It was anticipated that both routes would proceed via the 2-hydroxy open chain compound **194** which would spontaneously cyclise to form the bicyclic iminolactone compound **183** as their benzene counterparts were found to do in section 5.2.



However, the thiophene iminolactone compounds **183** were not formed. The attempted syntheses of these target compounds by routes 1 and 2 will now be presented in turn.

#### 5.6.4.2 Attempted synthesis of iminolactones using route 1

#### 5.6.4.2.1 Synthesis of 2-formyl-3-hydroxythiophene

The 3-hydroxy-2-formyl thiophene **195** required as starting material in the first proposed synthetic route (route 1, scheme 26) was prepared from the corresponding 3-methoxy compound **190**. This thienyl methyl ether deprotection allowed us to apply synthetic experience gained from an analogous procedure performed on a series of aryl methyl ether compounds during the antimelanoma research programme of work (section 4.1.2). As we had done previously, the Lewis acid boron tribromide was employed to facilitate this deprotection. This mild reaction was performed under conditions of low temperature and the methyl ether protection group was readily removed to afford the desired compound (scheme 27).



Purification provided the 3-hydroxythiophene compound **195** in a good yield (59%) with a sharp melting point and the spectroscopic data were found to be in agreement with the selected data reported in the literature. IR spectroscopy confirmed the presence of the hydroxyl group with a broad band in the region of 3100-2593 cm<sup>-1</sup> which we believed to be indicative of intramolecular hydrogen bonding. The relatively low value for the carbonyl absorption of 1613 cm<sup>-1</sup> provided further evidence suggesting the existence of this interaction between the thiophene 2- and 3-ring substituents. NMR spectroscopy confirmed the successful deprotection through the absence of the methyl ether signal at  $\delta$  3.92 and  $\delta$  59.2 on the respective <sup>1</sup>H and <sup>13</sup>C NMR spectra.

#### 5.6.4.2.2 Attempted synthesis of iminolactone

The first route designed to obtain the iminolactone compounds involved condensing 2-formyl-3-hydroxythiophene **195** and the appropriate malononitrile derivative **120** (malononitrile or 2-cyanoacetamide). However, various reaction conditions were investigated and applied but without success. The standard Knoevenagel condensation reaction conditions which we had successfully employed in section 4.3.2 and 5.3.2 were used with the catalytic source of piperidine (neat and diluted in ethanol) as well as aqueous potassium hydroxide. However under room temperature and refluxing reaction conditions neither of these bases was successful in catalysing the reaction. Furthermore the catalyst-free conditions used to condense 3-formyl-2-methoxythiophene and 2-cyanoacetamide in section 5.6.3.3 proved to be equally unsuccessful.

Naturally this synthetic outcome was disappointing but surprising too as we assumed that the 2-formyl-3-hydroxythiophene **195** would be a reactive species with respect to the aldehyde and hydroxyl groups enabling iminolactone synthesis with ease. However, on reflection, perhaps the intramolecular hydrogen-bonded character of this compound may have reduced the reactivity of the thiophene and this may have adversely affected the outcome of the reaction.

#### 5.6.4.3 Attempted synthesis of iminolactones using route 2

The second route designed to synthesise the thiophene iminolactone target compounds like the first route featured a thienyl methyl ether deprotection reaction. In fact within these methoxythiophene tyrphostins **184** the olefinic bond holds the cyano group in close proximity to the methoxy group such that it was anticipated that on demethylation the unmasked hydroxyl group would cyclise on to the nitrile carbon providing the bicyclic target compound **183** (see scheme 26, section 5.6.4.1).

We investigated boron tribromide and pyridinium hydrochloride as demethylating reagents to mediate the reaction in question but to no avail. Following numerous attempts, our synthesis of the thiophene iminolactone target compounds **183** by either condensing the hydroxythiophene and malononitrile derivative or in a demethylative ring closure reaction of the appropriate 2-methoxythiophene tyrphostin proved to be unsuccessful. As a result we could only supply the two methoxythiophene tyrphostin compounds **192** and **193** for biological evaluation by Dr David Gillespie.

# 5.6.5 BIOLOGICAL EVALUATION OF THIOPHENE TYRPHOSTIN TARGET COMPOUNDS

#### 5.6.5.1 Results

The thiophene tyrphostins **184** were biologically evaluated for JNK MAPK inhibitory activity and the *in vitro* assay conditions employed are detailed in section 5.2.4. The autoradiography results are presented in table 17.



Table 17 JNK MAPK inhibitory assay results for tyrphostins 184

		Cmpd conc. (µM)		
R	Cmpd	5	50	500
CONH <sub>2</sub>	192			$\square$
CN	193			$\Box$
Experime	ent control			

no inhibitory activity

From the results it is obvious that both the thiophene tyrphostins **192** and **193** showed some inhibitory activity at 50  $\mu$ M concentration and complete inhibition at 500  $\mu$ M.

#### 5.6.5.2 Conclusions

The inhibitory activity shown by the thiophene tyrphostins would appear to support the strategy behind their design (see section 5.6.1). The thiophene compounds were modelled on the corresponding electron rich benzene-based compounds found to possess inhibitory activity in sections 5.2.4, 5.3.3 and 5.4.4. We therefore propose future investigations could continue this use of electron rich heterocyclic compounds to prepare more isosteres of the active compounds e.g. using pyrroles and furans.

Following the successful biological evaluation of all the target compounds prepared, those possessing JNK MAPK inhibitory behaviour have been identified. As mentioned earlier in section 5.2.4 we now await a more detailed study involving these inhibitory compounds.

## CHAPTER 6 EXPERIMENTAL

#### 6.1GENERAL

Reagents were purchased from Aldrich Chemical Company (Gillingham, UK) or Lancaster Synthesis (UK) and were used without further purification. Organic solvents were obtained from Rhône-Poulenc Rorer and were dried, as necessary, using the procedures described in Leonard, Lygo and Procter.<sup>242</sup> Melting points were determined in open capillaries using Gallenkamp apparatus and are uncorrected. <sup>1</sup>H and <sup>13</sup>C NMR spectra were obtained on a Bruker AM200-SY spectrometer operating at 200 MHz and 50 MHz respectively, or where stated, on a Bruker DPX/400 spectrometer operating at 400 MHz and 100 MHz respectively. <sup>13</sup>C NMR spectra were assigned with the aid of Distortionless Enhancement by Polarisation Transfer (DEPT)-edited spectra and <sup>1</sup>H NMR coupling constants were calculated and reported in Hz. The numbering schemes shown are used for ease of assigning the NMR spectra and do not refer to the system of nomenclature. Mass spectra (MS) were recorded on a Jeol JMS700 high resolution mass spectrometer and percentage figures refer to relative intensity as a percentage of the base peak. MS were obtained using electron-impact ionisation (EI) mode or, where stated, chemical ionisation (CI) mode. Infrared (IR) spectra were recorded on a JASCO FT/IR 410 spectrometer. Ultraviolet (UV) spectra were recorded on a Shimadzu UV-1601 spectrophotometer and combustion data were obtained from a Carlo-Erba 1106 elemental analyser. Retention factors (Rf) were obtained by analytical Thin Layer Chromatography (TLC) on Merck aluminium-backed silica plates of 0.25 mm thickness and chromatograms were visualised using UV conditions at 254 nm or using a variety of common stains prepared using the procedures described in Leonard, Lygo and Procter.<sup>242</sup> All column chromatography was carried out on silica gel (particle size 70-230 mesh).

### **6.2 EXPERIMENTAL FOR CHAPTER 4**

#### 6.2.1 GENERAL PREPARATIVE METHODS 1 AND 2

#### **General method 1**

The method of Gazit and co-workers<sup>213</sup> was modified as follows. To a stirred solution of hydroxybenzaldehyde (1 equiv.) and malononitrile derivative (1 equiv.) in ethanol (*ca.* 0.5 mL/mmol), neat piperidine (a few drops) was added. The reaction mixture was heated at reflux until TLC indicated the reaction to be complete whereupon the mixture was cooled to room temperature. If precipitation of product was not observed dropwise addition of water was employed. The precipitated product was then filtered, dried under suction filtration and recrystallised from an appropriate solvent.

#### **General method 2**

General method 2 followed the same experimental protocol outlined for general method 1 with the exception that the reaction mixture was stirred at room temperature rather than heating under reflux.

#### 6.2.2 EXPERIMENTAL DATA

4,5-Dihydro-2-ethyloxazole 59

Compound **59** was prepared in 34% yield on a 36.3 mmol scale by the method of Witte and Seeliger<sup>189</sup> and gave bp 130-134 °C (760 mmHg) (lit.,<sup>243</sup> 124-126 °C/760 mmHg). The spectroscopic data were in agreement with the IR and <sup>1</sup>H NMR data reported.<sup>134</sup>

#### 4,5-Dihydro-2-cyclohexyloxazole 60



Compound **60** was prepared in 42% yield on a 45.8 mmol scale by the method of Witte and Seeliger<sup>189</sup> and gave bp 60-65 °C (1.5 mmHg) (lit.,<sup>244</sup> 125 °C/15 mmHg). The spectroscopic data were in agreement with the IR and <sup>1</sup>H NMR data reported.<sup>134,245</sup>

5,6-Dihydro-2-methyl-4H-1,3-oxazine 61



Compound **61** was prepared in 49% yield on a 48.7 mmol scale by the method of Witte and Seeliger<sup>189</sup> and gave bp 25-30 °C (1.5 mmHg) (lit.,<sup>246</sup> 132-133 °C/760 mmHg). The spectroscopic data were in agreement with the IR and <sup>1</sup>H NMR data reported.<sup>246</sup>

5,6-Dihydro-2-ethyl-4H-1,3-oxazine 62



Compound **62** was prepared in 43% yield on a 36.3 mmol scale by the method of Witte and Seeliger<sup>189</sup> and gave bp 38-40 °C (1.5 mmHg) (lit.,<sup>243</sup> 70-71 °C/43 Torr). The spectroscopic data were in agreement with the IR and <sup>1</sup>H NMR data reported.<sup>243</sup>

5,6-Dihydro-2-cyclohexyl-4H-1,3-oxazine 63



Compound **63** was prepared in 37% yield on a 23.1 mmol scale by the method of Witte and Seeliger<sup>189</sup> and gave bp 88-90 °C (1.5 mmHg) (lit.,<sup>134</sup> 74-76 °C/2 mmHg). The spectroscopic data were in agreement with the IR and <sup>1</sup>H NMR data reported.<sup>134</sup>

5,6-Dihydro-2-phenyl-4H-1,3-oxazine 64



Compound **64** was prepared in 29% yield on a 38.8 mmol scale by the method of Witte and Seeliger<sup>189</sup> and gave bp 80-82 °C (2 mmHg) (lit.,<sup>247</sup> 115 °C/1.5 mmHg). The spectroscopic data were in agreement with the IR and <sup>1</sup>H NMR data reported.<sup>248</sup>

N-{2-[(4-Hydroxyphenyl)thio]ethyl}acetamide 35



Compound **35** was prepared in 74% yield on a 48.2 mmol scale by the method of Padgette and co-workers<sup>129</sup> and gave mp 120-122 °C (EtOAc) (lit.,<sup>129</sup> 123-125 °C). The spectroscopic data were in agreement with the IR, high resolution mass spectroscopy and <sup>1</sup>H NMR data reported.<sup>133</sup>

N-{2-[(4-Hydroxyphenyl)thio]ethyl}cyclohexanecarboxamide 68



Compound **68** was prepared in 59% yield on a 25.5 mmol scale by the method of Padgette and co-workers<sup>129</sup> and gave mp 92-93 °C (EtOAc) (lit.,<sup>134</sup> 94-95 °C). The spectroscopic data were in agreement with the IR, high resolution mass spectroscopy and <sup>1</sup>H NMR data reported.<sup>134</sup>

N-{2-[(4-Methoxyphenyl)thio]ethyl}acetamide 73



The method of Padgette and co-workers<sup>129</sup> was used. A mixture of 2methyl-2-oxazoline (1.38 mL, 1.38 g, 16.3 mmol) and 4-methoxybenzenethiol (2.00 mL, 2.28 g, 16.3 mmol) was heated at reflux neat for 3 d under N<sub>2</sub>. The reaction was cooled to 0 °C whereupon scratching induced crystallisation. The product was purified by column chromatography, eluant ethyl acetate, and the title compound **73** (2.74 g, 12.2 mmol, 75%) was recrystallised from ethyl acetate to provide white needles, mp 100-101 °C (Found: C, 58.48; H, 6.74; N, 6.14. C11H15NO2S requires C, 58.67; H, 6.67; N, 6.22%); v<sub>max</sub> (KBr disc)/cm<sup>-1</sup> 3288m (NH), 2951w and 2936w (CH), 1636s (CO), 1562m and 1496m (C=C), 1441m (CH), 1249m and 1030m (C-O) and 815m (CH);  $\delta_{\rm H}$  (400 MHz; CDCl<sub>3</sub>) 1.87 (3 H, s, 11-H), 2.86 (2 H, t, *J* 6.4 Hz, 7-H), 3.32 (2 H, m, 8-H), 3.72 (3 H, s, 12-H), 5.88 (1 H, br. s, 9-H), 6.78 (2 H, AA'BB', d, *J* 8.8 Hz, 2,6-H) and 7.29 (2 H, AA'BB', d, *J* 8.8 Hz, 3,5-H);  $\delta_{\rm C}$  (100 MHz; CDCl<sub>3</sub>) 23.6 (11-CH<sub>3</sub>), 36.0 (7-CH<sub>2</sub>), 38.9 (8-CH<sub>2</sub>), 55.7 (12-CH<sub>3</sub>), 115.2 (2,6-CH), 125.2 (4-quat. C), 134.0 (3,5-CH), 159.7 (1-quat. C) and 170.5 (10-CO); *m/z* (EI) 225.0823 (M<sup>++</sup>. C<sub>11</sub>H<sub>15</sub>NO<sub>2</sub>S requires 225.0822); *m/z* (%) 225 (36.2), 166 (100), 139 (21.3), 135 (9.9), 109 (11.3), 86 (52.5), 77 (7.8) and 63 (5.0).

N-{2-[(4-Methoxyphenyl)thio]ethyl}cyclohexanecarboxamide 74



Compound 74 was prepared from oxazoline 60 (500 mg, 3.27 mmol) and 4-methoxybenzenethiol (0.40 mL, 458 mg, 3.27 mmol) by the method used to prepare compound **73**. Purification by column chromatography, eluant diethyl ether-hexane (2:1), and subsequent recrystallisation from ethyl acetate provided compound 74 (627 mg, 2.14 mmol, 65%) as pale tan crystals, mp 68-71 °C (Found: C, 65.28; H, 7.78; N, 4.69. C16H23NO2 requires C, 65.53; H, 7.85; N, 4.78%); v<sub>max</sub> (KBr disc)/cm<sup>-1</sup> 3313m (NH), 2927m and 2851m (CH), 1632s (CO), 1539m and 1493m (C=C), 1591w (NH), 1438w (CH), 1236m and 1027m (C-O) and 818m (CH); δH (400 MHz; CDCl<sub>3</sub>) 1.21 (6 H, m, 14 to 16-H), 1.70 (4 H, m, 12,13-H), 1.95 (1 H, m, 11-H), 2.86 (2 H, t, J 6.2 Hz, 7-H), 3.32 (2 H, m, 8-H), 3.72 (3 H, s, 17-H), 5.85 (1 H, br. s, 9-H), 6.78 (2 H, AA'BB', d, J 8.8 Hz, 2,6-H) and 7.29 (2 H, AA'BB', d, J 8.8 Hz, 3,5-H); δ<sub>C</sub> (100 MHz; CDCl<sub>3</sub>) 26.1 (14 to 16-CH<sub>2</sub>), 30.0 (12,13-CH<sub>2</sub>), 36.1 (7-CH<sub>2</sub>), 38.7 (8-CH<sub>2</sub>), 45.8 (11-CH), 55.7 (17-CH<sub>3</sub>), 115.2 (2,6-CH), 125.3 (4-quat. C), 134.0 (3,5-CH), 159.7 (1-quat. C) and 176.5 (10-CO); m/z (EI) 293.1447 (M<sup>+•</sup>. C<sub>16</sub>H<sub>23</sub>NO<sub>2</sub>S requires 293.1444); m/z (%) 293 (19.5), 166 (100), 154 (36.9), 139 (17.7), 83 (17.7), 77 (31.9) and 55 (18.1).

N-{2-[(4-Methoxyphenyl)thio]ethyl}benzamide 75



Compound 75 was prepared from 2-phenyl-2-oxazoline (0.54 mL, 600 mg. 4.08 mmol) and 4-methoxybenzenethiol (0.50 mL, 572 mg, 4.08 mmol) by the method used to prepare compound 73. Purification by column chromatography, eluant diethyl ether-hexane (2:1), and subsequent recrystallisation from ethyl acetate provided the title compound 75 (588 mg, 2.05 mmol, 50%) as a cream powder, mp 63-65 °C (Found: C, 66.77; H, 5.90; N, 4.73. C16H17NO2 requires C, 66.90; H, 5.92; N, 4.88%); vmax (KBr disc)/cm<sup>-1</sup> 3315m (NH), 1644s (CO), 1541m and 1494m (C=C), 1454m (CH), 1243s and 1029m (C-O) and 810s, 695s and 676s (CH); δ<sub>H</sub> (400 MHz; CDCl<sub>3</sub>) 2.97 (2 H, t, J 6.3 Hz, 7-H), 3.53 (2 H, m, 8-H), 3.69 (3 H, s, 17-H), 6.57 (1 H, br. s, 9-H), 6.75 (2 H, AA'BB', d, J 8.8 Hz, 2,6-H), 7.32 (4 H, m, 3,5-H and 14,15-H), 7.40 (1 H, m, 16-H) and 7.62 (2 H, m, 12,13-H); δC (100 MHz; CDCl<sub>3</sub>) 35.9 (7-CH<sub>2</sub>), 39.6 (8-CH<sub>2</sub>), 55.7 (17-CH<sub>3</sub>), 115.3 (2,6-CH), 125.2 (4-quat. C), 127.3 (14,15-CH), 128.9 (12,13-CH), 131.9 (16-CH), 134.1 (3,5-CH), 134.7 (11-quat. C), 159.7 (1-quat. C) and 167.8 (10-CO); m/z (EI) 287.0977 (M<sup>+•</sup>. C<sub>16</sub>H<sub>17</sub>NO<sub>2</sub>S requires 287.0974); *m/z* (%) 287 (25.4), 166 (100), 14.8 (34.6), 139 (9.2), 105 (55.8), 77 (46.3) and 51 (9.5).

N-{2-[(4-Methoxyphenyl)thio]ethyl}N-methylacetamide 78



A modified version of the method of Orito and co-workers<sup>192</sup> was used. To a stirred suspension of sodium hydride (60% dispersion in mineral oil, 427 mg, 10.7 mmol) in tetrahydrofuran (20 mL), at rt, was added dropwise Nacetamide 73 (1.20 g, 5.33 mmol) in tetrahydrofuran (20 mL) followed by methyl iodide (1.16 mL, 2.65 g, 18.7 mmol). The reaction was slowly heated to 40 °C. After hydrogen gas evolution ceased the reaction was maintained at a gentle reflux for 24 h. On cooling to rt white precipitates of sodium iodide were removed by suction filtration and the reaction mixture concentrated in vacuo. The oily residue was dissolved in chloroform (30 mL), washed with 10% sodium thiosulfate solution (2 x 15 mL) and water (2 x 15 mL), dried over anhydrous sodium sulfate and concentrated in vacuo. Purification by column chromatography, eluant ethyl acetate, provided the title compound 78 (1.16 g, 4.85 mmol, 91%) as a yellow oil.  $v_{max}$  (thin film)/cm<sup>-1</sup> 3002w and 2933w (CH), 1646s (CO), 1593m and 1495s (C=C), 1246s and 1030m (C-O) and 828m (CH); δH (400 MHz; CDCl<sub>3</sub>) [1.95 (1.2 H, s) and 2.02 (1.8 H, s) 11-H], [2.85 (1.2 H, s) and 2.98 (1.8 H, s) 12-H], [2.94 (0.8 H, t, J7.4 Hz) and 3.01 (1.2 H, t, J7.0 Hz) 7-H], [3.42 (0.8 H, t, J7.6 Hz) and 3.51 (1.2 H, t, J7.2 Hz) 8-H], [3.78 (1.2 H, s) and 3.80 (1.8 H, s) 13-H], 6.86 (2 H, m, 2,6-H) and 7.37 (2 H, m, 3,5-H); δ<sub>C</sub> (100 MHz; CDCl<sub>3</sub>) 21.5 and 22.2 (11-CH<sub>3</sub>), 32.9 and 34.3 (7-CH<sub>2</sub>), 33.5 and 37.6 (12-CH3), 48.3 and 50.7 (8-CH2), 55.7 and 55.7 (13-CH3), 115.0 and 115.2 (2,6-CH), 125.2 and 126.2 (4-quat. C), 132.9 and 134.2 (3,5-CH), 159.2 and 159.8 (1-quat. C) and 170.7 and 171.0 (10-CO); m/z (EI) 239.0980 (M+•. C<sub>12</sub>H<sub>17</sub>NO<sub>2</sub>S requires 239.0980); *m/z* (%) 239 (19.3), 166 (100), 139 (15.0), 135 (8.9), 100 (85.7), 86 (5.7), 77 (5.0) and 58 (14.6).

N-{2-[(4-Methoxyphenyl)thio]ethyl}N-ethylacetamide 79



Compound **79** was prepared from compound **73** (1.80 g, 8.00 mmol), ethyl bromide (2.1 mL, 3.05 g, 28.0 mmol) and sodium hydride (60% dispersion in mineral oil, 576 mg, 14.4 mmol) by the method used to prepare compound

**78**. Purification by column chromatography, eluant ethyl acetate-hexane (4:1), provided title compound **79** (1.82 g, 7.21 mmol, 90%) as a yellow oil.  $v_{\text{max}}$  (thin film)/cm<sup>-1</sup> 2971w and 2933w (CH), 1639s (CO), 1593m and 1495s (C=C), 1459m and 1442m (CH), 1246m and 1030m (C-O) and 829m (CH);  $\delta_{\text{H}}$  (400 MHz; CDCl<sub>3</sub>) [0.98 (1.2 H, t, *J* 7.1 Hz) and 1.03 (1.8 H, t, *J* 7.2 Hz) 13-H], [1.84 (1.2 H, s) and 1.95 (1.8 H, s) 11-H], [2.84 (0.8 H, t, *J* 7.8 Hz) and 2.94 (1.2 H, t, *J* 7.4 Hz) 7-H], 3.23 (2 H, m, 12-H), [3.31 (0.8 H, t, *J* 7.8 Hz) and 3.38 (1.2 H, t, *J* 7.4 Hz) 8-H], [3.70 (1.8 H, s) and 3.71 (1.2 H, s) 14-H], 6.78 (2 H, m, 2,6-H) and 7.30 (2 H, m, 3,5-H);  $\delta_{\text{C}}$  (100 MHz; CDCl<sub>3</sub>) 13.4 and 14.5 (13-CH<sub>3</sub>), 21.7 and 21.8 (11-CH<sub>3</sub>), 32.9 and 34.8 (7-CH<sub>2</sub>), 40.8 and 44.7 (12-CH<sub>2</sub>), 46.2 and 48.4 (8-CH<sub>2</sub>), 55.7 and 55.8 (14-CH<sub>3</sub>), 115.0 and 115.2 (2,6-CH), 125.1 and 125.2 (4-quat. C), 132.7 and 134.4 (3,5-CH), 159.1 and 159.9 (1-quat. C) and 170.2 and 170.6 (10-CO); *m/z* (EI) 253.1134 (M<sup>+•</sup>. C13H19NO<sub>2</sub>S requires 253.1132); *m/z* (%) 253 (20.6), 166 (100), 151 (9.2), 139 (20.6), 114 (93.6), 100 (6.4), 72 (11.3) and 58 (87.2).

N-{2-[(4-Methoxyphenyl)thio]ethyl}N-benzylacetamide 80



Compound **80** was prepared from compound **73** (225 mg, 1.00 mmol), benzyl bromide (0.42 mL, 599 mg, 3.50 mmol) and sodium hydride (60% dispersion in mineral oil, 72 mg, 1.80 mmol) by the method used to prepare compound **78**. Purification by column chromatography, eluant diethyl ether, provided title compound **80** (165 mg, 0.52 mmol, 52%) as a cloudy yellow oil.  $v_{max}$  (thin film)/cm<sup>-1</sup> 2930w (CH), 1645s (CO), 1592w and 1494s (C=C), 1464m (CH), 1245s and 1029m (C-O) and 827m and 729m (CH);  $\delta_{H}$  (400 MHz; CDCl<sub>3</sub>) [1.94 (1.3 H, s) and 1.99 (1.7 H, s) 11-H], [2.74 (0.9 H, t, *J* 8.0 Hz) and 2.94 (1.1 H, t, *J* 7.2 Hz) 7-H], [3.26 (0.9 H, t, *J* 7.8 Hz) and 3.43 (1.1 H, t, *J* 7.4 Hz) 8-H], [3.70 (1.7 H, s) and 3.73 (1.3 H, s) 19-H], 4.43 (2 H, s, 12-H), 6.76 (2 H, m, 2,6-H), 7.05 (2 H, m, 3,5-H) and 7.24 (5 H, m, 14 to 18-H);  $\delta_{C}$  (100 MHz;

CDCI<sub>3</sub>) 20.3 and 20.7 (11-CH<sub>3</sub>), 31.4 and 32.9 (7-CH<sub>2</sub>), 45.5 and 46.5 (8-CH<sub>2</sub>), 47.3 and 52.3 (12-CH<sub>2</sub>), 54.3 and 54.4 (19-CH<sub>3</sub>), 113.7 and 113.8 (2,6-CH), 123.6 and 124.8 (4-quat. C), 126.4 and 126.7 (18-CH), 125.4 and 127.2 (16,17-CH), 127.6 and 127.9 (14,15-CH), 131.3 and 133.3 (3,5-CH), 135.7 and 136.5 (13-quat. C), 157.8 and 158.5 (1-quat. C) and 169.4 and 170.1 (10-CO); *m/z* (EI) 315.1296 (M<sup>+•</sup>. C18H<sub>21</sub>NO<sub>2</sub>S requires 315.1299); *m/z* (%) 315 (14.9), 272 (1.4), 176 (67.4), 166 (89.4), 139 (19.9), 120 (32.6), 91 (100) and 65 (6.4).

N-{2-[(4-Methoxyphenyl)thio]ethyl}N-methylcyclohexanecarboxamide 81



Compound 81 was prepared from compound 74 (610 mg, 2.08 mmol), methyl iodide (0.45 mL, 1.03 g, 7.29 mmol) and sodium hydride (60% dispersion in mineral oil, 167 mg, 4.16 mmol) by the method used to prepare compound 78. Purification by column chromatography, eluant ethyl acetate-hexane (1:1), provided title compound 81 (520 mg, 1.69 mmol, 81%) as a clear yellow oil.  $v_{max}$  (thin film)/cm<sup>-1</sup> 2929s and 2853m (CH), 1638s (CO), 1593m and 1495s (C=C), 1463m (CH), 1246s and 1031m (C-O) and 828m (CH); δ<sub>H</sub> (400 MHz; CDCl<sub>3</sub>) 0.90-1.64 (10 H, m, 12 to 16-H), [2.00 (0.4 H, tt, J 11.6 and 3.6 Hz) and 2.34 (0.6 H, tt, J 11.6 and 3.3 Hz) 11-H], [2.79 (1.4 H, s) and 2.95 (1.6 H, s) 17-H], [2.83 (0.9 H, t, J 7.6 Hz) and 2.91 (1.1 H, t, J 7.2 Hz) 8-H], [3.71 (1.9 H, s) and 3.73 (1.6 H, s) 18-H], [6.76 (1.1 H, AA'BB', d, J 8.8 Hz) and 6.81 (0.9 H, AA'BB', d, J 8.8 Hz) 2,6-H] and [7.29 (1.1 H, AA'BB', d, J 8.8 Hz) and 7.33 (1.1 H, AA'BB', d, J 8.8 Hz) 3,5-H]; δ<sub>C</sub> (100 MHz; CDCl<sub>3</sub>) 26.0 and 26.1 (14 to 16-CH<sub>2</sub>), 29.4 and 29.9 (12,13-CH<sub>2</sub>), 32.9 and 34.9 (7-CH<sub>2</sub>), 33.9 and 36.1 (17-CH<sub>3</sub>), 40.8 and 41.3 (11-CH), 48.7 and 49.5 (8-CH<sub>2</sub>), 55.7 and 55.8 (18-CH<sub>3</sub>), 115.1 and 115.2 (2,6-CH), 125.1 and 126.2 (4-quat. C), 133.0 and 134.6 (3,5-CH), 159.2 and 160.0 (1-quat. C) and 176.4 and 176.5 (10-CO); m/z (EI) 307.1605 (M+\*. C17H25NO2S requires 307.1604); m/z (%) 307 (8.6), 166 (100), 139 (15.0), 111 (5.0), 83 (37.9) and 55 (16.4).

N-{2-[(4-Methoxyphenyl)thio]ethyl}N-ethylcyclohexanecarboxamide 82



Compound 82 was prepared from compound 74 (1.80 g, 6.14 mmol), ethyl bromide (1.60 mL, 2.34 g, 21.5 mmol) and sodium hydride (60% dispersion in mineral oil, 442 mg, 11.1 mmol) by the method used to prepare compound 78. Purification by column chromatography, eluant diethyl etherhexane (3:1), provided title compound 82 (524 mg, 1.63 mmol, 27%) as a yellow oil. vmax (thin film)/cm<sup>-1</sup> 2929s and 2854m (CH), 1638s (CO), 1593m and 1494s (C=C), 1450m and 1427m (CH), 1246s and 1031m (C-O) and 828m (CH); δ<sub>H</sub> (400 MHz; CDCl<sub>3</sub>) [0.97 (1.7 H, t, J 7.1 Hz) and 1.06 (1.3 H, t, J 7.1 Hz) 18-H], 1.07-1.70 (10 H, m, 12 to 16-H), [1.91 (0.4 H, tt, J 11.4 and 3.6 Hz) and 2.30 (0.6 H, tt, J 11.6 and 3.3 Hz) 11-H], [2.80 (0.9 H, t, J 8.0 Hz) and 2.91 (1.1 H, t, J 6.4 Hz) 7-H], 3.21-3.30 (2 H, m, 17-H), [3.21-3.30 (0.9 H, m) and 3.36 (1.1 H, t, J 7.4 Hz) 8-H], [3.71 (1.7 H, s) and 3.73 (1.3 H, s) 19-H], [6.77 (1.1 H, AA'BB', d, J 8.8 Hz) and 6.81 (0.9 H, AA'BB', d, J 8.4 Hz) 2,6-H] and [7.29 (1.1 H, AA'BB', d, J 8.8 Hz) and 7.34 (0.9 H, AA'BB', d, J 8.8 Hz) 3,5-H];  $\delta_{\rm C}$  (100 MHz; CDCl<sub>3</sub>) 13.5 and 15.3 (18-CH<sub>3</sub>), 25.9 and 26.2 (14 to 16-CH<sub>2</sub>), 29.9 and 30.0 (12,13-CH), 33.0 and 35.5 (7-CH<sub>2</sub>), 41.2 (11-CH), 41.0 and 43.6 (17-CH<sub>2</sub>), 46.4 and 47.2 (8-CH<sub>2</sub>), 55.7 and 55.8 (19-CH<sub>3</sub>), 115.3 and 115.2 (2,6-CH), 125.1 and 126.3 (4-quat. C), 132.7 and 134.8 (3,5-CH), 159.2 and 160.0 (1-quat. C) and 176.0 and 176.4 (10-CO); m/z (EI) 321.1765 (M+•. C<sub>18</sub>H<sub>27</sub>NO<sub>2</sub>S requires 321.1767); *m/z* (%) 321 (14.9), 293 (1.4), 182 (100), 166 (96.5), 139 (16.3), 111 (5.7), 83 (39.7) and 58 (29.8).

N-{2-[(4-Methoxyphenyl)thio]ethyl}N-benzylcyclohexanecarboxamide 83



Compound 83 was prepared from compound 74 (300 mg, 1.02 mmol), benzyl bromide (0.43 mL, 613 mg, 3.57 mmol) and sodium hydride (60% dispersion in mineral oil, 74 mg, 1.85 mmol) by the method used to prepare compound 78. Purification by column chromatography, eluant hexane-diethyl ether (1:1), provided title compound 83 (149 mg, 0.39 mmol, 38%) as a pale yellow oil. vmax (thin film)/cm<sup>-1</sup> 2928s and 2854m (CH), 1641s (CO), 1593m and 1494s (C=C), 1450m and 1424m (CH), 1246s and 1030m (C-O) and 828s and 699s (CH); 8H (400 MHz; CDCl3) 0.97 (4 H, m, 12,13-H), 1.13 (6 H, m, 14 to 16-H), [2.02 (0.6 H, tt, J 11.2 and 3.5 Hz) and 2.35 (0.4 H, tt, J 11.6 and 3.3 Hz) 11-H], [2.72 (1.1 H, t, J 8.0 Hz) and 2.90 (0.9 H, t, J 7.3 Hz) 7-H], [3.23 (1.1 H, t, J 8.0 Hz) and 3.39 (0.9 H, t, J 7.3 Hz) 8-H], [3.68 (1.4 H, s) and 3.69 (1.6 H, s) 24-H], [4.44 (1.1 H, s) and 4.47 (0.9 H, s) 17-H], [6.73 (0.9 H, AA'BB', d, J 8.8 Hz) and 6.78 (1.1 H, AA'BB', d, J 8.8 Hz) 2,6-H], 7.01 (2 H, m, 3,5-H) and 7.20 (5 H, 19 to 23-H); δ<sub>C</sub> (100 MHz; CDCl<sub>3</sub>) 24.6 and 24.7 (14 to 16-CH<sub>2</sub>), 28.5 and 28.6 (12,13-CH<sub>2</sub>), 31.4 and 33.7 (7-CH<sub>2</sub>), 39.6 and 39.9 (11-CH), 45.4 and 45.5 (8-CH<sub>2</sub>), 47.3 and 51.1 (17-CH<sub>2</sub>), 54.3 and 54.3 (24-CH<sub>3</sub>), 113.7 and 113.9 (2,6-CH), 123.6 and 124.8 (4-quat. C), 125.4 and 126.9 (21,22-CH), 126.3 and 126.6 (23-CH), 127.5 and 127.8 (19,20-CH), 133.4 and 136.2 (3,5-CH), 136.2 and 137.1 (18-quat. C), 157.7 and 158.6 (1-quat. C) and 175.2 and 175.7 (10-CO); m/z (EI) 383.1917 (M+•. C23H29NO2S requires 383.1914); m/z (EI) (%) 383 (12.8), 272 (2.5), 244 (81.6), 230 (4.3), 217 (2.1), 166 (83.7), 120 (26.9), 83 (100) and 55 (20.6).

N-{2-[(4-Methoxyphenyl)thio]ethyl}N-methylbenzamide 84



Compound 84 was prepared from compound 75 (660 mg, 2.30 mmol), methyl iodide (0.57 mL, 1.31 g, 9.20 mmol) and sodium hydride (60% dispersion in mineral oil, 166 mg, 4.15 mmol) by the method used to prepare compound **78**. Purification by column chromatography, eluant diethyl ether-hexane (4:1), provided title compound 84 (475 mg, 1.58 mmol, 68%) as a yellow oil. vmax (thin film)/cm<sup>-1</sup> 2929w and 2836w (CH), 1634s (CO), 1593m and 1494s (C=C), 1444m (CH), 1246s and 1028m (C-O) and 827m and 712m (CH); δH (400 MHz; CDCl<sub>3</sub>) [2.78 (0.9 H, br. t) and 3.07 (1.1 H, br. t) 7-H], [2.88 (1.7 H, br. s) and 2.95 (1.3 H, br. s) 17-H], [3.33 (0.9 H, br. t) and 3.61 (1.1 H, br. t) 8-H], 3.70 (3 H, s, 18-H), [6.63 (0.9 H, AA'BB', br. d) and 6.78 (1.1 H, AA'BB', br. d, AA'BB') 2,6-H], [7.01 (0.9 H, AA'BB', br. d) and 7.19-7.36 (1.1 H, AA'BB', br. d) 3,5-H], 7.21-7.36 (5 H, br. m, 12 to 16-H); δ<sub>C</sub> (100 MHz; CDCl<sub>3</sub>) 31.5 and 32.6 (7-CH<sub>2</sub>), 32.1 and 37.6 (17-CH<sub>3</sub>), 46.6 and 49.9 (8-CH<sub>2</sub>), 54.3 (18-CH<sub>3</sub>), 113.7 and 113.9 (2,6-CH), 123.5 and 124.5 (4-quat. C), 125.5 and 126.0 (14,15-CH), 127.3 and 127.4 (12,13-CH), 128.3 and 128.6 (16-CH), 131.9 and 132.3 (3,5-CH), 135.2 (11-quat. C), 158.0 and 158.1 (1-quat. C) and 170.5 and 170.9 (10-CO); m/z (EI) 301.1138 (M<sup>+•</sup>. C<sub>17</sub>H<sub>19</sub>NO<sub>2</sub>S requires 301.1139); m/z (%) 301 (23.9), 166 (100), 162 (74.3), 139 (9.3), 105 (99.3), 77 (49.6) and 51 (6.4).

N-{2-[(4-Methoxyphenyl)thio]ethyl}N-ethylbenzamide 85



Compound 85 was prepared from compound 75 (287 mg, 1.00 mmol) ethyl bromide (0.26 mL, 381 mg, 3.50 mmol), and sodium hydride (60% dispersion in mineral oil, 72 mg, 1.80 mmol) by the method used to prepare compound **78**. Purification by column chromatography, eluant hexane-ethyl acetate (3:2), provided title compound 85 (164 mg, 0.52 mmol, 52%) as a yellow oil. vmax (thin film)/cm<sup>-1</sup> 2969w and 2935w (CH), 1633s (CO), 1593m and 1494m (C=C), 1464m and 1443m (CH), 1246s and 1029m (C-O) and 828m and 707m (CH); δ<sub>H</sub> (400 MHz; CDCl<sub>3</sub>) [0.95 (1.8 H, br. t) and 1.11 (1.2 H, br. t) 18-H], [2.75 (0.8 H, br. t) and 3.09 (1.2 H, br. t) 7-H], [3.19 (1.2 H, br. g) and 3.29 (0.8 H, br. q) 17-H], [3.42 (0.8 H, br. t) and 3.55 (1.2 H, br. t) 8-H], 3.69 (3 H, s, 19-H), [6.62 (0.8 H, AA'BB', br. d) and 6.78 (1.2 H, AA'BB', br. d) 2,6-H], [6.99 (0.8 H, AA'BB', br. d) and 7.19-7.34 (1.2 H, AA'BB', br. d) 3,5-H] and 7.19-7.34 (4 H, br. m, 12 to 16-H); δ<sub>C</sub> (100 MHz; CDCl<sub>3</sub>) 12.0 and 13.2 (18-CH<sub>3</sub>), 31.5 and 32.8 (7-CH<sub>2</sub>), 39.1 and 47.5 (17-CH<sub>2</sub>), 43.8 and 43.9 (8-CH<sub>2</sub>), 54.3 and 54.5 (19-CH3), 113.7 and 113.9 (2,6-CH), 125.3 (14,15-CH), 127.4 (12,13-CH), 128.1 (4-quat.C), 128.3 (16-CH), 131.5 and 132.3 (3,5-CH), 135.6 (11-quat.C), 157.9 (1-quat. C) and 170.8 (10-CO); m/z (EI) 315.1291 (M<sup>+•</sup>. C<sub>18</sub>H<sub>21</sub>NO<sub>2</sub>S requires 315.1289); m/z (%) 315 (16.3), 176 (66.7), 166 (100), 139 (8.5), 105 (100), 77 (39.7) and 51 (4.3).

N-{2-[(4-Methoxyphenyl)thio]ethyl}N-benzylbenzamide 86



Compound **86** was prepared from compound **75** (1.0 g, 3.48 mmol), benzyl bromide (1.66 mL, 2.38 g, 13.9 mmol) and sodium hydride (60% dispersion in mineral oil, 251 mg, 6.28 mmol) by the method used to prepare compound **78**. Purification by column chromatography, eluant diethyl etherhexane (2:1), provided title compound **86** (541 mg, 1.44 mmol, 41%) as a pale yellow oil.  $v_{max}$  (thin film)/cm<sup>-1</sup> 2938w and 2835w (CH), 1633s (CO), 1494s and 1593m (C=C), 1420m (CH), 1247s and 1029s (C-O) and 827m and 700s

(CH);  $\delta_{\rm H}$  (400 MHz; CDCl<sub>3</sub>) [2.67 (0.9 H, br. t) and 3.05 (1.1 H, br. t) 7-H], [3.25 (0.9 H, br. t) and 3.52 (1.1 H, br. t) 8-H], 3.70 (3 H, s, 24-H), [4.45 (1.1 H, br. s) and 4.63 (0.9 H, br. s) 17-H], [6.62 (0.9 H, AA'BB', br. d) and 6.74 (1.1 H, AA'BB', br. d) 2,6-H], [6.99 (1.1 H, AA'BB', br. d) and 7.19-7.31 (0.9 H, AA'BB', br. d) 3,5-H] and 7.19-7.31 (10 H, br. m, 12 to 16-H and 19 to 23-H);  $\delta_{\rm C}$  (100 MHz; CDCl<sub>3</sub>) 32.8 and 33.9 (7-CH<sub>2</sub>), 45.5 and 48.4 (8-CH<sub>2</sub>), 54.3 and 60.8 (17-CH<sub>2</sub>), 55.7 (24-CH<sub>3</sub>), 115.1 (2,6-CH), 126.8 (23-CH), 127.2 (14,15-CH), 127.4 (20,21-CH), 128.6 (4-quat. C), 128.9 (12,13-CH), 129.2 (19,20-CH), 130.0 (16-CH), 132.9 (3,5-CH), 133.9 (11-quat. C), 136.6 (18-quat. C), 159.4 (1-quat. C) and 172.7 (10-CO); *m/z* (El) 377.1451 (M<sup>+•</sup>. C<sub>23</sub>H<sub>23</sub>NO<sub>2</sub>S requires 377.1453) *m/z* (%) 377 (8.6), 328 (4.3), 272 (2.5), 238 (43.6), 224 (6.1), 166 (3.6), 139 (11.1), 105 (100), 91 (40.7) and 77 (32.1).

N-{2-[(4-Hydroxyphenyl)thio]ethyl}N-methylacetamide 87



To a solution of acetamide 78 (600 mg, 2.51 mmol) in dichloromethane (20 mL), at -78 °C, was added dropwise boron tribromide (1 M in DCM, 10.0 mL, 10.0 mmol). After stirring at -78 °C for 1 h the reaction mixture was allowed to slowly warm to rt whereupon stirring was continued for a further 2 h. The reaction mixture was poured into ice/water (40 mL) and the product extracted with chloroform (3 x 20 mL). The combined organic extracts were washed with saturated sodium bicarbonate (2 x 15 mL) and brine (2 x 10 mL), dried over anhydrous magnesium sulfate and concentrated in vacuo. Purification by column chromatography, eluant ethyl acetate, followed by recrystallisation from ethyl acetate provided the title compound 87 (434 mg, 1.93 mmol, 77%) as white needles, mp 95-97 °C (Found: C, 58.57; H, 6.60; N, 6.18. C11H15NO2S requires C, 58.67; H, 6.67; N, 6.22%); λ<sub>max</sub> (MeOH)/nm 254 (ε/dm<sup>3</sup> mol<sup>-1</sup> cm<sup>-1</sup> 7000); vmax (KBr disc)/cm<sup>-1</sup> 3078-2605m (OH), 1609s (CO), 1574s and 1495s (C=C), 1447s (CH), 1277s (OH), 1240s (C-O) and 827s (CH); δ<sub>H</sub> (400 MHz; CDCl<sub>3</sub>) [1.91 (1.0 H, s) and 1.97 (2.0 H, s) 11-H], [2.73 (1.0 H, s) and 2.93 (2.0 H, s) 12-H], [2.87 (0.7 H, t, J 7.2 Hz) and 2.91 (1.3 H, t, J 7.2 Hz) 7-H], [3.83

(0.7 H, t, *J* 7.2 Hz) and 3.49 (1.3 H, t, *J* 7.2 Hz) 8-H], 6.76 (2 H, m, 2,6-H) and [7.22 (0.7 H, AA'BB', d, *J* 8.8 Hz) and 7.26 (1.3 H, AA'BB', d, *J* 8.8 Hz) 3,5-H];  $\delta_{\rm C}$  (100 MHz; CDCl<sub>3</sub>) 20.1 and 20.5 (11-CH<sub>3</sub>), 31.1 and 33.4 (7-CH<sub>2</sub>), 32.2 and 36.0 (12-CH<sub>3</sub>), 47.2 and 49.5 (8-CH<sub>2</sub>), 115.3 and 115.6 (2,6-CH), 122.2 and 122.5 (4-quat. C), 132.2 and 133.4 (3,5-CH), 155.9 and 156.3 (1-quat. C) and 170.5 and 170.9 (10-CO); *m/z* (EI) 225.0823 (M<sup>+•</sup>. C<sub>11</sub>H<sub>15</sub>NO<sub>2</sub>S requires 225.0823); *m/z* (%) 225 (18.5), 166 (17.8), 152 (100), 125 (12.8), 100 (71.9), 83 (13.5) and 58 (14.2).

N-{2-[(4-Hydroxyphenyl)thio]ethyl}N-ethylacetamide 88



Compound 88 was prepared from 79 (450 mg, 1.78 mmol) and boron tribromide (1 M in DCM, 7.10 mL, 7.10 mmol) by the method used to prepare compound 87. Purification by column chromatography, eluant ethyl acetate, and recrystallisation from benzene provided the title compound 88 (254 mg, 1.06 mmol, 60%) as white needles, mp 68-70 °C (Found: C, 59.86; H, 6.75; N, 5.69. C12H17NO2S requires C, 60.25; H, 7.11; N, 5.86%); λmax (MeOH)/nm 254 (ɛ/dm<sup>3</sup> mol<sup>-1</sup> cm<sup>-1</sup> 7360); v<sub>max</sub> (KBr disc)/cm<sup>-1</sup> 3088-2608m (OH), 2964m and 2932m (CH), 1611s (CO), 1576s and 1493s (C=C), 1434s (CH), 1271s (OH) and 828m (CH);  $\delta_{H}$  (400 MHz; CDCl<sub>3</sub>) [0.98 (0.9 H, t, J7.1 Hz) and 1.08 (2.1 H, t, J 7.2 Hz) 13-H], [1.84 (0.9 H, s) and 1.97 (2.1 H, s) 11-H], [2.84 (0.6 H, t, J 7.5 Hz) and 2.93 (1.4 H, t, J7.5 Hz) 7-H], [3.21 (0.6 H, q, J7.0 Hz) and 3.25 (1.4 H, q, J 7.2 Hz) 12-H], [3.33 (0.6 H, t, J 7.4 Hz) and 3.44 (1.4 H, t, J 7.6 Hz) 8-H], 6.75 (2 H, AA'BB', d, J 8.8 Hz, 2,6-H), 7.24 (2 H, m, 3,5-H) and [8.69 (0.3 H, br. s) and 9.09 (0.7 H, br. s) 14-H]; δ<sub>C</sub> (100 MHz; CDCl<sub>3</sub>) 13.2 and 14.4 (13-CH<sub>3</sub>), 21.3 and 21.6 (11-CH<sub>3</sub>), 32.6 and 35.1 (7-CH<sub>2</sub>), 40.9 and 44.8 (12-CH<sub>2</sub>), 46.8 and 48.5 (8-CH<sub>2</sub>), 116.7 and 116.9 (2,6-CH), 123.5 and 124.0 (4-quat. C), 133.2 and 135.0 (3,5-CH), 157.1 and 157.7 (1-quat. C) and 171.4 and 171.9 (10-CO); *m/z* (EI) 239.0983 (M<sup>+•</sup>. C<sub>12</sub>H<sub>17</sub>NO<sub>2</sub>S requires 239.0986); *m/z* (%) 239 (15.5), 237 (0.7), 152 (100), 114 (78.9), 100 (7.7), 72 (10.6) and 58 (79.6).

N-{2-[(4-Hydroxyphenyl)thio]ethyl}N-benzylacetamide 89



Compound 89 was prepared from 80 (400 mg, 1.27 mmol) and boron tribromide (1 M in DCM, 5.08 mL, 5.08 mmol) by the method used to prepare compound 87. Purification by column chromatography, eluant diethyl ether, and recrystallisation from chloroform/hexane provided the title compound 89 (291 mg, 0.97 mmol, 76%) as white crystals, mp 90-92 °C (Found: C, 67.70; H, 6.49; N, 4.46. C17H19NO2S requires C, 67.77; H, 6.31; N, 4.65%); λmax (MeOH)/nm 253 (ε/dm<sup>3</sup> mol<sup>-1</sup> cm<sup>-1</sup> 6800); v<sub>max</sub> (KBr disc)/cm<sup>-1</sup> 3100-2500m (OH), 2933m (CH), 1606s (CO), 1494s and 1476s (C=C), 1433s (CH), 1360m (CH), 1284s (C-O) and 835s and 695s (CH); δH (400 MHz; CDCl<sub>3</sub>) [1.92 (1.2 H, s) and 2.01 (1.8 H, s) 11-H], [2.74 (0.8 H, t, J 7.3 Hz) and 2.92 (1.2 H, t, J 6.9 Hz) 7-H], [3.30 (0.8 H, t, J 7.2 Hz) and 3.48 (1.2 H. t, J 6.9 Hz) 8-H], [4.41 (0.8 H, s) and 4.45 (1.2 H, s), 12-H], 6.75 (2 H, AA'BB', d, J 8.0 Hz, 2,6-H), 7.03 (2 H, AA'BB', d, J 6.0 Hz, 3,5-H), 7.17-7.26 (5 H, m, 14 to 18-H) and [8.40 (0.4 H, br. s) and 8.76 (0.6 H, br. s) 19-H]; δ<sub>C</sub> (100 MHz; CDCl<sub>3</sub>) 21.5 and 21.8 (11-CH<sub>3</sub>), 32.6 and 34.6 (7-CH<sub>2</sub>), 47.4 and 53.6 (12-CH<sub>2</sub>), 48.0 and 48.8 (8-CH<sub>2</sub>), 116.8 and 117.0 (2,6-CH), 123.5 and 124.2 (4-quat. C), 126.8 and 128.6 (16,17-CH), 128.0 and 128.3 (18-CH), 129.1 and 129.5 (14,15-CH), 133.3 and 135.1 (3,5-CH), 136.4 and 137.3 (13-quat. C), 157.0 and 157.7 (1-quat. C) and 171.9 and 173.0 (10-CO); m/z (EI) 301.1134 (M+•. C17H19NO2S requires 301.1132); m/z (%) 301 (13.4), 258 (1.4), 176 (54.9), 152 (93.7), 120 (42.3), 106 (4.2), 91 (100), 65 (8.8) and 63 (1.4).

N-{2-[(4-Hydroxyphenyl)thio]ethyl}N-methylcyclohexanecarboxamide 90



Compound 90 was prepared from 81 (400 mg, 1.30 mmol) and boron tribromide (1 M in DCM, 5.46 mL, 5.46 mmol) by the method used to prepare compound 87. Purification by column chromatography, eluant diethyl ether, and recrystallisation from diethyl ether provided the title compound 90 (259 mg, 0.88 mmol, 68%) as cream crystals, mp 90-92 °C (Found: C, 65.55; H, 7.81; N, 4.73. C<sub>16</sub>H<sub>23</sub>NO<sub>2</sub>S requires C, 65.53; H, 7.85; N, 4.78%); λ<sub>max</sub> (MeOH)/nm 253 (ɛ/dm<sup>3</sup> mol<sup>-1</sup> cm<sup>-1</sup> 3960); v<sub>max</sub> (KBr disc)/cm<sup>-1</sup> 3155m (OH), 2929m and 2852m (CH), 1615s (CO), 1580s and 1497s (C=C), 1447m (CH), 1269m (OH), 1234m (C-O) and 835m (CH) ; δ<sub>H</sub> (400 MHz; CDCl<sub>3</sub>) 0.96-1.72 (10 H, m, 12 to 16-H), [2.13 (0.4 H, tt, J 11.6 and 3.2 Hz) and 2.38 (0.6 H, tt, J 11.6 and 3.2 Hz) 11-H], [2.76 (1.2 H, s) and 2.99 (1.8 H, s) 17-H], 2.83-2.89 (2 H, m, 7-H), [3.38 (0.8 H, t, J 7.4 Hz) and 3.46 (1.2 H, t, J 7.4 Hz) 8-H], [6.71 (1.2 H, AA'BB', d, J 8.8 Hz) and 6.80 (0.8 H, AA'BB', d, J 8.8 Hz) 2,6-H], [7.22 (0.8 H, AA'BB', d, J 8.4 Hz) and 7.25 (1.2 H, AA'BB', d, J 8.4 Hz) 3,5-H] and [8.00 (0.4 H, br. s) and 8.20 (0.60 H, br. s) 18-H]; δ<sub>C</sub> (100 MHz; CDCl<sub>3</sub>) 26.0 and 26.1 (14 to 16-CH<sub>2</sub>), 29.3 and 29.9 (12,13-CH<sub>2</sub>), 33.1 and 35.1 (7-CH<sub>2</sub>), 34.2 and 36.6 (17-CH<sub>3</sub>), 41.0 and 41.5 (11-CH), 48.8 and 49.8 (8-CH<sub>2</sub>), 116.8 and 117.0 (2,6-CH), 123.7 and 124.0 (4-quat. C), 134.0 and 134.9 (3,5-CH), 157.2 and 157.6 (1-quat. C) and 177.4 and 177.5 (10-quat. C); m/z (EI) 293.1448 (M<sup>+•</sup>. C<sub>16</sub>H<sub>23</sub>NO<sub>2</sub>S requires 293.1447); m/z (%) 293 (13.7), 170 (86.3), 152 (100), 142 (17.3), 125 (15.1), 83 (48.9) and 55 (20.5).

N-{2-[(4-Hydroxyphenyl)thio]ethyl}N-ethylcyclohexanecarboxamide 91



Compound 91 was prepared from 82 (380 mg, 1.18 mmol) and boron tribromide (1 M in DCM, 5.92 mL, 5.92 mmol) by the method used to prepare compound 87. Purification by column chromatography, eluant ethyl acetatehexane (3:1), and recrystallisation from benzene provided the title compound 91 (154 mg, 0.50 mmol, 43%) as white crystals, mp 103-104 °C ;  $\lambda_{max}$  (MeOH)/nm 253 (ε/dm<sup>3</sup> mol<sup>-1</sup> cm<sup>-1</sup> 5990); v<sub>max</sub> (KBr disc)/cm<sup>-1</sup> 3100-2900m (OH), 2933m and 2855m (CH), 1600s (CO), 1583s and 1499s (C=C), 1431m (CH), 1272s (OH), 1226m (C-O) and 833m (CH); δ<sub>H</sub> (400 MHz; CDCl<sub>3</sub>) [0.97 (1.5 H, t, J 6.8 Hz) and 1.09 (1.5 H, t, J 7.0 Hz) 18-H], 0.95-1.17 (6 H, m, 14 to 16-H), 1.33-1.70 (4 H, m, 12,13-H), [2.02 (0.9 H, br. t, J 11.6 Hz) and 2.33 (1.1 H, br. t, J 11.4 Hz) 11-H], [2.81 (0.9 H, t, J7.6 Hz) and 2.89 (1.1 H, t, J7.4 Hz) 7-H], 3.32-3.20 (2 H, m, 17-H), [3.32 (0.9 H, t) and 3.41 (1.1 H, t, J 7.4 Hz) 8-H], [6.70 (1.1 H, AA'BB', d, J 8.2 Hz) and 6.81 (0.9 H, AA'BB', d, J 8.2 Hz) 2,6-H], [7.22 (1.1 H, AA'BB', d, J 8.3 Hz) and 7.26 (0.9 H, AA'BB', d, J 8.2 Hz) 3,5-H] and [8.32 (0.4 H, br. s) and 8.40 (0.6 H, br. s) 19-H]; δC (100 MHz; CDCl<sub>3</sub>) 13.4 and 15.3 (18-CH<sub>3</sub>), 26.0 and 26.1 (14 to 16-CH<sub>2</sub>), 29.8 and 29.9 (12,13-CH<sub>2</sub>), 33.4 and 35.7 (7-CH<sub>2</sub>), 41.1 and 41.3 (17-CH<sub>2</sub>), 41.3 and 43.7 (11-CH), 46.8 and 47.5 (8-CH<sub>2</sub>), 116.8 and 117.0 (2,6-CH), 123.4 and 123.9 (4-quat. C), 133.9 and 135.1 (3,5-CH), 157.2 and 157.8 (1-quat. C) and 176.9 and 177.4 (10-CO); m/z (EI) 307.1609 (M<sup>+•</sup>. C<sub>17</sub>H<sub>25</sub>NO<sub>2</sub>S requires 307.1612); *m/z* (%) 307 (14.2), 182 (100), 232 (0.7), 152 (93.6), 125 (17.0), 111 (8.5), 83 (56.0) and 58 (46.1).

N-{2-[(4-Hydroxyphenyl)thio]ethyl}N-benzylcyclohexanecarboxamide 92



Compound 92 was prepared from 83 (265 mg, 0.69 mmol) and boron tribromide (1 M in DCM, 2.77 mL, 2.77 mmol) by the method used to prepare compound 87. Purification by column chromatography, eluant diethyl etherhexane (2:1), provided the title compound 92 (135 mg, 0.37 mmol, 53%) as a pale tan powder, mp 117-120 °C; λmax (MeOH)/nm 256 (ε/dm<sup>3</sup> mol<sup>-1</sup> cm<sup>-1</sup> 10120); v<sub>max</sub> (KBr disc)/cm<sup>-1</sup> 3100-2900m (OH), 2954m and 2958m (CH), 1618s (CO), 1583s and 1495s (C=C), 1436s (CH), 1263s (OH), 1146m (C-O) and 836m and 696m (CH);  $\delta_{H}$  (400 MHz; CDCl<sub>3</sub>) 0.95-1.18 (4 H, m, 12,13-H), 1.41-1.68 (6 H, m, 14 to 16-H), [2.11 (0.5 H, tt, J 11.4 Hz) and 2.39 (0.6 H, tt, J 11.6 and 3.2 Hz) 11-H], [2.73 (1.0 H, t, J7.8 Hz) and 2.88 (1.0 H, t, J7.4 Hz) 7-H], [3.28 (1.0 H, t, J 7.8 Hz) and 3.43 (1.0 H, t, J 7.4 Hz) 8-H], [4.44 (1.0 H, s) and 4.51 (1.0, s) 17-H], [6.69 (1.0 H, AA'BB', d, J 6.8 Hz) and 6.77 (1.0 H, AA'BB', d, J 8.8 Hz) 2,6-H], 7.01 (2 H, m, 3,5-H), 7.14-7.29 (5 H, m, 19 to 23-H) and [7.77 (0.5 H, br. s) and 7.84 (0.5 H, br. s) 24-H]; δ<sub>C</sub> (100 MHz; CDCl<sub>3</sub>) 24.6 and 24.7 (14,15,16-CH<sub>2</sub>), 28.4 and 28.5 (12,13-CH<sub>2</sub>), 31.8 and 33.8 (7-CH<sub>2</sub>), 39.7 and 40.1 (11-CH), 45.6 and 45.8 (8-CH<sub>2</sub>), 47.5 and 51.1 (17-CH<sub>2</sub>), 115.4 and 115.6 (2,6-CH), 122.3 and 123.0 (4-quat. C), 125.4 and 126.8 (21,22-CH), 126.4 and 126.7 (23-CH), 127.6 and 128.0 (19,20-CH), 132.3 and 133.7 (3,5-CH), 135.6 and 136.4 (18-quat. C), 155.5 and 156.1 (1-quat. C) and 176.1 and 176.7 (10-CO); m/z (EI) 369.1761 (M<sup>+•</sup>. C<sub>22</sub>H<sub>27</sub>NO<sub>2</sub>S requires 369.1760); m/z (%) 369 (13.5), 244 (100), 218 (46.1), 205 (1.4), 152 (94.3), 120 (51.1), 91 (73.1), 83 (53.2) and 55 (33.3).

N-{2-[(4-Hydroxyphenyl)thio]ethyl}N-methylbenzamide 93



Compound 93 was prepared from 84 (300 mg, 1.00 mmol) and boron tribromide (1 M in DCM, 4.0 mL, 4.0 mmol) by the method used to prepare compound 87. Recrystallisation from ethyl acetate provided the title compound 93 (253 mg, 0.88 mmol, 88%) as white crystals, mp 121-122 °C (Found: C, 66.73; H, 5.87; N, 4.79. C16H17NO2S requires C, 66.90; H, 5.92; N, 4.88%);  $\lambda_{max}$  (MeOH)/nm 253 ( $\epsilon$ /dm<sup>3</sup> mol<sup>-1</sup> cm<sup>-1</sup> 9380);  $\nu_{max}$  (KBr disc)/cm<sup>-1</sup> 3100-2600m (OH), 2918m and 2859m (CH), 1608s (CO), 1574s and 1494s (C=C), 1440s (CH), 1236s (C-O) and 831m and 698m (CH); δ<sub>H</sub> (400 MHz; CDCl<sub>3</sub>) [2.80 (0.6 H, t, J 7.2 Hz) and 3.04 (1.4 H, t, J 7.0 Hz) 7-H], 2.92 (3 H, s, 17-H), [3.36 (0.6 H, t, J7.2 Hz) and 3.66 (1.4 H, t, J7.0 Hz) 8-H], [6.56 (0.6 H, AA'BB', d, J 8.2 Hz) and 6.69 (1.4 H, AA'BB', d, J 8.4 Hz) 2,6-H], [6.98 (0.6 H, AA'BB', d, J 8.1 Hz) and 7.22-7.33 (1.4 H, AA'BB', d) 3,5-H], 7.22-7.33 (5 H, m, 12 to 16-H) and 7.83 (1 H, br. s, 18-H); SC (100 MHz; CDCl3) 32.2 and 37.5 (17-CH3), 31.7 and 33.2 (7-CH<sub>2</sub>), 46.9 and 50.1 (8-CH<sub>2</sub>), 115.4 (2,6-CH), 122.5 (4-quat. C), 125.5 and 125.9 (14,15-CH), 127.5 (16-CH), 128.6 and 129.0 (12,13-CH), 132.6 and 132.9 (3,5-CH), 134.5 and 137.1 (11-quat. C), 155.8 (1-quat. C) and 171.4 (10-CO); *m/z* (EI) 287.0977 (M<sup>+•</sup>. C<sub>16</sub>H<sub>17</sub>NO<sub>2</sub>S requires 287.0974); *m/z* (%) 287 (12.0), 163 (4.2), 152 (80.9), 136 (10.6), 105 (100), 77 (45.8) and 51 (7.7).

N-{2-[(4-Hydroxyphenyl)thio]ethyl}N-ethylbenzamide 94



Compound 94 was prepared from 85 (880 mg, 2.79 mmol) and boron tribromide (1 M in DCM, 11.2 mL, 11.2 mmol) by the method used to prepare compound 87. Purification by column chromatography, eluant ethyl acetatehexane (3:2), and recrystallisation from ethyl acetate provided the title compound **94** (441 mg, 1.47 mmol, 53%) as white needles, mp 120-122 °C (Found: C, 67.58; H, 6.32; N, 4.56. C17H19NO2S requires C, 67.77; H, 6.31; N, 4.65%); λ<sub>max</sub> (MeOH)/nm 253 (ε/dm<sup>3</sup> mol<sup>-1</sup> cm<sup>-1</sup> 7780); ν<sub>max</sub> (KBr disc)/cm<sup>-1</sup> 3109-2792m (OH), 3077m and 2934m (CH), 1604s (CO), 1572s and 1495s (C=C), 1431m (CH), 1269s (OH) and 829w (CH); δ<sub>H</sub> (400 MHz; CDCl<sub>3</sub>) [1.00 (2.1 H, t, J 7.0 Hz) and 1.22 (0.9 H, br. t) 18-H], [2.74 (0.6 H, br. t) and 3.07 (1.4 H, t, J 7.2 Hz) 7-H], [3.24 (1.4 H, q, J 6.9 Hz) and 3.31 (0.6 H, br. q) 17-H], [3.41 (0.6 H, br. t) and 3.60 (1.4 H, t, J 7.2 Hz) 8-H], [6.53 (0.6 H, d, J 7.5 Hz) and 6.67 (1.4 H, AA'BB', d, J 8.3 Hz) 2,6-H], [6.92 (0.6 H, AA'BB', d, J 7.4 Hz) and 7.22-7.45 (1.4 H, AA'BB', d) 3,5-H], 7.22-7.45 (5 H, m) and 8.05 (1 H, br. s, 19-H); δ<sub>C</sub> (100 MHz; CDCl<sub>3</sub>) 11.9 and 13.1 (18-CH<sub>3</sub>), 31.7 and 33.3 (7-CH<sub>2</sub>), 39.4 and 47.7 (17-CH<sub>2</sub>), 43.8 and 44.3 (8-CH<sub>2</sub>), 115.4 and 115.5 (2,6-CH), 121.9 and 122.7 (4-quat. C), 125.2 (14,15-CH), 127.5 (12,13-CH), 128.5 and 128.7 (16-CH), 132.3 and 132.9 (3,5-CH), 134.8(11-quat. C), 155.7 (1-quat. C) and 171.2 and 171.6 (10-CO); m/z (EI) 301.1135 (M<sup>+•</sup>. C<sub>17</sub>H<sub>19</sub>NO<sub>2</sub>S requires 301.1133); m/z (%) 301 (15.5), 176 (36.9), 152 (68.1), 150 (12.8), 105 (100), 77 (36.9) and 51 (4.9).

N-{2-[(4-Hydroxyphenyl)thio]ethyl}N-benzylbenzamide 95



Compound **95** was prepared from **86** (346 mg, 0.92 mmol) and boron tribromide (1 M in DCM, 3.67 mL, 3.67 mmol) by the method used to prepare compound **87**. Recrystallisation from ethyl acetate provided the title compound **95** (241 mg, 0.66 mmol, 72%) as white crystals, mp 101-102 °C (Found: C, 72.59; H, 5.79; N, 3.83. C<sub>22</sub>H<sub>21</sub>NO<sub>2</sub>S requires C, 72.73; H, 5.71; N, 3.86%);

 $\lambda_{max}$  (MeOH)/nm 254 (ε/dm<sup>3</sup> mol<sup>-1</sup> cm<sup>-1</sup> 9700); v<sub>max</sub> (KBr disc)/cm<sup>-1</sup> 3400-3000m (OH), 2949m (CH), 1613s (CO), 1598s and 1498s (C=C), 1465m (CH), 1277s (OH), 1360m (CH), 1158m (C-O) and 826m and 698m (CH); δ<sub>H</sub> (400 MHz; CDCl<sub>3</sub>) [2.68 (0.7 H, br. t) and 3.02 (1.3 H, t, *J* 7.0 Hz) 7-H], [3.26 (0.7 H, br. t) and 3.55 (1.3 H, t, *J* 7.0 Hz) 8-H], [4.48 (1.3 H, s) and 4.61 (0.7 H, s) 17-H], 6.36 (1 H, br. s, 24-H), [6.54 (0.6 H, AA'BB', d, *J* 7.7 Hz) and 6.67 (1.4 H, AA'BB', d, *J* 8.2 Hz) 2,6-H], [6.90 (0.6 H, AA'BB', d, *J* 7.7 Hz) and 7.02 (1.4 H, AA'BB', d, *J* 7.0 Hz) 3,5-H] and 7.16-7.34 (10 H, m, 12 to 16-H and 19 to 23-H); δ<sub>C</sub> (100 MHz; CDCl<sub>3</sub>) 31.4 and 32.9 (7-CH<sub>2</sub>), 44.3 and 47.1 (8-CH<sub>2</sub>), 52.8 (17-CH<sub>2</sub>), 116.7 (2,6-CH), 123.0 (4-quat. C), 125.3 (23-CH), 125.7 and 126.0 (14,15-CH), 126.0 (21,22-CH), 126.8 and 127.2 (12,13-CH), 127.6 and 127.9 (19,20-CH), 128.6 (11-quat. C), 129.0 (16-CH), 134.4 and 135.1 (18-quat. C), 155.4 (1-quat. C) and 171.9 (10-CO); *m/z* (EI) 363.1296 (M<sup>+•</sup>. C<sub>22</sub>H<sub>21</sub>NO<sub>2</sub>S requires 363.1299); *m/z* (%) 363 (10.2), 258 (2.1), 238 (37.3), 212 (20.4), 152 (80.3), 125 (7.0), 105 (100), 91 (35.9), 77 (33.5) and 65 (4.9).

N-(2-Hydroxyethyl)-N'-(4-hydroxyphenyl)ethanamidine hydrochloride 101



Compound **101** was prepared in a 77% yield on a 13.7 mmol scale by the method of Kormendy and co-workers<sup>199</sup> and gave mp 154-155 °C ( $^{i}$ PrOH) (lit.,<sup>134</sup> 155-156 °C). The spectroscopic data were in agreement with the IR and <sup>1</sup>H NMR data reported.<sup>134</sup>

*N*-(2-Hydroxyethyl)-*N*'-(4-hydroxyphenyl)propionamidine hydrochloride **102** 



The method of Kormendy and co-workers<sup>199</sup> was used. To a solution of 4-aminophenol hydrochloride (735 mg, 5.05 mmol) in ethanol (20 mL) was added oxazoline 59 (500mg, 5.05 mmol). The reaction mixture was stirred at rt for 3 d whereupon chilled diethyl ether was added drowise to precipitate the product. The precipitate was filtered and dried under suction. The title compound **102** (850 mg, 3.48 mmol, 69%) was recrystallised from isopropanol to provide cream crystals, mp 158-161 °C (Found: C, 53.90; H, 6.88; N, 11.52. C11H17N2O2Cl requires C, 53.99; H, 6.95; N, 11.45%); λmax (H2O)/nm 223 (ɛ/dm<sup>3</sup> mol<sup>-1</sup> cm<sup>-1</sup> 16670); v<sub>max</sub> (thin film)/cm<sup>-1</sup> 3468m (OH), 3172-2920s (NH), 1639vs (C=N), 1585w (NH), 1516s (C=C), 1265m (OH) and 850w (CH); δH (400MHz, D2O) 1.23 (3 H, t, J 7.6 Hz, 11-H) and 0.95 (3 H, t, J 7.6 Hz, 11-H), 2.32 (2 H, q, J7.6 Hz, 10-H) and 2.63 (2 H, q, J7.6 Hz, 10-H), 3.37 (2 H, t, J 5.2 Hz, 8-H) and 3.43 (2 H, t, J 5.2 Hz, 8-H), 3.52 (2 H, t, J 5.2 Hz, 9-H) and 3.73 (2 H, t, J 5.2 Hz, 9-H), 6.83 (2 H, AA'BB', d, J 8.8 Hz, 2,6-H) and 6.86 (2 H, AA'BB', d, J 8.8 Hz, 2,6-H) and 7.07 (2 H, m, 3,5-H); δ<sub>C</sub> (100MHz, D<sub>2</sub>O) 10.1 and 10.5 (11-CH<sub>3</sub>), 23.7 and 24.4 (10-CH<sub>2</sub>), 44.6 and 45.6 (8-CH<sub>2</sub>), 59.0 and 60.0 (9-CH<sub>2</sub>), 116.5 and 117.3 (3,5-CH), 125.3 and 127.3 (4-quat. C), 128.4 and 128.9 (2,6-CH), 156.3 and 156.7 (1-guat. C) and 169.7 and 169.9 (7-guat. C); m/z (EI) 208.1212 (M+-HCI. C11H16N2O2 requires 208.1211); m/z (%) 208 (7.8), 189 (2.5), 163 (2.8), 148 (8.5), 134 (4.6), 109 (100), 99 (29.8), 80 (35.5), 69 (31.2) and 54 (16.3).

N-(3-Hydroxypropyl)-N'-(4-hydroxyphenyl)ethanamidine hydrochloride 103



Compound **103** was prepared from oxazine **61** (612 mg, 6.18 mmol) and 4-aminophenol hydrochloride (900 mg, 6.18 mmol) by the method used to prepare compound **102**. Recrystallisation from isopropanol provided **103** (950 mg, 3.89 mmol, 63%) as a white powder, mp 185-187 °C (Found: C, 53.88; H, 7.01; N, 11.41. C11H17N2O2Cl requires C, 53.99; H, 6.95; N, 11.45%);  $\lambda_{max}$  (H<sub>2</sub>O)/nm 225 ( $\epsilon$ /dm<sup>3</sup> mol<sup>-1</sup> cm<sup>-1</sup> 9580);  $\nu_{max}$  (thin film)/cm<sup>-1</sup> 3409m (OH), 3225-2923s (NH), 1649vs (C=N), 1520s and 1448s (C=C), 1280m (OH), 1068m (C-O) and 847w (CH);  $\delta_{H}$  (400MHz, D<sub>2</sub>O) 1.60 (2 H, m, 9-H) and 1.80 (2 H, m,

9-H), 1.92 (3 H, s, 11-H) and 2.25 (3 H, s, 11-H), 3.30 (2 H, m, 8-H), 3.45 (2 H, t, *J* 6.0 Hz, 10-H) and 3.57 (2 H, t, *J* 6.2 Hz, 10-H), 6.78 (2 H, AA'BB', d, *J* 8.8 Hz, 2,6-H) and 6.81 (2 H, AA'BB', d, *J* 8.8 Hz, 2,6-H) and 7.01 (2 H, AA'BB', d, *J* 8.4 Hz, 3,5-H) and 7.02 (2 H, AA'BB', d, *J* 8.8 Hz, 3,5-H);  $\delta_{\rm C}$  (100MHz, D<sub>2</sub>O) 16.6 and 17.7 (11-CH<sub>3</sub>), 29.5 and 30.8 (9-CH<sub>2</sub>), 39.5 and 41.6 (8-CH<sub>2</sub>), 58.9 and 59.2 (10-CH<sub>2</sub>), 116.4 and 117.2 (3,5-CH), 125.4 and 127.7 (4-quat. C), 128.3 and 128.7 (2,6-CH), 156.1 and 156.6 (1-quat. C) and 164.8 and 165.3 (7-quat. C); *m/z* (EI) 208.1210 (M<sup>+</sup>-HCI. C11H16N<sub>2</sub>O<sub>2</sub> requires 208.1209); *m/z* (%) 208 (26.5), 177 (15.6), 164 (17.0), 134 (51.8), 109 (100), 99 (28.4), 80 (25.5), 65 (9.2) and 57 (6.4).

N-(3-Hydroxypropyl)-N'-(4-hydroxyphenyl)propionamidine hydrochloride 104



Compound 104 was prepared from oxazine 62 (600 mg, 5.31 mmol) and 4-aminophenol hydrochloride (773 mg, 5.31 mmol) by the method used to prepare compound 102. Recrystallisation from isopropanol provided 104 (935 mg, 3.62 mmol, 68%) as a pale pink powder, mp 131-133 °C (Found: C, 55.74; H, 7.33; N, 10.79. C<sub>12</sub>H<sub>19</sub>N<sub>2</sub>O<sub>2</sub>Cl requires C, 55.71; H, 7.35; N, 10.83%);  $\lambda_{max}$  (H<sub>2</sub>O)/nm 223 ( $\epsilon$ /dm<sup>3</sup> mol<sup>-1</sup> cm<sup>-1</sup> 11070);  $\nu_{max}$  (thin film)/cm<sup>-1</sup> 3238-2883s (OH, NH), 1650vs (C=N), 1517s (C=C), 1271m (OH), 1068m (C-O) and 850w (CH); δ<sub>H</sub> (400 MHz, D<sub>2</sub>O) 0.94 (3 H, t, J 7.6 Hz, 12-H) and 1.23 (3 H, t, J 7.6 Hz, 12-H), 1.64 (2 H, m, 9-H) and 1.83 (2 H, m, 9-H), 2.30 (2 H, q, J 7.6 Hz, 11-H) and 2.61 (2 H, q, J 7.6 Hz, 11-H), 3.36 (2 H, m, 8-H), 3.48 (2 H, t, J 5.9 Hz, 10-H) and 3.60 (2 H, t, J 6.2 Hz, 10-H), 6.83 (2 H, AA'BB', d, J 8.8 Hz, 2,6-H) and 6.86 (2 H, AA'BB', d, J 8.8 Hz, 2,6-H) and 7.05 (2 H, AA'BB', d, J 8.6 Hz, 3,5-H) and 7.07 (2 H, AA'BB', d, J 8.6 Hz, 3,5-H); δ<sub>C</sub> (100 MHz, D<sub>2</sub>O) 10.1 and 10.6 (12-CH<sub>3</sub>), 23.5 and 24.3 (9-CH<sub>2</sub>), 29.5 and 31.0 (11-CH<sub>2</sub>), 39.5 and 41.5 (8-CH<sub>2</sub>), 59.0 and 59.4 (10-CH<sub>2</sub>), 116.5 and 117.3 (3,5-CH), 125.3 and 127.3 (4-quat. C), 128.4 and 129.0 (2,6-CH), 156.3 and 156.7 (1-quat. C) and 168.9 and 169.2 (7-quat. C); m/z (EI) 222.1367 (M<sup>+</sup>-HCI. C<sub>12</sub>H<sub>18</sub>N<sub>2</sub>O<sub>2</sub> requires 222.1365); m/z (%) 222 (8.2), 204 (9.9), 178 (5.7), 148 (28.4), 136 (7.1), 109 (100), 108 (14).
2-Cyano-3-(4-hydroxyphenyl)propenonitrile 121



Compound **121** was prepared in 81% yield on a 4.10 mmol scale by general method 2 and gave mp 181-182 °C (CHCl<sub>3</sub>) (lit.,<sup>213</sup> 180 °C). The spectroscopic data were in agreement with the IR and <sup>1</sup>H NMR data reported.<sup>249</sup>

Ethyl 2-Cyano-3-(4-hydroxyphenyl)propenoate 122



Compound **122** was prepared in 58% yield on a 4.09 mmol scale by general method 2 and gave mp 170-171 °C ( $^{i}$ PrOH) (lit.,<sup>249</sup> 169 °C). The spectroscopic data were in agreement with the IR and <sup>1</sup>H NMR data reported.<sup>249</sup>

2-Cyano-3-(4-hydroxyphenyl)propenamide 123



Compound **123** was prepared in 61% yield on a 4.10 mmol scale by general method 2 and gave mp 247-248 °C (CHCl<sub>3</sub>/MeOH) (lit.,<sup>213</sup> 250 °C). The spectroscopic data were in agreement with the IR and <sup>1</sup>H NMR data reported.<sup>249</sup>

3-Amino-2,4-dicyano-5-(4-hydroxyphenyl)penta-2,4-dienonitrile 124



Compound **124** was prepared in 19% yield on a 4.09 mmol scale by general method 2 and gave mp 223-224 °C (MeOH/EtOH) (lit.,<sup>213</sup> 225 °C). The spectroscopic data were in agreement with the IR and <sup>1</sup>H NMR data reported.<sup>249</sup>

2-Cyano-3-(3,4-dihydroxyphenyl)propenonitrile 125



Compound **125** was prepared in 59% yield on a 7.97 mmol scale by general method 1 and gave mp 185 °C (decomp.) (EtOH) (lit.,<sup>249</sup> 220 °C (decomp.)). The spectroscopic data were in agreement with the IR and <sup>1</sup>H NMR data reported.<sup>249</sup>

Ethyl 2-cyano-3-(3,4-dihydroxyphenyl)propenonitrile 126



Compound **126** was prepared in 43% yield on a 3.62 mmol scale by general method 1 and gave mp 162-165 °C (EtOH/H<sub>2</sub>O) (lit.,<sup>249</sup> 162 °C). The spectroscopic data were in agreement with the IR and <sup>1</sup>H NMR data reported.<sup>249</sup>

2-Cyano-3-(3,4-dihydroxyphenyl)propenamide 127



Compound **127** was prepared in 30% yield on a 3.62 mmol scale by general method 1 and gave mp 225-226 °C ( $^{i}$ PrOH) (lit.,<sup>249</sup> 221 °C). The spectroscopic data were in agreement with the IR and <sup>1</sup>H NMR data reported.<sup>249</sup>

3-Amino-2,4-dicyano-5-(3, 4-dihydroxyphenyl)penta-2,4-dienonitrile 128



Compound **128** was prepared in 25% yield on a 3.62 mmol scale by general method 1 and gave mp 227-229 °C (EtOH) (lit.,<sup>249</sup> 229 °C). The spectroscopic data were in agreement with the IR and <sup>1</sup>H NMR data reported.<sup>249</sup>

2-Cyano-3-(3-methoxy-4-hydroxyphenyl)propenonitrile 129



Compound **129** was prepared from vanillin (2.00 g, 13.1 mmol) and malononitrile (868 mg, 13.1 mmol) by general method 2. Recrystallisation from isopropanol provided the title compound **129** (2.14 g, 10.7 mmol, 81%) as pale yellow needles, mp 130-131 °C (lit.,<sup>250</sup> 133.5-134.5 °C). The spectroscopic data were in agreement with the IR and combustion data reported.<sup>180,250</sup>

New data:  $\lambda_{max}$  (MeOH)/nm 246, 371 and 438 ( $\epsilon$ /dm<sup>3</sup> mol<sup>-1</sup> cm<sup>-1</sup> 6060, 14050 and 5590);  $\delta_{H}$  (400 MHz, (CD<sub>3</sub>)<sub>2</sub>SO)) 3.79 (3 H, s, 11-H), 6.97 (1 H, d, *J* 8.3 Hz, 5-H), 7.47 (1 H, d, *J* 7.9 Hz, 6-H), 7.61 (1 H, s, 2-H) and 8.20 (1 H, s, 7-

H);  $\delta_{C}$  (100 MHz, (CD<sub>3</sub>)<sub>2</sub>SO)) 55.8 (11-CH<sub>3</sub>), 75.3 (8-quat.C), 113.3 (5-CH), 114.6 and 115.4 (9,10-CN), 116.4 (6-CH), 123.4 (1-quat.C), 128.0 (2-CH), 148.2 (4-quat.C), 154.2 (3-quat.C) and 161.0 (7-CH); *m/z* (EI) 200.0587 (M<sup>+•</sup>. C11H<sub>8</sub>N<sub>2</sub>O<sub>2</sub> requires 200.0588); *m/z* (%) 200 (100), 185 (12.0), 157 (58.5), 143 (2.8), 129 (16.9), 102 (22.5), 83 (8.5), 75 (7.0) and 51 (5.0).

Ethyl 2-cyano-3-(3-methoxy-4-hydroxyphenyl)propenonitrile 130



Compound **130** was prepared from vanillin (2.00 g, 13.1 mmol) and ethyl isocyanoacetate (1.40 mL, 1.49 g, 13.1 mmol) by general method 2. Recrystallisation from isopropanol provided the title compound **130** (3.06 g, 12.39 mmol, 94%) as a pale yellow powder, mp 105-107 °C (lit.,<sup>251</sup> 107 °C).

New data:  $\lambda_{max}$  (MeOH)/nm 245 and 360 ( $\epsilon$ /dm<sup>3</sup> mol<sup>-1</sup> cm<sup>-1</sup> 8160 and 19430);  $\nu_{max}$  (KBr disc)/cm<sup>-1</sup> 3374m (OH), 2986w (CH), 2941w (CH), 2220m (CN), 1702s (CO), 1573s and 1509s (C=C), 1277s (CH) and 1174w (C-O);  $\delta$ H (400 MHz, (CD<sub>3</sub>)<sub>2</sub>SO)) 1.29 (3 H, t, *J*7.1 Hz, 12-H), 3.82 (3 H, s, 13-H), 4.27 (2 H, q, *J*7.1 Hz, 11-H), 6.94 (1 H, d, *J* 8.3 Hz, 5-H), 7.58 (1 H, dd, *J* 2.0 and 8.4 Hz, 6-H), 7.73 (1 H, d, *J* 2.0 Hz, 2-H) and 8.18 (1 H, s, 7-H);  $\delta$ C (100 MHz, (CD<sub>3</sub>)<sub>2</sub>SO)) 14.3 (12-CH<sub>3</sub>), 55.9 (13-CH<sub>3</sub>), 62.2 (11-CH<sub>2</sub>), 97.3 (8-quat.C), 114.3 (5-CH), 116.3 (6-CH), 116.9 (9-CN), 123.2 (1-quat.C), 127.5 (2-CH), 148.1 (4-quat.C), 153.1 (3-quat.C), 155.2 (7-CH) and 162.9 (10-CO); *m/z* (EI) 247.0847 (M<sup>+•</sup>. C1<sub>3</sub>H1<sub>3</sub>NO4 requires 247.0850); *m/z* (%) 247 (100), 219 (27.5), 202 (15.5), 170 (30.3), 158 (7.0), 114 (9.2), 103 (6.3), 76 (5.6) and 51 (2.0).

2-Cyano-3-(3-methoxy-4-hydroxyphenyl)propenamide 131



Compound **131** was prepared from vanillin (2.00 g, 13.1 mmol) and 2cyanoacetamide (1.11 g, 13.1 mmol) by general method 2. Recrystallisation from methanol provided the title compound **131** (1.82 g, 8.35 mmol, 64%) as yellow crystals, mp 206-207 °C (lit.,<sup>252</sup> 209 °C). The spectroscopic data were in agreement with the IR and combustion data reported.<sup>180,252</sup>

New data:  $\lambda_{max}$  (MeOH)/nm 243 and 356 ( $\epsilon$ /dm<sup>3</sup> mol<sup>-1</sup> cm<sup>-1</sup> 8710 and 16590);  $\delta_{H}$  (400 MHz, (CD<sub>3</sub>)<sub>2</sub>SO)) 3.82 (3 H, s, 12-H), 6.94 (1 H, d, *J* 8.3 Hz, 5-H), 7.46 (1 H, dd, *J* 1.5 and 8.3 Hz, 6-H), 7.66 (1 H, d, *J* 1.4 Hz, 2-H), 7.68 (2 H, 2 br. s, 11-H) and 8.07 (1 H, s, 7-H);  $\delta_{C}$  (100 MHz, (CD<sub>3</sub>)<sub>2</sub>SO)) 55.9 (12-CH<sub>3</sub>), 101.8 (8-quat.C), 113.5 (5-CH), 116.3 (6-CH), 117.8 (9-CN), 123.6 (1-quat.C), 126.3 (2-CH), 148.1 (4-quat.C),151.2 (7-CH), 151.8 (3-quat.C) and 164.0 (10-CO); *m/z* (El) 218.0691 (M<sup>+•</sup>. C11H10N2O3 requires 218.0691); *m/z* (%) 218 (100), 202 (11.9), 170 (10.9), 158 (11.9), 114 (10.6), 76 (9.9) and 51 (6.3).

3-Amino-2,4-dicyano-5-(3-methoxy-4-hydroxyphenyl)penta-2,4-dienonitrile 132



Compound **132** was prepared from vanillin (2.00 g, 13.14 mmol) and malononitrile dimer (1.74 g, 13.14 mmol) by general method 2. Recrystallisation from ethanol provided the title compound **132** (999 mg, 3.76 mmol, 29%) as an orange powder, mp 220-221 °C ;  $\lambda_{max}$  (EtOH)/nm 253 and 372 ( $\epsilon$ /dm<sup>3</sup> mol<sup>-1</sup> cm<sup>-1</sup> 9350 and 11960);  $\nu_{max}$  (KBr disc)/cm<sup>-1</sup> 3396m (OH), 3396-3212m (NH), 2959w (CH), 2215m (CN), 1574m and 1512s (C=C), 1292s (CH) and 1131m (C-O);  $\delta_{C}$  (100 MHz, (CD<sub>3</sub>)<sub>2</sub>SO)) 55.9 (15-CH<sub>3</sub>), 97.2 (11-quat.C), 113.3 (5-CH), 115.4 (8-quat.C), 116.1 and 116.2 (9,12,13-CN), 116.4 (6-CH), 123.4 (1-quat.C), 127.0 (2-CH), 148.2 (4-quat.C), 152.6 (3-quat.C), 153.7 (7-CH) and 166.6 (10-quat.C); *m/z* (El) 266.0805 (M<sup>+•</sup>. C14H10N4O<sub>2</sub> requires 266.0806); *m/z* (%) 266 (100), 251 (12.0), 240 (47.2), 235 (19.7), 195 (15.5), 168 (28.9), 157 (6.3), 141 (14.1), 130 (5.6), 114 (12.0), 76 (7.7) and 51 (6.3).

# **6.3 EXPERIMENTAL FOR CHAPTER 5**

# 6.3.1 GENERAL PREPARATIVE METHODS 3 TO 6

### **General method 3**

The method of Baker and Howes<sup>182</sup> was modified as follows. To a stirred solution of hydroxybenzaldehyde (1 equiv.) and malononitrile derivative (1 equiv.) in ethanol (*ca.* 1 mL/mmol), neat piperidine (a few drops) was added. The reaction mixture was stirred at room temperature until TLC indicated the reaction to be complete whereupon the crude product was filtered, washed with cold ethanol and dried under suction. The iminolactone product required careful handling as it could not be recrystallised unchanged<sup>181,182</sup> and subsequently purification was achieved by trituration in cold ethanol.

## **General method 4**

The method of Gazit and co-workers<sup>213</sup> was modified as follows. To a stirred solution of methoxybenzaldehyde (1 equiv.) and malononitrile derivative (1 equiv.) in ethanol (*ca.* 4 mL/mmol), neat piperidine (a few drops) was added. The reaction mixture was heated at reflux until TLC indicated the reaction to be complete whereupon the mixture was cooled to room temperature. If precipitation of product was not observed dropwise addition of water was employed. The precipitated product was then filtered, dried under suction and recrystallised from an appropriate solvent

#### **General method 5**

General method 5 followed the same experimental protocol outlined for general method 4 with the exception of stirring the reaction mixture at room temperature rather than heating under reflux.

#### **General method 6**

The method of Cabello and workers<sup>181</sup> was used. A solution of iminolactone in dilute hydrochloric acid (4 M, *ca.* 15 mL/mmol) was heated at 60 °C. The reaction's progress was monitored by TLC and on reaching completion, the mixture was cooled to room temperature. The precipitated product was filtered, dried under suction and recrystallised from an appropriate solvent.

# 6.3.2 EXPERIMENTAL DATA

2-Imino-2H-1-benzopyran-3-carbonitrile 136



Compound **136** was prepared from salicylaldehyde (0.50 mL, 573 mg, 4.69 mmol) and malononitrile (310 mg, 4.69 mmol) by general method 3. Trituration in cold ethanol provided the title compound **136** (495 mg, 2.91 mmol, 62%) as a yellow powder, mp 160 °C (decomp.) (lit.,<sup>181</sup> mp 160 °C (decomp.)). The spectroscopic data were in agreement with the IR and combustion data reported.<sup>181,182</sup>

New data:  $\delta_{\text{H}}$  (200 MHz; (CD<sub>3</sub>)<sub>2</sub>SO)) 7.12-7.34 (2 H, m, 6,8-H), 7.59-7.66 (2 H, m, 5,7-H), 8.40 (1 H, s, 11-H) and 8.89 (1 H, s, 4-H);  $\delta_{\text{C}}$  (50 MHz; (CD<sub>3</sub>)<sub>2</sub>SO)) 104.2 (3-quat. C), 115.1 (9-CN), 117.5 (10-quat. C), 115.6 (8-CH), 124.3 (6-CH), 129.7 (7-CH), 134.2 (5-CH), 148.0 (4-CH), 151.6 (9-quat. C) and 153.9 (2-quat. C); *m/z* (CI/NH<sub>3</sub>) 171.0559 (MH<sup>+</sup>. C<sub>10</sub>H<sub>6</sub>N<sub>2</sub>O + H<sup>+</sup> requires 171.0560); *m/z* (CI/NH<sub>3</sub>) (%) 171 (100), 152 (1.0), 136 (2.5), 113 (4.0), 96 (39.4) and 86 (2.0).

2-Imino-2H-1-benzopyran-3-carboxamide 137



Compound **137** was prepared in 75% yield on a 4.92 mmol scale by general method 3 and gave mp 175-176 °C (lit.,<sup>253</sup> 178 °C). The spectroscopic data were in agreement with the IR and <sup>1</sup>H and <sup>13</sup>C NMR data reported.<sup>180,254</sup>



Compound **142** was prepared from *o*-vanillin (600 mg, 3.94 mmol) and malononitrile (261 mg, 3.94 mmol) by general method 3. Trituration in cold ethanol provided the title compound **142** (660 mg, 3.30 mmol, 84%) as a mustard yellow powder, mp 176-177 °C (lit.,<sup>219</sup> 172-174 °C). The spectroscopic data were in agreement with the IR and combustion data reported.<sup>219</sup>

New data:  $\delta_{H}$  (400 MHz; (CD<sub>3</sub>)<sub>2</sub>SO)) 3.88 (3 H, s, 11-H), 7.13 (1 H, dd, *J* 1.4 and 7.8 Hz, 7-H), 7.20 (1 H, t, *J* 8.0 Hz, 6-H), 7.31 (1 H, dd, *J* 1.2 and 8.0 Hz, 5-H), 8.35 (1 H, s, 12-H) and 8.95 (1 H, s, 4-H);  $\delta_{C}$  (100 MHz; (CD<sub>3</sub>)<sub>2</sub>SO)) 56.4 (11-CH<sub>3</sub>), 105.1 (3-quat. C), 115.6 (13-CN), 117.1 (7-CH), 118.2 (10-quat. C), 121.0 (6-CH), 124.5 (5-CH), 138.0 (9-quat. C), 146.4 (8-quat. C), 147.5 (4-CH) and 151.4 (2-quat. C); *m/z* (EI) 200.0585 (M<sup>+•</sup>. C11H8N<sub>2</sub>O<sub>2</sub> requires 200.0584); *m/z* (%) 200 (100), 185 (14.2), 157 (19.9), 129 (14.9), 102 (14.2), 84 (23.8), 79 (14.2) and 63 (13.5).

8-Methoxy-2-imino-2H-1-benzopyran-3-carboxamide 143



Compound **143** was prepared from *o*-vanillin (600 mg, 3.94 mmol) and 2cyanoacetamide (332 mg, 3.94 mmol) by general method 3. Trituration in cold ethanol provided the title compound **143** (735 mg, 3.37 mmol, 86%) as a grey powder, mp 166-167 °C (lit.,<sup>255</sup> 195 °C). The spectroscopic data were in agreement with the IR and <sup>1</sup>H and <sup>13</sup>C NMR data reported.<sup>254,255</sup>



Compound **144** was prepared from 4-methoxysalicylaldehyde (300 mg, 1.97 mmol) and malononitrile (130 mg, 1.97 mmol) by general method 3. Trituration in cold ethanol provided the title compound **144** (340 mg, 1.70 mmol, 86%) as a yellow powder, mp 143-145 °C (lit.,<sup>182</sup> 162-165 °C).

New data:  $v_{max}$  (KBr disc)/cm<sup>-1</sup> 3436m and 3286m (NH), 3025w and 2938w (CH), 2222m (CN), 1617s and 1553m (C=C), 1278m (C-O) and 852m (CH);  $\delta_{H}$  (400 MHz; (CD<sub>3</sub>)<sub>2</sub>SO)) 3.91 (3 H, s, 11-H), 6.81 (1 H, d, *J* 2.1 Hz, 8-H), 6.93 (1 H, dd, *J* 2.4 and 8.6 Hz, 6-H), 7.57 (1 H, d, *J* 8.6 Hz, 5-H), 8.34 (1 H, s, 12-H) and 8.75 (1 H, s, 4-H);  $\delta_{C}$  (100 MHz; (CD<sub>3</sub>)<sub>2</sub>SO)) 56.4 (11-CH<sub>3</sub>), 100.5 (3-quat. C), 100.9 (8-CH), 111.2 (10-quat. C), 111.3 (6-CH), 116.1 (13-CN), 131.1 (5-CH), 147.2 (4-CH), 152.1 (9-quat. C), 156.0 (7-quat. C) and 164.5 (2-quat. C); *m/z* (EI) 200.0587 (M<sup>+•</sup>. C11H8N<sub>2</sub>O<sub>2</sub> requires 200.0588); *m/z* (%) 200 (100), 173 (68.3), 158 (30.2), 149 (4.3), 130 (5.8), 102 (10.8), 84 (8.6), 63 (5.8) and 57 (3.6).

6-Methoxy-2-imino-2H-1-benzopyran-3-carbonitrile 49



Compound **49** was prepared from 5-methoxysalicylaldehyde (0.41 mL, 500 mg, 3.28 mmol) and malononitrile (217 mg, 3.28 mmol) by general method 3. Trituration in cold ethanol provided the title compound **49** (405 mg, 2.03 mmol, 62%) as a yellow powder, mp 184-185 °C.  $v_{max}$  (KBr disc)/cm<sup>-1</sup> 3456m and 3344m (NH), 2924w (CH), 2218m (CN), 1573m and 1495m (C=C) and 1211s (C-O);  $\delta_{\rm H}$  (400 MHz; (CD<sub>3</sub>)<sub>2</sub>SO)) 3.80 (3 H, s, 11-H), 7.15 (3 H, m, 5,7,8-H), 8.28 (1 H, s, 12-H) and 8.74 (1 H, s, 4-H);  $\delta_{\rm C}$  (100 MHz; (CD<sub>3</sub>)<sub>2</sub>SO)) 55.9 (11-CH<sub>3</sub>), 104.7 (3-quat. C), 112.2 (7-CH), 115.4 (13-CN), 116.7 (5-CH), 121.0 (8-CH), 146.9 (4-CH), 148.1 (9-quat. C), 151.9 (6-quat. C) and 155.3 (2-

quat. C); m/z (EI) 200.0595 (M<sup>+•</sup>. C<sub>11</sub>H<sub>8</sub>N<sub>2</sub>O<sub>2</sub> requires 200.0604); m/z (%) 200 (100), 185 (22.7), 173 (10.3), 158 (16.5), 130 (6.2), 102 (11.3), 78 (48.3), 63 (45.4) and 61 (7.2).

6-Methoxy-2-imino-2H-1-benzopyran-3-carboxamide 145



Compound **145** was prepared from 5-methoxysalicylaldehyde (0.20 mL, 250 mg, 1.64 mmol) and 2-cyanoacetamide (138 mg, 1.64 mmol) by general method 3. Trituration in cold ethanol provided the title compound **145** (266 mg, 1.22 mmol, 74%) as a pale yellow powder, mp 193-194 °C (lit.,<sup>218</sup> mp 192-194 °C). The spectroscopic data were in agreement with the <sup>1</sup>H and <sup>13</sup>C NMR data reported.<sup>254</sup>

New data:  $v_{max}$  (KBr disc)/cm<sup>-1</sup> 3280s (NH, NH<sub>2</sub>), 2950-2759w (CH), 1681s (CO), 1601m and 1573s (C=C), 1287m (C-O) and 853m (CH); *m/z* (EI) 218.0690 (M<sup>+•</sup>. C<sub>11</sub>H<sub>10</sub>N<sub>2</sub>O<sub>3</sub> requires 218.0689); *m/z* (%) 218 (54.6), 201 (100), 186 (19.1), 158 (21.3), 148 (3.5), 130 (5.0), 102 (4.3), 79 (48.2), 63 (47.5) and 61 (7.1).

5-Methoxy-2-imino-2H-1-benzopyran-3-carbonitrile 146



Compound **146** was prepared from 5-methoxysalicylaldehyde (600 mg, 3.94 mmol) and malononitrile (260 mg, 3.94 mmol) by general method 3. Trituration in cold ethanol provided the title compound **146** (528 mg, 2.64 mmol, 67%) as a salmon pink powder, mp 182-183 °C.  $v_{max}$  (KBr disc)/cm<sup>-1</sup> 3433w and 3276m (NH), 2231m (CN), 1673s and 1608s (C=C), 1272m (C-O) and 785s (CH);  $\delta_{\rm H}$  (400 MHz; (CD<sub>3</sub>)<sub>2</sub>SO)) 3.90 (3 H, s, 11-H), 6.73 (1 H, d, *J* 8.4 Hz, 8-

H), 7.52 (1 H, t, *J* 8.4 Hz, 7-H), 8.21 (1 H, s, 12-H) and 8.77 (1 H, s, 4-H);  $\delta_{\rm C}$  (100 MHz; (CD<sub>3</sub>)<sub>2</sub>SO)) 56.8 (11-CH<sub>3</sub>), 102.7 (3-quat. C), 106.6 (8-CH), 107.7 (11-CN), 108.1 (6-CH), 115.7 (10-quat. C), 135.6 (7-CH), 141.3 (4-CH), 151.7 (9-quat. C), 154.8 (5-quat. C) and 156.7 (2-quat. C); *m/z* (EI) 200.0586 (M<sup>+•</sup>. C11H8N<sub>2</sub>O<sub>2</sub> requires 200.0587); *m/z* (%) 200 (100), 173 (24.1), 158 (13.5), 135 (18.4), 102 (12.1), 76 (5.0), 63 (2.8) and 52 (2.8).

5-Methoxy-2-imino-2H-1-benzopyran-3-carboxamide 147



Compound **147** was prepared from 5-methoxysalicylaldehyde (395 mg, 2.60 mmol) and 2-cyanoacetamide (218 mg, 2.60 mmol) by general method 3. Trituration in cold ethanol provided the title compound **147** (420 mg, 1.93 mmol, 74%) as a cream solid, mp 210-212 °C.  $v_{max}$  (KBr disc)/cm<sup>-1</sup> 3268-3073m (NH, NH<sub>2</sub>), 2945w and 2843w (CH), 1672s (CO), 1603s and 1571m (C=C), 1288m (C-O) and 773m (CH);  $\delta_{H}$  (400 MHz; (CD<sub>3</sub>)<sub>2</sub>SO)) 3.96 (3 H, s, 11-H), 6.78 (1 H, d, *J* 8.2 Hz, 8-H), 6.87 (1 H, d, *J* 8.3 Hz, 6-H), 7.50 (1 H, t, *J* 8.3 Hz, 7-H), 7.76 (1 H, s, 4-H), 8.39 (2 H, 2 br. s, 14-H) and 9.51 (1 H, s, 12-H);  $\delta_{C}$  (100 MHz; (CD<sub>3</sub>)<sub>2</sub>SO)) 56.7 (11-CH<sub>3</sub>), 106.2 (8-CH), 107.6 (6-CH), 108.6 (3-quat. C), 119.4 (10-quat. C), 134.2 (7-CH), 135.5 (4-CH), 154.8 (9-quat. C), 155.6 (5-quat. C), 157.1 (2-quat. C) and 163.3 (13-quat. C); *m/z* (EI) 218.0690 (M<sup>+•</sup>. C11H10N2O3 requires 218.0689); *m/z* (%) 218 (20.6), 201 (30.5), 187 (3.5), 175 (10.6), 149 (14.2), 137 (6.4), 96 (5.7), 79 (97.9), 63 (100) and 61 (14.9).

2-Cyano-3-(2-methoxyphenyl)propenonitrile 150



Compound **150** was prepared in 43% yield on a 5.88 mmol scale by general method 5 and gave mp 80-81 °C ( $^{i}$ PrOH) (lit.,<sup>250</sup> 84-85 °C). The spectroscopic data were in agreement with the IR and <sup>1</sup>H and <sup>13</sup>C NMR data reported.<sup>222</sup>

2-Cyano-3-(2-methoxyphenyl)propenamide 151



Compound **151** was prepared from *o*-anisaldehyde (800 mg, 5.88 mmol) and 2-cyanoacetamide (494 mg, 5.88 mmol) by general method 4. Recrystallisation from ethyl acetate/methanol provided the title compound **151** (550 mg, 2.72 mmol, 46%) as pale yellow crystals, mp 157-158 °C (lit.,<sup>222</sup> 151-153 °C).The spectroscopic data were in agreement with the IR and <sup>1</sup>H and <sup>13</sup>C NMR data reported.<sup>222</sup>

2-Cyano-3-(2,3-dimethoxyphenyl)propenonitrile 152



Compound **152** was prepared from 2,3-dimethoxybenzaldehyde (1.0 g, 6.02 mmol) and malononitrile (398 mg, 6.02 mmol) by general method 5. Recrystallisation from methanol provided the title compound **152** (881 mg, 4.12 mmol, 68%) as pale yellow needles, mp 98-100 °C (Found: C, 67.14; H, 4.63; N, 13.01. C12H10N2O2 requires C, 67.29; H, 4.67; N, 13.08%); v<sub>max</sub> (KBr disc)/cm<sup>-1</sup> 3048-2841w (CH), 2225m (CN), 1570s and 1481s (C=C), 1281s (C-O) and 751m (CH);  $\delta_{\rm H}$  (400 MHz; (CD3)2SO)) 3.86 and 3.87 (two s, 6 H, 11-and 12-H), 7.29 (1 H, d, *J* 8.0 Hz, 4-H), 7.40 (1 H, dd, *J* 1.2 and 8.0 Hz, 5-H), 7.60 (I H, dd, *J* 1.2 and 8.0 Hz, 6-H) and 8.51 (1 H, s, 7-H);  $\delta_{\rm C}$  (100 MHz, (CD3)2SO)) 56.4 and 62.0 (11 and 12-CH3), 83.2 (8-quat. C), 113.4 and 114.6

(9,10-CN), 119.2 (4-CH), 119.6 (5-CH), 125.0 (6-CH), 125.3 (1-quat. C), 149.2 and 152.9 (2 and 3-quat. C) and 156.7 (7-CH); m/z (EI) 214.0743 (M<sup>+•</sup>. C<sub>12</sub>H<sub>10</sub>N<sub>2</sub>O<sub>2</sub> requires 214.0744); m/z (%) 214 (100), 199 (66.0), 184 (4.3), 171 (18.4), 169 (11.3), 130 (15.6), 101 (14.9), 84 (5.7), 75 (5.0) and 51 (3.5).

2-Cyano-3-(2,3-dimethoxyphenyl)propenamide 153



Compound **153** was prepared from 2,3-dimethoxybenzaldehyde (1.0 g, 6.02 mmol) and 2-cyanoacetamide (506 mg, 6.02 mmol) by general method 5. Recrystallisation from isopropanol provided the title compound **153** (981 mg, 4.23 mmol, 70%) as a lemon fluffy solid, mp 164-166 °C (Found: C, 62.02; H, 5.31; N, 12.04. C1<sub>2</sub>H<sub>12</sub>N<sub>2</sub>O<sub>3</sub> requires C, 62.07; H, 5.17; N, 12.07%); vmax (KBr disc)/cm<sup>-1</sup> 3470s and 3142m (NH), 2947w (CH), 2216m (CN), 1750s (CO), 1603m and 1576m (C=C), 1277s (C-O) and 745m (CH);  $\delta$ H (400 MHz; (CD<sub>3</sub>)<sub>2</sub>SO)) 3.81 and 3.85 (6 H, two s, 12 and 13-H), 7.25 (1 H, d, *J* 8.0 Hz, 4-H), 7.29 (1 H, dd, *J* 1.2 and 8.4 Hz, 5-H), 7.62 (1 H, dd, *J* 1.2 and 7.6 Hz, 6-H), 7.91 (2 H, two br. s, 11-H) and 8.34 (1 H, s, 7-H);  $\delta$ C (100 MHz, (CD<sub>3</sub>)<sub>2</sub>SO)) 56.3 and 61.6 (12 and 13-CH<sub>3</sub>), 108.3 (8-quat. C), 116.6 (4-CH), 117.2 (9-CN), 119.8 (5-CH), 124.8 (6-CH), 126.1 (1-quat. C), 145.9 (7-CH), 148.8 and 152.9 (2 and 3-quat. C) and 162.9 (10-quat. C); *m/z* (EI) 232.0849 (M<sup>+•</sup>. C1<sub>2</sub>H<sub>12</sub>N<sub>2</sub>O<sub>3</sub> requires 232.0850); *m/z* (%) 232 (21.3), 201 (100), 186 (7.1), 174 (9.2), 159 (8.5), 149 (5.3), 102 (4.3), 84 9 (12.1) and 51 (4.3).

2-Cyano-3-(2,4-dimethoxyphenyl)propenonitrile 154



Compound **154** was prepared from 2,4-dimethoxybenzaldehyde (200 mg, 1.20 mmol) and malononitrile (80 mg, 1.20 mmol) by general method 5. Recrystallisation from isopropanol provided the title compound **154** (206 mg, 0.96 mmol, 80%) as yellow crystals, mp 139-141 °C (lit.,<sup>256</sup> 142.2-142.6 °C). The spectroscopic data were in agreement with the <sup>1</sup>H NMR and combustion data reported.<sup>256</sup>

New data:  $v_{max}$  (KBr disc)/cm<sup>-1</sup> 2924w (CH), 2223m (CN), 1609m and 1563m (C=C) and 1281s (C-O);  $\delta_{C}$  (100 MHz; (CD3)2SO)) 56.5 (12-CH3), 56.6 (11-CH3), 76.1 (8-quat. C), 98.6 (3-CH), 108.1 (5-CH), 113.4 (1-quat. C), 114.6 and 115.6 (9,10-CN), 130.8 (6-CH), 154.1 (7-CH), 161.7 (4-quat. C) and 167.2 (2-quat. C); *m/z* (EI) 214.0747 (M<sup>+•</sup>. C<sub>12</sub>H<sub>10</sub>N<sub>2</sub>O<sub>2</sub> requires 214.0751); *m/z* (%) 214 (100), 199 (4.9), 186 (10.7), 171 (9.9), 149 (30.7), 128 (8.9), 121 (21.8), 101 (8.9), 89 (4.9) and 63 (3.9).

2-Cyano-3-(2,4-dimethoxyphenyl)propenamide 155



Compound **155** was prepared from 2,4-dimethoxybenzaldehyde (200 mg, 1.20 mmol) and 2-cyanoacetamide (101 mg, 1.20 mmol) by general method 5. Recrystallisation from acetone provided the title compound **155** (140 mg, 0.60 mmol, 80%) as pale yellow crystals, mp 198-200 °C (Found: C, 62.09; H, 5.18; N, 12.08. C1<sub>2</sub>H1<sub>2</sub>N<sub>2</sub>O<sub>3</sub> requires C, 62.07; H, 5.17; N, 12.07%); v<sub>max</sub> (KBr disc)/cm<sup>-1</sup> 3461m and 3356m (NH<sub>2</sub>), 2931w (CH), 2210m (CN), 1667s (CO), 1612m and 1570s (C=C) and 1274s (C-O);  $\delta$ H (400 MHz; (CD<sub>3</sub>)<sub>2</sub>SO)) 3.86 (3 H, s, 12-H), 3.89 (3 H, s, 13-H), 6.68 (1 H, s, 3-H), 6.72 (1 H, d, *J* 8.8 Hz, 4-H), 7.70 (2 H, 2 br. s, 11-H), 8.10 (1 H, d, *J* 8.0 Hz, 6-H) and 8.36 (1 H, s, 7-H);  $\delta$ C (100 MHz, (CD<sub>3</sub>)<sub>2</sub>SO)) 56.1 (12-CH<sub>3</sub>), 56.4 (13-CH<sub>3</sub>), 98.7 (3-CH), 102.4 (8-quat. C), 106.9 (5-CH), 113.6 (1-quat. C), 117.7 (9-CN), 129.9 (6-CH), 145.1 (7-CH), 160.8 (2-quat. C), 163.4 (4-quat. C) and 165.0 (10-CO); *m/z* (EI) 232.0836 (M<sup>+•</sup>. C1<sub>2</sub>H1<sub>2</sub>N<sub>2</sub>O<sub>3</sub> requires 232.0825); *m/z* (%) 232 (61.3), 201 (100), 186 (6.9), 173 (6.9), 158 (7.9), 149 (5.9), 124 (4.4), 121 (6.9), 109 (5.9), 102 (4.9), 77 (3.9), 76 (3.9), 63 (2.9) and 51 (1.9).

2-Cyano-3-(2,5-dimethoxyphenyl)propenonitrile 156



Compound **156** was prepared from 2,5-dimethoxybenzaldehyde (200 mg, 1.20 mmol) and malononitrile (80 mg, 1.20 mmol) by general method 5. Recrystallisation from isopropanol provided the title compound **156** (146 mg, 0.68 mmol, 57%) as bright yellow crystals, mp 105-106°C (lit.,<sup>256</sup> 107.4-108.2 °C). The spectroscopic data were in agreement with the IR and <sup>1</sup>H NMR data reported.<sup>256,257</sup>

New data:  $\delta_{C}$  (100 MHz, (CD<sub>3</sub>)<sub>2</sub>SO)) 55.9 and 56.7 (11,12-CH<sub>3</sub>), 81.8 (8-quat. C), 112.7 (4-CH), 113.7 and 114.7 (9,10-CN), 114.0 (3-CH), 120.3 (1-quat. C), 123.1 (6-CH), 153.0 and 153.6 (2- and 5-quat. C) and 155.7 (7-CH); *m/z* (EI) 214.0740 (M<sup>+•</sup>. C<sub>12</sub>H<sub>10</sub>N<sub>2</sub>O<sub>2</sub> requires 214.0738); *m/z* (%) 214 (100), 199 (91.1), 184 (9.4), 171 (10.4), 158 (2.9), 143 (3.5), 128 (7.4), 116 (11.4), 101 (7.9), 89 (4.9), 75 (3.9) and 54 (2.9).

2-Cyano-3-(2,5-dimethoxyphenyl)propenamide 157



Compound **157** was prepared from 2,5-dimethoxybenzaldehyde (100 mg, 0.60 mmol) and 2-cyanoacetamide (51 mg, 0.60 mmol) by general method 5. Recrystallisation from ethanol provided the title compound **157** (120 mg, 0.48 mmol, 81%) as fluorescent yellow crystals, mp 193-194 °C (Found: C, 62.01; H, 5.14; N, 12.02. C<sub>12</sub>H<sub>10</sub>N<sub>2</sub>O<sub>3</sub> requires C, 62.07; H, 5.17; N, 12.07%); v<sub>max</sub> (KBr disc)/cm<sup>-1</sup> 3391m (NH), 2966-2923w (CH), 2213m (CN), 1698s (CO), 1585m and 1497s (C=C) and 1228s (C-O);  $\delta_{\rm H}$  (400 MHz; (CD<sub>3</sub>)<sub>2</sub>SO)) 3.75 (3 H, s, 12-H), 3.84 (3 H, s, 13-H), 7.12 (1 H, d, *J* 9.1 Hz, 3-H), 7.17 (1 H, dd, *J* 9.1 and 2.8 Hz, 4-H), 7.60 (1 H, d, *J* 2.6 Hz, 6-H), 7.85 (2 H, 2 br. s, 11-H) and 8.38

(1 H, s, 7-H);  $\delta_{C}$  (100 MHz, (CD<sub>3</sub>)<sub>2</sub>SO)) 55.8 and 56.6 (12,13-CH<sub>3</sub>), 106.9 (8quat. C), 113.1 and 113.4 (4,6-CH), 116.9 (9-CN), 120.1 (3-CH), 121.1 (1-quat. C), 145.7 (7-CH), 153.0 and 153.1 (2- and 5-quat. C) and 162.9 (10-CO); *m/z* (EI) 232.0841 (M<sup>+•</sup>. C<sub>12</sub>H<sub>10</sub>N<sub>2</sub>O<sub>3</sub> requires 232.0834); *m/z* (%) 232 (53.5), 201 (100), 186 (4.9), 174 (31.7), 159 (8.9), 130 (8.8), 102 (6.9), 76 (5.9) and 63 (3.0).

OMe

2-Cyano-3-(2,6-dimethoxyphenyl)propenonitrile 158



MeC

New data:  $v_{max}$  (KBr disc)/cm<sup>-1</sup> 2982-2845w (CH), 2226m (CN), 1598s and 1482m (C=C) and 1270m (C-O);  $\delta_{\rm H}$  (400 MHz; (CD<sub>3</sub>)<sub>2</sub>SO)) 3.35 (6 H, s, 11,12-H), 6.79 (2 H, d, *J* 8.5 Hz, 3,5-H), 7.59 (1 H, t, *J* 8.5 Hz, 4-H) and 8.23 (1 H, s, 7-H);  $\delta_{\rm C}$  (100 MHz, (CD<sub>3</sub>)<sub>2</sub>SO)) 56.2 (11,12-CH<sub>3</sub>), 87.8 (8-quat. C), 104.5 (3,5-CH), 109.7 (1-quat. C), 113.3 and 115.3 (9,10-CN), 136.6 (4-CH), 154.2 (7-CH) and 159.3 (2,6-quat. C); *m/z* (EI) 214.0744 (M<sup>+•</sup>. C1<sub>2</sub>H<sub>10</sub>N<sub>2</sub>O<sub>2</sub> requires 214.0745); *m/z* (%) 214 (100), 195 (8.6), 186 (12.1), 156 (7.9), 149 (66.4), 128 (10.7), 91 (12.1), 78 (4.6) and 63 (5.0).

2-Cyano-3-(2,6-dimethoxyphenyl)propenamide 159



Compound **159** was prepared from 2,6-dimethoxybenzaldehyde (1.1 g, 6.62 mmol) and 2-cyanoacetamide (557 mg, 6.62 mmol) by general method 4. Recrystallisation from ethanol provided the title compound **159** (940 mg, 4.05 mmol, 61%) as cream crystals, mp 135-137 °C;  $v_{max}$  (KBr disc)/cm<sup>-1</sup> 3440m, 3396m and 3177m (NH<sub>2</sub>), 2978-2841w (CH), 2212m (CN), 1692s (CO), 1599s and 1477s (C=C) and 1262s (C-O);  $\delta_{\rm H}$  (400 MHz; (CD<sub>3</sub>)<sub>2</sub>SO)) 3.84 (6 H, s, 12,13-), 6.76 (2 H, d, *J* 8.5 Hz, 3,5-H), 7.46 (1 H, t, *J* 8.4 Hz, 4-H), 7.78 (2 H, 2 br. s, 11-H) and 8.08 (1 H, s, 7-H);  $\delta_{\rm C}$  (100 MHz, (CD<sub>3</sub>)<sub>2</sub>SO)) 56.0 and 56.1 (12,13-CH<sub>3</sub>), 104.4 (3,5-CH), 110.4 (1-quat. C), 116.2 (9-CN), 133.6 (4-CH), 148.6 (7-CH), 158.5 (2,6-quat. C) and 163.1 (10-CO); *m/z* (EI) 232.0847 (M<sup>+•</sup>. C1<sub>2</sub>H1<sub>0</sub>N<sub>2</sub>O<sub>3</sub> requires 232.0846); *m/z* (%) 232 (15.8), 201 (100), 186 (12.9), 158 (5.0), 149 (4.3), 130 (2.9), 92 (4.0), 77 (2.5) and 63 (1.4).

2-Oxo-2H-1-benzopyran-3-carbonitrile 174



Compound **174** was prepared from iminolactone **136** (225 mg, 1.32 mmol) by general method 6. Recrystallisation from ethanol provided the title compound **174** (193 mg, 1.13 mmol, 86%) as cream crystals, mp 184-185 °C (lit.,<sup>182</sup> mp 184-185 °C (EtOH)). The spectroscopic data were in agreement with the IR and combustion data reported.<sup>181,182</sup>

New data:  $\delta_{H}$  (400 MHz; (CD<sub>3</sub>)<sub>2</sub>SO)) 7.49 (2 H, m, 6,8-H), 7.81 (2 H, m, 5,7-H) and 8.95 (1 H, s, 4-H);  $\delta_{C}$  (100 MHz; (CD<sub>3</sub>)<sub>2</sub>SO)) 102.5 (3-quat. C), 114.9 (11-CN), 117.9 (10-quat. C), 117.1 (8-CH), 125.8 (6-CH), 130.3 (5-CH), 135.8 (7-CH), 153.8 (4-CH), 154.4 (9-quat. C) and 157.2 (2-quat. C); *m/z* (EI) 171.0319 (M<sup>+•</sup>. C<sub>10</sub>H<sub>5</sub>NO<sub>2</sub> requires 171.0317); *m/z* (%) 171 (100), 144 (7.9), 143 (75.7), 114 (33.2), 113 (10.0), 89 (12.1), 88 (4.3), 63 (7.9) and 62 (7.1).



Compound **175** was prepared from iminolactone **137** (390 mg, 2.07 mmol) by general method 6. Recrystallisation from ethanol provided the title compound **175** (350 mg, 1.85 mmol, 89%) as white fluffy needles, mp 264-265 °C (lit.,<sup>253</sup> 258-260 °C). The spectroscopic data were in agreement with the IR and <sup>1</sup>H NMR data reported.<sup>258</sup>

New data: δ<sub>C</sub> (100 MHz; (CD<sub>3</sub>)<sub>2</sub>SO)) 116.5 (8-CH), 118.8 (3-quat. C), 119.6 (10-quat. C), 125.4 (6-CH), 130.6 (5-CH), 134.4 (7-CH), 148.1 (4-CH), 154.4 (9-quat. C), 160.7 (2-quat. C) and 162.9 (11-quat. C).

8-Methoxy-2-oxo-2H-1-benzopyran-3-carbonitrile 176



Compound **176** was prepared from iminolactone **142** (200 mg, 1.00 mmol) by general method 6. Recrystallisation from methanol provided the title compound **176** (165 mg, 0.89 mmol, 89%) as yellow fluffy needles, mp 224-225 °C (lit.,<sup>259</sup> 222-223 °C). The spectroscopic data were in agreement with the IR and combustion data reported.<sup>259</sup>

New data:  $\delta_{\text{H}}$  (400 MHz; (CD<sub>3</sub>)<sub>2</sub>SO)) 3.93 (3 H, s, 11-H), 7.33 (1 H, dd, *J* 1.4 and 7.8 Hz, 7-H), 7.40 (1 H, m, 6-H), 7.49 (1 H, dd, *J* 1.4 and 8.1 Hz, 5-H) and 8.92 (1 H, s, 4-H);  $\delta_{\text{C}}$  (100 MHz; (CD<sub>3</sub>)<sub>2</sub>SO)) 56.6 (11-CH<sub>3</sub>), 102.7 (3-quat. C), 114.9 (12-CN), 117.8 (7-CH), 118.3 (10-quat. C), 121.1 (6-CH), 125.8 (5-CH), 143.6 (9-quat. C), 146.8 (8-quat. C), 154.0 (4-CH) and 156.9 (2-quat. C); *m*/*z* (EI) 201.0424 (M<sup>+•</sup>. C<sub>11</sub>H7NO<sub>3</sub> requires 201.0422); *m*/*z* (%) 201 (100), 186 (9.9), 158 (22.8), 130 (22.1), 102 (21.0), 76 (9.9) and 52 (7.2).



Compound **177** was prepared from iminolactone **143** (320 mg, 1.47 mmol) by general method 6. Recrystallisation from methanol provided the title compound **177** (189 mg, 0.86 mmol, 59%) as a pale yellow solid, mp 239-241 °C (lit.,<sup>255</sup> 247 °C). The spectroscopic data was in agreement with the combustion data reported.<sup>255</sup>

New data:  $v_{max}$  (KBr disc)/cm<sup>-1</sup> 3460m and 3392m (NH<sub>2</sub>), 3040-2947w (CH), 1710s (CO), 1611s and 1591s (C=C), 1281s (C-O) and 799s (CH);  $\delta_{\rm H}$  (400 MHz; (CD<sub>3</sub>)<sub>2</sub>SO)) 3.94 (3 H, s, 11-H), 7.36 (1 H, m, 6-H), 7.42 (1 H, d, *J* 8.0 Hz, 7-H), 7.50 (1 H, d, *J* 7.6 Hz, 5-H), 8.00 (2 H, 2 br. s, 13-H) and 8.82 (1 H, s, 4-H);  $\delta_{\rm C}$  (100 MHz; (CD<sub>3</sub>)<sub>2</sub>SO)) 56.5 (11-CH<sub>3</sub>), 116.4 (7-CH), 119.3 (3-quat. C), 119.7 (10-quat. C), 121.5 (5-CH), 125.3 (6-CH), 143.9 (9-quat. C), 146.6 (8-quat. C), 148.3 (4-CH), 160.4 (2-quat C) and 162.8 (12-quat. C); *m/z* (EI) 219.0533 (M<sup>+•</sup>. C11H9NO4 requires 219.0535); *m/z* (%) 219 (100), 230 (30.1), 176 (15.6), 148 (9.9), 133 (14.2), 105 (11.3), 90 (7.1), 78 (10.6) and 52 (7.1).

7-Methoxy-2-oxo-2H-1-benzopyran-3-carbonitrile 178



Compound **178** was prepared from iminolactone **144** (350 mg, 1.75 mmol) by general method 6. Recrystallisation from methanol/acetone provided the title compound **178** (320 mg, 1.59 mmol, 91%) as pale yellow needles, mp 217-219 °C (lit.,<sup>259</sup> 218-219 °C). The spectroscopic data were in agreement with the IR and combustion data reported.<sup>259</sup>

New data: δ<sub>H</sub> (400 MHz; (CD<sub>3</sub>)<sub>2</sub>SO)) 3.90 (3 H, s, 11-H), 7.05 (1 H, d, *J* 8.3 Hz, 6-H), 7.10 (1 H, s, 8-H), 7.71 (1 H, d, *J* 8.6 Hz, 5-H) and 8.8 (1 H, s, 4-

H);  $\delta_{C}$  (100 MHz; (CD<sub>3</sub>)<sub>2</sub>SO)) 56.8 (11-CH<sub>3</sub>), 97.8 (3-quat. C), 101.3 (6-CH), 111.6 (10-quat. C), 114.2 (8-CH), 115.4 (12-CN), 131.6 (5-CH), 153.5 (4-CH), 156.9 (9-quat. C), 157.7 (7-quat. C) and 165.7 (2-quat. C); *m/z* (EI) 201.0425 (M<sup>+•</sup>. C<sub>11</sub>H<sub>7</sub>NO<sub>3</sub> requires 201.0424); *m/z* (%) 201 (100), 173 (50.4), 158 (44.0), 148 (14.2), 130 (10.6), 102 (13.5), 79 (11.3), 70 (17.0) and 57 (14.9).

6-Methoxy-2-oxo-2H-1-benzopyran-3-carbonitrile 179



Compound **179** was prepared from iminolactone **49** (81 mg, 0.40 mmol) by general method 6. Recrystallisation from ethanol provided the title compound **179** (54 mg, 0.29 mmol, 66%) as brown crystals, mp 223-224°C (lit.,<sup>260</sup> 226-228 °C). The spectroscopic data were in agreement with the IR and combustion data reported.<sup>260</sup>

New data:  $\delta_{H}$  (400 MHz; (CD<sub>3</sub>)<sub>2</sub>SO)) 3.82 (3 H, s, 11-H), 7.31 (1 H, d, *J* 3.2 Hz, 5-H), 7.38 (1 H, dd, *J* 9.2 and 3.2 Hz, 7-H), 7.44 (1 H, d, *J* 8.8 Hz, 8-H) and 8.85 (1 H, s, 4-H);  $\delta_{C}$  (100 MHz; (CD<sub>3</sub>)<sub>2</sub>SO)) 56.2 (11-CH<sub>3</sub>), 102.7 (3-quat. C), 111.7 (5-CH), 115.0 (12-CN), 118.2 (10-quat. C), 118.3 (7-CH), 123.5 (8-CH), 148.9 (9-quat. C), 153.4 (4-CH), 156.4 (6-quat. C) and 157.3 (2-quat. C); *m/z* (EI) 201.0426 (M<sup>+•</sup>. C<sub>11</sub>H7NO<sub>3</sub> requires 201.0426); *m/z* (%) 201 (100), 186 (25.2), 158 (35.3), 143 (3.6), 130 (13.7), 102 (12.2), 77 (7.9), 63 (8.6) and 57 (2.9).

6-Methoxy-2-oxo-2H-1-benzopyran-3-carboxamide 180



Compound **180** was prepared from iminolactone **145** (230 mg, 1.06 mmol) by general method 6. Recrystallisation from methanol provided the title compound **180** (224 mg, 1.02 mmol, 96%) as pale yellow crystals, mp 228-230

°C (Found: C, 60.19; H, 3.90; N, 6.19. C11H9NO4 requires C, 60.27; H, 4.11; N, 6.39%);  $v_{max}$  (KBr disc)/cm<sup>-1</sup> 3419m and 3258w (NH<sub>2</sub>), 1716s (CO), 1623m and 1570s (C=C), 1278m (C-O) and 829m (CH);  $\delta_{H}$  (400 MHz; (CD<sub>3</sub>)<sub>2</sub>SO)) 3.83 (3 H, s, 11-H), 7.35 (1 H, dd, *J* 3.0 and 9.1 Hz, 7-H), 7.45 (1 H, d, *J* 9.1 Hz, 8-H), 7.54 (1 H, d, *J* 3.0 Hz, 5-H), 8.01 (2 H, 2 br. s, 13-H) and 8.85 (1 H, s, 4-H);  $\delta_{C}$  (100 MHz; (CD<sub>3</sub>)<sub>2</sub>SO)) 56.2 (11-CH<sub>3</sub>), 112.2 (7-CH), 117.6 (8-CH), 119.2 (3-quat. C), 119.7 (10-quat. C), 122.3 (5-CH), 148.0 (4-CH), 148.9 (9-quat. C), 156.3 (6-quat. C), 160.8 (2-quat. C) and 162.9 (12-quat. C); *m/z* (EI) 219.0532 (M<sup>+•</sup>. C11H9NO4 requires 219.0533); *m/z* (%) 219 (100), 203 (23.1), 176 (12.1), 161 (14.2), 133 (13.5), 119 (6.4), 78 (6.5) and 63 (3.6).

5-Methoxy-2-oxo-2H-1-benzopyran-3-carbonitrile 181



Compound **181** was prepared from iminolactone **146** (200 mg, 1.00 mmol) by general method 6. Recrystallisation from ethanol provided the title compound **181** (177 mg, 0.96 mmol, 96%) as tan crystals, mp 196-197 °C.  $v_{max}$  (KBr disc)/cm<sup>-1</sup> 3060-2849w (CH), 2234m (CN), 1727s (CO), 1604s and 1480s (C=C), 1266s (C-O) and 785m (CH);  $\delta_{H}$  (400 MHz; (CD<sub>3</sub>)<sub>2</sub>SO)) 3.95 (3 H, s, 11-H), 7.02 (2 H, d, *J* 8.4 Hz, 6,8-H), 7.73 (1 H, t, *J* 8.4 Hz, 7-H) and 8.82 (1 H, s, 4-H);  $\delta_{C}$  (100 MHz; (CD<sub>3</sub>)<sub>2</sub>SO)) 57.1 (11-CH<sub>3</sub>), 100.4 (3-quat. C), 107.3 (8-CH), 108.4 (12-CN), 109.0 (7-CH), 115.0 (10-quat. C), 137.2 (6-CH), 148.1 (4-CH), 155.1 (9-quat. C), 157.1 (5-quat. C) and 157.2 (2-quat. C); *m/z* (EI) 201.0426 (M<sup>+•</sup>. C1<sub>1</sub>H7NO<sub>3</sub> requires 201.0427); *m/z* (%) 201 (100), 173 (24.1), 158 (27.7), 130 (14.2), 102 (13.5), 76 (6.4) and 57 (7.1).

5-Methoxy-2-oxo-2H-1-benzopyran-3-carboxamide 182



Compound **182** was prepared from iminolactone **147** (200 mg, 0.92 mmol) by general method 6. Recrystallisation from ethanol provided the title compound **182** (165 mg, 0.75 mmol, 82%) as white crystals, mp 266-267 °C (Found: C, 60.09; H, 4.08; N, 6.21. C11H9NO4 requires C, 60.27; H, 4.11; N, 6.39%); v<sub>max</sub> (KBr disc)/cm<sup>-1</sup> 3406m and 3160m (NH<sub>2</sub>), 1720s (CO), 1610s and 1567s (C=C), 1259m (C-O) and 791s (CH);  $\delta_{\rm H}$  (400 MHz; (CD<sub>3</sub>)<sub>2</sub>SO)) 3.98 (3 H, s, 11-H), 6.70 (1 H, t, *J* 8.4 Hz, 7-H), 7.02 (1 H, d, *J* 8.3 Hz, 8-H), 7.05 (1 H, d, *J* 8.4 Hz, 6-H), 7.97 (2 H, 2 br. s, 13-H) and 8.89 (1 H, s, 4-H);  $\delta_{\rm C}$  (100 MHz; (CD<sub>3</sub>)<sub>2</sub>SO)) 56.9 (11-CH<sub>3</sub>), 106.8 (8-CH), 108.6 (6-CH), 109.1 (3-quat. C), 117.5 (10-quat. C), 135.8 (7-CH), 142.4 (4-CH), 155.3 (9-quat. C), 157.3 (5-quat. C), 160.6 (2-quat. C) and 162.8 (12-quat. C); *m/z* (EI) 219.0532 (M<sup>+•</sup>. C11H9NO4 requires 219.0533); *m/z* (%) 219 (100), 203 (67.6), 201 (18.5), 173 (13.5), 134 (16.4), 105 (6.4), 78 (7.8) and 51 (3.6).

2-Formyl-3-methoxythiophene 190



To a stirred solution of 3-methoxythiophene (0.87 mL, 1.0 g, 8.67 mmol) in tetrahydrofuran (10 mL), at rt, was gradually added *n*-butyllithium (1.6 M in hexanes, 5.47 mL, 8.76 mmol). Following this, the reaction mixture was gently heated at reflux for 3.5 h. On cooling to 0 °C, a solution of *N*, *N*-dimethylformamide (0.70 mL, 656 mg, 8.98 mmol) in tetrahydrofuran (10 mL) was added dropwise and stirring continued at 0 °C for 1 h. The mixture was then warmed to rt and the reaction progress monitored by TLC. After stirring at rt for a further 15 h the reaction was quenched by pouring the mixture into water (50 mL) and the product was extracted with ethyl acetate (3 x 30 mL). The organic extracts were combined, washed with brine (2 x 30 mL), dried over anhydrous magnesium sulfate and concentrated *in vacuo*. Purification by column chromatography, eluant hexane-ethyl acetate (1:1), and subsequent recrystallisation from isopropanol provided the title compound **190** (717 mg, 5.05 mmol, 58%) as yellow needles, mp 80-82 °C (lit., <sup>261</sup> 82 °C).

New data: Found C, 50.73; H, 4.14. C<sub>6</sub>H<sub>6</sub>SO<sub>2</sub> requires C, 50.70; H, 4.23%);  $v_{max}$  (KBr disc)/cm<sup>-1</sup> 3102m and 3089m (CH), 1643vs (CO), 1538s (C=C) and 1215s and 1078s (C-O);  $\delta_{H}$  (400 MHz; CDCl<sub>3</sub>) 3.92 (3 H, s, 7-H),

6.80 (1 H, d, J 5.2 Hz, 4-H), 7.58 (1 H, d, J 5.2 Hz, 5-H) and 9.91 (1 H, s, 6-H);  $\delta_{\rm C}$  (100 MHz; CDCl<sub>3</sub>) 59.2 (7-CH<sub>3</sub>), 116.2 (4-CH), 121.5 (2- quat. C), 135.6 (5-CH), 165.5 (3-quat. C) and 181.7 (6-CH); m/z (EI) 142.0089 (M<sup>+•</sup>. C<sub>6</sub>H<sub>6</sub>SO<sub>2</sub> requires 142.0090); m/z (%) 142 (100), 124 (30.4), 111 (25.7), 96 (22.1), 84 (84.3), 70 (7.1) and 51 (21.4).

2-Cyano-3-[2-(3-methoxythienyl)]propenamide 192



To a stirred solution of compound 190 (150 mg, 1.06 mmol) and 2cyanoacetamide (89 mg, 1.06 mmol) in ethanol (10 mL), at rt, neat piperidine (2 drops) was added. Stirring was maintained at rt for 6 h whereupon dropwise addition of water to the filtrate caused the product to precipitate from the solution. The product was filtered, dried under suction and recrystallised from isopropanol to provide the title compound **192** (158 mg, 0.76 mmol, 72%) as yellow crystals, mp 201-203 °C (Found: C, 51.84; H, 3.79; N, 13.31. C9H8N2O2S requires C, 51.92; H, 3.85; N, 13.46%); v<sub>max</sub> (KBr disc)/cm<sup>-1</sup> 3467s (NH<sub>2</sub>), 3140-3079m (CH<sub>3</sub>), 2198m (CN), 1704s (CO), 1565vs and 1521s (C=C), 1361s and 1313s (CH<sub>3</sub>), 1052m (C-O) and 764m (CH); δ<sub>H</sub> (400 MHz; CDCl<sub>3</sub>) 4.00 (3 H, s, 11-H), 7.23 (1 H, d, J 5.6 Hz, 4-H), 7.63 (2 H, 2 br. s, 10-H), 8.10 (1 H, d, J 5.6 Hz, 5-H) and 8.27 (1 H, s, 6-H); δ<sub>C</sub> (100 MHz; CDCl<sub>3</sub>) 59.7 (11-CH<sub>3</sub>), 97.5 (7-quat. C), 113.5 (8-CN), 116.7 (4-CH), 117.8 (2-quat. C), 135.5 (5-CH), 139.9 (6-CH), 163.3 (3-quat. C) and 164.6 (9-quat. C); m/z (EI) 208.0307 (M+\*. C9H8N2O2S requires 208.0307); m/z (%) 208 (78.0), 177 (100), 175 (11.3), 150 (16.3), 121 (12.1), 109 (16.7), 83 (39.7) and 63 (18.4).

2-Cyano-3-[2-(3-methoxythienyl)]propenonitrile 193



The method of Henrio and co-workers<sup>241</sup> was used. To a stirred solution of compound **190** (250 mg, 1.76 mmol) in ethanol (5 mL), at rt, was added a solution of malononitrile (116 mg, 1.76 mmol) in ethanol (5 mL). The mixture was then heated at reflux for 15.5 h. On cooling to 0 °C the precipitated product was filtered, dried under suction and recrystallised from ethanol to provide the title compound **193** (226 mg, 1.19 mmol, 68%) as yellow needles, mp 142-143 °C (Found: C, 56.67; H, 3.13; N, 14.65. C9H6N<sub>2</sub>OS requires C, 56.84; H, 3.16 ; N, 14.74%); v<sub>max</sub> (KBr disc)/cm<sup>-1</sup> 3100-2945m (CH<sub>3</sub>), 2219s (CN), 1569vs and 1512vs (C=C), 1332s and 1242s (CH<sub>3</sub>) and 1070s (C-O);  $\delta_{\rm H}$  (400 MHz; CDCl<sub>3</sub>) 3.96 (3 H, s, 10-H), 6.83 (1 H, d, *J* 5.6 Hz, 4-H), 7.72 (1 H, dd, *J* 5.6 and 0.8 Hz, 5-H) and 7.95 (1 H, d, *J* 0.8 Hz, 6-H);  $\delta_{\rm C}$  (100 MHz; CDCl<sub>3</sub>) 58.3 (10-CH<sub>3</sub>), 71.6 (7-quat. C), 113.0 (2-quat. C), 113.7 (4-CH), 113.8 and 113.9 (8,9-CN), 136.1 (5-CH), 146.5 (6-CH) and 164.5 (3-quat. C); *m/z* (EI) 190.0200 (M<sup>+•</sup>. C9H6N<sub>2</sub>OS requires 190.0199); *m/z* (%) 190 (100), 175 (43.8), 147 (20.6), 121 (37.0), 94 (15.7), 83 (7.1) and 45 (17.8).

2-Formyl-3-hydroxythiophene 195



To a stirred solution of compound **190** (200 mg, 1.41 mmol) in dichloromethane (10 mL), at -78 °C, was added boron tribromide (1 M in DCM, 4.22 mL, 4.22 mmol). After stirring at -78 °C for 5 h, the reaction mixture was allowed to slowly warm to rt and stirring was continued overnight. The reaction was quenched by pouring the mixture into ice/water (50 mL) and the product was extracted with ethyl acetate (3 x 30 mL). The combined organic extracts were washed with aqueous sodium hydroxide (2 M, 3 x 15 mL) and the

resulting aqueous layers were combined, acidified to *ca.* pH 3 then extracted with ethyl acetate (3 x 50 mL). The organic extracts were dried over anhydrous magnesium sulfate, concentrated *in vacuo*. Purification by column chromatography, eluant hexane-ethyl acetate (1:1), provided the title compound **195** (0.83 mmol, 106 mg, 59%) as a pale pink powder, mp 88-90 °C (lit.,<sup>262</sup> 88-89.5 °C). The spectroscopic data were in agreement with the <sup>1</sup>H NMR and combustion data reported.<sup>262</sup>

New data:  $v_{max}$  (KBr disc)/cm<sup>-1</sup> 3100-2593m (OH, H-bonded), 1613s (CO), 1449s and 1411s (C=C), 1312s (OH) and 1021s (C-O);  $\delta_{C}$  (100 MHz; CDCl<sub>3</sub>) 114.6 (2-quat. C), 118.6 (4-CH), 135.4 (5-CH), 165.3 (3-quat. C) and 184.7 (6-CH); *m/z* (EI) 142.0089 (M<sup>+•</sup>. C<sub>6</sub>H<sub>6</sub>SO<sub>2</sub> requires 142.0090); *m/z* (%) 142 (100), 124 (30.4), 111 (25.7), 96 (22.1), 84 (84.3), 70 (7.1) and 51 (21.4).

# REFERENCES

- 1 American Institute for Cancer Research Diet and Cancer Project, *Food, Nutrition and the Prevention of Cancer: A Global Perspective*, American Institute for Cancer Research, 1997
- Imperial Cancer Research Fund Cancer Statistics Fact sheet, April
   1998
- J. Higginson, C-S. Muir and N. Munoz, *Human Cancer: epidemiology* and environmental causes, Cambridge University Press, UK, 1992, p. xvii
- 4 Ref. 3, p. 90-91
- 5 IARC, IARC Monographs on the evaluation of carcinogenic risks to humans, Suppl. 7. Overall Evaluation of Carcinogenecity: An Updating of IARC Monographs, Volumes 1 to 42, IARC Publication, Lyon 1987.
- 6 Ref. 3, p. 153
- F. F. Becker, *Cancer; A Comprehensive Treatise*, Volume 1, Plenum
   Press, USA, 1975, p. 391
- 8 L. Tomatis, *Cancer: Causes, Occurrence and Control*, IARC Scientific Publications No 100, Oxford University Press, Lyon, 1990, p. 155

9 Ref. 8, p. 156-158

- 10 J. Cushion, *Skin Cancer*, House of Commons research paper 96/84, House of Commons, Great Britain, 1996, p. 14-15
- 11 R. M. MacKie, J. M. Elwood and J. L. M. Hawk, *Ann. R. Coll. Surg. Engl.*, 1987, **21**, 2, 91
- 12 Ref. 10, p. 22-23
- 13 Ref. 8, p. 169-170
- 14 A. Tannenbaum, *Cancer Res.*, 1942, **2**, 468
- 15 I. F. Tannock and R. P. Hill, *The Basic Science of Oncology*, McGraw-Hill Inc., USA,1992, Second Edition, p. 88
- 16 Ref. 15, p. 93
- 17 Ref. 15, p. 98
- 18 G. M. Cooper, *Oncogenes*, Jones and Bartlett Publishers Inc., USA, 1990
- 19 Ref. 15, p. 62-63
- 20 Ref. 18, p. 299
- 21 A. J. Levine, *Tumour Suppressor Genes, the Cell Cycle and Cancer*, volume 12, Cold Spring Harbour Laboratory Press, USA, 1992, p. 6
- 22 Ref. 21, p. 107

- A. Kamb, D. Shattuck-Eidens, R. Ecles, Qiu, N. A. Gruis, W. Ding, C. Hussey, T. Tran, Y. Miki, J. Weaver-Feldhaus, M. McClure, J. F. Aitken, D. E. Anderson, W. Bergman, R. Frants, D. E. Goldgar, A. Green, R. MacLennan, N. G. Martin, L. J. Meyer, P. Youl, J. J. Zone, M. H. Skolnick and L. A. Cannon-Albright, *Nat. Genet.*, 1994, 8, 22
- 24 C. C. Widnell and K. H. Pfenninger, *Essential Cell Biology*, Williams and Williams, USA, 1990, International Edition
- 25 Ref. 24, p. 66
- 26 Ref. 24, p. 58-59
- 27 Ref. 18, p. 3
- 28 S. K. Carter, E. Glastein and R. B. Livingston, *Principles of Cancer Treatment*, McGraw-Hill, USA, 1982, p. 54-57
- 29 C. A. Perez and L. W. Brady, *Principles and Practices of Radiation Oncology*, J.B. Lippincott Company, USA, 1987
- 30 Ref. 29, p. 1239-1243
- 31 H. R. Withers and L. J. Peters, *Innovations in Radiation Oncology*, Spring-Verlag Berlin Heidelberg, Germany, 1998, p. 191
- 32 Ref. 31, p. 202
- 33 Ref. 28, p. 115-135
- 34 W. O. Foye, *Cancer Chemotherapeutic agents*, American Chemical Society, USA, 1995
- 35 R. W. Ruddon, W. D. Ensminger and J. Maybaum, *The Anticancer drugs*, Oxford University Press Inc., USA, 1994, Second Edition
- 36 J. A. Montgomery, *Methods Cancer Res.*, 1979, **16**, 3
- 37 D. J. Black and R. B. Livingston, *Drugs*, 1990, **39**, 489
- 38 Adapted from ref. 35, p. 110
- 39 Ref. 28, p. 117
- 40 Ref. 34, p. 121
- 41 Ref. 34, p. 123
- 42 J. Leiter, M. A. Schneiderman, *Cancer Res.*, 1959, **19**, 31
- 43 Ref. 35, p. 129-130
- 44 Ref. 35, p. 131-132
- 45 Ref. 34, p. 48-49
- F. Calabresi, L. Di Lauro, P. Marolla, C. D. Curcio, G. Paoletti, A.
  Calabro, D. Giannarelli, P. Ballatore, C. M. Foggi, M. Di Palma, R. Stofi and E. Cortesi, *Semin. Oncol.*, 1991, 18, Suppl. 4, 66
- 47 M. Yasumoto, I. Yamawaki, T. Marunaka, S. Hashimoto, *J. Med. Chem.*,
   1978, **21**, 738
- 48 Ref. 34, p. 58-59

- 49 M. S. Lyman and J. H. Burchenal, *Am. J. Nurs.*, 1963, **63**, 82
- 50 E. Boesen and W. Davis, *Cytotoxic Drugs in the Treatment of Cancer*, Edward Arnold Publishers Ltd, London, 1995, p. 120-121
- 51 Ref. 28, p. 118-119
- 52 Ref. 35, p. 183
- 53 Ref. 35, p. 155
- 54 K. M. Tewey, G. L. Chen, E. M. Nelso and L. F. Liu, *J. Biol. Chem.*, 1984,
  259, 14, 9182
- 55 Ref. 34, p. 210-211
- 56 Ref. 34, p. 216
- 57 R. L. Margolis, L. Wilson and S. I. Kiefer, *Nature*, 1978, 272, 450
- 58 Ref. 34, p. 350
- 59 Ref. 28, p. 124
- 60 Ref. 34, p. 352
- 61 P. B. Schiff, J. Frant, S. B. Horwitz, *Nature*, 1979, 277, 665
- 62 M. C. Wani, H. L. Taylor, M. E. Wall, P. Coggon, A. T. McPhail, *J. Am. Chem. Soc.*, 1971, **93**, 2325
- K. C. Nicolaou, Z. Yang, J. J. Liu, H. Veno, P. G. Nantermet, R. K. Guy,
  C. F. Claiborne, J. Renaud, E. A. Couladouros, K. Paulvannan, E. J.
  Sorensen, *Nature*, 1994, **367**, 630
  K. C. Nicolaou and R. K. Guy, *Angew. Chem., Int. Ed. Engl.*, 1995, **34**, 2079
- R. A. Holton, C. Somoza, H-B Kim, F. F. Liang, R. T. Biediger, P. D.
  Boatman, M. Shindo, C. C. Smith, S. Kim, H. Nadizaden, Y. Suzuki, C.
  Tao, P. Vu, S. Tang, P. Zhang, K. K. Murchi, L. N. Gentile and J. H. Lui, J. Am. Chem. Soc., 1994, 116, 1597
- J. J. Masters, J. T. Link, L. B. Snyder, W. B. Young and S. J.Danishefsky, *Angew. Chem., Int. Ed. Engl.*, 1995, **34**, 1723
- 66 J. M. Koeller, Ann. Pharmacol., 1994, 28, 5
- 67 J. W. Yarbo (Editor in chief), Semin. Oncol., 1993, 20, 4, suppl. 3, 40
- 68 G. Currie, *Cancer and the Immune Response*, Edward Arnold Publishers Ltd, 1980, Second Edition
- 69 Ref. 68, p. 1-2
- K. A. Foon, M. I. Bernhardt and R. K. Oldman, *J. Biol. Response Modif.*, 1982, 1, 249
- 71 Ref. 15, p. 349
- K. A. Foon, M. I. Bernhardt and R. K. Oldman, J. Biol. Response Modif., 1982, 1, 227
- 73 K. A. Foon, *Cancer Res.*, 1989, 49, 1621

- 75 R. M. Mackie, *Malignant Melanoma A guide to early diagnosis*, Pillans and Wilson Ltd, Glasgow, 1994, p. 15-21
- 76 Office for National Statistics, personal correspondence 30/6/98
- 77 Cancer Research Campaign fact sheet 4.1:Malignant Melanoma, Cancer Research Campaign, 1995
- 78 Ref. 75, p. 11
- 79 Scottish Cancer Intelligence Unit, *Cancer Registration Statistics* Scotland 1986-1995 extract ICD9 172 (Malignant Melanoma of the Skin), Scottish Cancer Intelligence Unit Publication, 1998
- 80 S. Rosso, R. Mackie, R. Zanetti, *Eur. J. Cancer*, 1994, **30A**, 4, 550
- 81 Ref. 75, p. 28
- 82 Ref. 10, p. 29-30
- D. Ford, J. M. Bliss, A. J. Swerdlow, B. K. Armstrong, S. Franceshi, A. Green, E. A. Holly, T. Mack, R. M. Mackie, A. Osterlind, S. O. Walter, J. Peto and D.F. Easton, *Int. J. Cancer*, 1995, 62, 377
- 84 R. M. Mackie, *Eur. J. Cancer*, 1998, **34**, Suppl. 3, S3
- D. J. Nancarrow, G. J. Mann, E. A. Holland, G. T. Walker, S. C. Beaton,
  M. K. Watters, C. Luxford, J. M. Palmer, J. A. Donald, J. L. Weber, J. W.
  Fountain, R. F. Kefford and N. K. Hayward, *Am. J. Hum. Genet.*, 1993,
  53, 936
- 86 A. G. Kudson, *Nature Genet.*, 1993, **5**, 103
- A. Kamb, D. Shattuck-Eidens, R. Ecles, Qiu, N. A. Gruis, W. Ding, C. Hussey, Tran, Y Miki, J. Weaver-Feldhaus, M. McClure, J. F. Aitken, D. E. Anderson, W Bergman, R. Frants, D. E. Goldgar, A. Green, R. MacLennan, N. G. Martin, L. J. Meyer, P. Youl, J. J. Zone, M. H. Skolnick and L. A. Cannon-Albright, *Nature Genet.*, 1994, 8, 22
- L. A. Cannon-Albright, D. E. Goldgar, L. J. Meyer, C. M. Lewis, D. E.
  Anderson, J.W. Fountain, M. E. Hegi, R. W. Wiseman, E. M. Petty, A. E.
  Bale, O. I. Olopade, M. O. Diaz, D. J. Kwiatkowski, M. W. Piepkorn, J. J.
  Zone and M. H. Skolnick, *Science*, 1992, **258**, 1148
- C. J. Hussussian, J. P. Struewing, A. M. Goldstein, P. A. T. Higgins, D. S.
   Ally, M.D. Sheahan, W. H. Clark Jr., M. A. Tucker and N. C. Dracopoli, *Nature Genet.*, 1994, 8, 15
- 90 A. Breslow, Ann. Surgery, 1970, 172, 902
- 91 R. A. Schwartz, *Skin Cancer: Recognition and Management*, Springer-Verlag New York Inc., USA, 1988, p. 127-131
- 92 U. Veronesi and N. Cascinelli, Arch. Surg., 1991, 126, 4, 438

<sup>74</sup> Ref. 15, p. 247

- U. Ringborg, R. Andersson, J. Eldh, B. Glaumann, L. Hafstrom, S. Jacobson, P. E. Johansson, L. Krysander and B. Lagerlof, *Cancer*, 1996, **77**, 9, 1809
- 94 T. K. Overett and M. H. Shiu, *Cancer*, 1985, 56, 5, 1222
- 95 A. R. Harwood, F. Dancuart, P. J. Fitzpatrick and T. Brown, *Cancer*, 1981, **48** 2599
- 96 S. S. Legh, Semin. Oncol., 1989, 16, 1, suppl. 1, 34
- 97 J. K. Luce, *Cancer*, 1972, **30**, 1604
- 98 Z. Mechl and P. Krejci, Neoplasma, 1983, 30, 371-377,
- 99 S. Retsas, I. Peat, R. Ashford et al, *Cancer Treat. Rev.*, 7, 87-90, 19804
- 100 L. Nathanson, B. K. Wittenberg, Cancer Treat. Rep., 1980, 64, 133-137,
- 101 C. Jacquillat, D. Khayat, P. Banzet, M. Well, P. Fumoleau, M-F Avril, M. Namer, J. Bonneterre, P. Kerbrat, J. J. Bonerandi, R. Bugrat, P. Montcuquet, D. Cupissol, R. Lauvin, C. Vilmer, C. Prache and J. P. Bizzari, *Cancer*, 1990, 66, 1873
- 102 N. E. Constanza, L. Nathanson, D. Schoenfeld, J. Wolter, J. Colsky, W. Regelson, T. Cunningham and N. Sedransk, *Cancer*, 1977, **40**, 1010
- J. H. Mulder, P. Dodion, F. Cavalli, B. Czarnetzki, M. Clavel, D. Thomas,
  S. Suciu and M. Rozencweig, *Eur. J. Cancer Clin. Oncol.*, 1982, 18, 12, 1297
- 104 A. Seeber, M. Binder, A. Steiner, K. Wolff and H. Pehamberger, *Eur. J. Cancer*, 1998, **34**, 13, 2129
- 105 J. M. Kirkwood, *Eur. J. Cancer*, 1998, **34**, Suppl. 3, S5
- 106 J. M. Kirkwood, M. H. Strawderman, M. S. Einstoff, J. J. Smith, E. C. Borden and R. H. Blum, *J. Clin. Oncol.*, 1996, **14**, 4
- P. O. Livingston, G. Rutter, P. Srivastava, M. Padavan, M. J. Calves,
  H. F. Dettegen, L. J. Old, *Cancer Res.*, 1989, 49, 7045
- 108 P. O. Livingston, G. Y. C. Wong, S. Adluri, Y. Tao, M. Padavan, R. Parente, C. Hanlon, M. J. Calves, F. Helling, G. Ritter, H. F. Oettegen, and J. O. Lloyd, *J. Clin. Oncol.*, 1994, **12**, 1036
- 109 G. Parmiani, Eur. J. Cancer, 1998, 34, Suppl. 3, S42
- 110 S. A. Rosenberg, J. Natl. Cancer Inst., 1996, 88, 1635
- 111 Ref. 10, p. 29-30
- 112 Ref. 10, p. 24-27
- 113 C. M. Balch, A. N. Houghton, G. W. Milton, A. J. Sober and S-J Song, *Cutaneous Melanoma*, J.B. Lippincott Company, USA, 1992, Second Edition, p. 83
- 114 P. Rumke, *Therapy of Advanced Melanoma*, Karger, Switzerland, 1990,p. 216-218

- 115 W. H. Clark Jr., L. I. Goldman and M. J. Mastrangelo, *Human Malignant Melanoma*, Grune and Stratton, USA, 1979, p. 3
- 116 Ref. 113, p. 89-90
- 117 Adapted from ref. 114, p. 215 and ref. 115, p. 5
- A. Sanchez-Ferrer, J. N. Rodriguez-Lopez, F. Garcia-Canovas and
   F. Garcia-Carmona, *Biochim. Biophys. Acta.*, 1995, **1247**, 1
- 119 C. J. Cooksey, P. J. Garratt, E. J. Land, S. Pavel, C. A. Ramsden,P. A. Riley and N. P. M. Smit, *J. Biol. Chem.*, 1997, **272**, 26226
- 120 G. S. Johnson and I. Pasten, Nature (New Biol.), 1962, 237, 267
- 121 T. B. Fitzpatrick, W. Becker Jr., A. B. Lerner and H. Montgomery, Science, 1950, **112**, 223
- 122 Ref. 115, p. 7
- 123 Ref. 115, p. 10
- 124 P. A. Riley, *Eur. J. Cancer*, 1991, **27**, 9, 1172.
- 125 M. Picardo, S. Passi, M. Nazzaro-Porro, A. Breathnach, C. Zopmetta, A. Faggioni and P. Riley, *Biochem. Pharmacol.*, 1987, **36**, 417
- 126 S. Miura, K. Jimbow and S. Ito, Int. J. Cancer ,1990, 46, 931
- 127 F. Alena, K. Jimbow and S. Ito, *Cancer Res.*, 1990, **50**, 3743
- 128 S. Miura, T. Veda, K. Jimbow, S. Ito, and K. Fujita, *Arch. Dermatol. Res.*, 1987, **279**, 219
- 129 S. R. Padgette, H. H. Herman, J. Hee Han, S. H. Pollock and S. W. May, *J. Med. Chem.*, 1984, **27**, 10, 1354
- 130 S. Inoue, S. Ito, K. Wakamatsu, K. Jimbow and K. Fujita, *Biochem. Pharmacol*, 1990, **39**, 1077
- 131 G. Proto, M. d'Ischia and D. Mascagna, Melanoma Res., 1994, 4, 351
- 132 Adapted from ref. 131, figure 4, p. 355
- 133 P. McKeown, Ph.D. Thesis, University of Glasgow, 1996
- 134 N. J. Lant, Ph.D. Thesis, University of Glasgow, 1998
- 135 Adapted from ref. 133, p. 119 and ref. 134, p. 49
- 136 P. Yaish, A. Gazit, C. Gilon and A. Levitzki, *Science*, 1988, 242, 933
- 137 T. Hunter, *Cell*, **80**, 1995, 225
- 138 A. M. Edelman, D. K. Blumenthal and E. G. Krebs, *Ann. Rev. Biochem.*, 1987, **56**, 567
- 139 T. Hunter, *Cell*, 1987, **50**, 823
- 140 S. K. Hanks and T. Hunter, *J. FASEB*, 1995, **9**, 576
- 141 Adapted from ref. 133, p. 26
- 142 O. M. Rosen, *Science*, 1987, 237, 1452
- 143 A. Ulrich and J. Schlessinger, *Cell*, 1990, **61**, 203

- 144 T. W. Sturgill and L. B. Ray, *Biochem. Biophys. Res. Commun.*, 1986, 134, 565
- 145 T. W. Sturgill, L. B. Ray, E. Erikson and J. L. Maller, *Nature*, 1988, **334**, 715
- 146 A. J. Rossomando, D. M. Payne, M. J. Weber and T. W. Sturgil, *Proc. Natl. Acad. Sci. USA*, 1988, **86**, 6940
- 147 C. J. Marshall, *Cell*, 1995, **80**, 179
- 148 B. Errede and D. E. Levin, Curr. Opin. Cell Biol., 1993, 5, 254
- 149 E. A. Elion, P. L. Grisafi and G. R. Fink, Cell, 1990, 60, 649
- 150 W. E. Courchesne, S. Kunisawa and J. Thorner, *Cell*, 1989, 58, 1107
- 151 T. G. Boulton, J. S. Gregory and M. H. Cobb, *Biochem.*, 1991 30, 278
- T. G. Boulton, S. H. Nye, D. J. Robbins, N. Y. Ip, E. Radziejewska,
  S. D. Morgenbesser, R. A. DePinho, N. Panayotatos, M. H. Cobb and
  G. D. Yancopoulos, *Cell*, 1991, 65, 663
- 153 J. M. Kyriakis and J. Arruch, J. Biol. Chem, 1990, 265, 17355
- 154 J. R. Woodgette, *Protein Kinases*, Oxford University Press Inc., USA, 1994, p. 112-113
- 155 C. S. Hill and R. Triesman, *Cell*, 1995, **80**, 199
- 156 Adapted from ref. 147
- 157 N. G. Ahn, R. Seger, R. L. Bratlien, C. D. Diltz, N. K. Tonks and E. G. Krebs, *J. Biol. Chem.*, 1991, **266**, 4220
- 158 N. Gomez and P. Cohen, *Nature*, 1991, **353**, 170
- 159 C. M. Crews, A. A. Alessandrini and R. L. Erikson, *Science*, 1992, **258**, 478
- 160 N. G. Anderson, J. L. Maller, N. K. Tonks and T. W. Sturgill, *Nature*, 1990, **343**, 651
- 161 C-F. Zheng and K-L. Guan, *EMBO J.*, 1994, **13**, 1123
- 162 Ref. 154, p. 71
- 163 P. Van der Geer, T. Hunter and R. A. Lindberg, *Annu. Rev. Cell. Biol.*, 1994, **10**, 251
- 164 H. R. Bourne, D. A. Sanders and F. McCormick, *Nature*, 1991, **349**, 117
- 165 R. Treisman, Curr. Opin. Genet. Dev., 1994, 4, 96
- 166 J. M. Kyriakis, P. Banerjee, E. Nikolakaki, T. Dai, E. A. Rubie, M. F. Ahmad, J. Arruch and J. R. Woodgett, *Nature*, 1994, **369**, 156
- 167 R. J. Davis, *Trends Biochem. Sci.*, 1994, **19**, 470
- 168 Ref. 15, p. 74
- 169 C. Shih, L. C. Padhy, M. Murray, R. A. Weinberg, *Nature*, 1981, **290**, 261-264.

S. Pulciani, E. Santos, A. V. Lauver, L. K. Long, K. C. Robbins, M. Barbacid, *Proc. Natl. Aca. Sci. USA*, 1982, **79**, 2845

- 170 J. J. Harvey, Nature, 1964, 204, 1104
- 171 W. H. Kirsten, L. A Mayer, J. Natl. Cancer Inst., 1967, 39, 311
- 172 M. Barbacid, Ann. Rev. Biochem., 1987, 56, 779
- 173 V. P. Stanton Jr. and G. M. Cooper, *Mol. Cell Biol.*, 1987, 7, 1171
- 174 V. P. Stanton Jr., D. W. Nichols, A. P. Laundano and G. M. Cooper, *Mol. Cell Biol.*, 1989, **9**, 639
- 175 D. W. Fry and A. J. Bridges, *Curr. Opin. Biotechnol.*, 1995, 6, 662
  A. Levitzki and A. Gazit, *Science*, 1995, 267, 1782
- 176 W. Harris, C. H. Hill, E. J Lewis, J. S Nixon, S. E. Wilkinson, *Drugs Future*, 1993, **18**, 727
- 177 J. C. Lee, J. L. Adams, *Curr. Opin. Biotech.*, 1995, **6**, 657
- 178 L. Pang, T. Sawada, S. J. Decker, A. R. Saltiel, *J. Biol. Chem.*, 1995, 270, 13585
- 179 R. C. Coombes, CRC Laboratories, Dept. Medical Oncology, Charing Cross Hospital, London.
- 180 G. P. Schiemenz, *Chem. Ber.*, 1962, **95**, 483
- 181 J. A. Cabello, J. M. Campelo, A. Garcia, D. Luna and J. M. Marinas, *J. Org. Chem.*, 1984, **49**, 5195
- 182 W. Baker and C. S. Howes, J. Chem. Soc., 1953, 119
- 183 R. D. H. Murray, J. Méndez and S. A. Brown, *The Natural Coumarins; Occurrence, Chemistry and Biochemistry*, John Wiley and Sons Ltd, UK, 1982
- 184 T. G. Gant and A. I. Meyers, *Tetrahedron*, 1994, 50, 2297
- 185 R. Andreasch, *Monatsh. Chem.*, 1884, **5**, 33
- 186 T. W. Greene and P. G. M. Nuts, *Protective Groups in Organic Synthesis*, Wiley and Sons, New York, 1991, Second Edition, 265-266 and 433-436
- 187 J. A. Frump, Chem. Rev., 1971, 5, 483
- 188 A. I. Meyers and E. Mihelich, Angew. Chem., Int. Ed. Engl., 1976, 15,270
- 189 H. Witte and W. Seeliger, Angew. Chem., Int. Ed. Engl., 1972, 11, 287
- 190 M. J. Fazio, J. Org. Chem., 1984, 49, 4889
- 191 H. L. Wehrmeister, J. Org. Chem., 1963, 2587
- 192 K. Orito, H. Kaga and M. Ito, J. Heterocycl. Chem., 1980, 17, 417
- 193 G. A. Olah, S. C. Norang, B. G. B. Gupta and R. Malhotra, *J. Org. Chem.*, 1979, **44**, 1247
- 194 J. F. W. McOmie, M. L. Watts and D. E. West, *Tetrahedron*, 1968, **24**, 2289

- 195 L. R. Kelland, CRC Centre for Cancer Therapeutics, Charing Cross Hospital, Surrey
- 196 H. W. Duckworth and J. E. Coleman, J. Biol. Chem., 1970, 245, 1613
- 197 S. H. Pomerantz, J. Biol. Chem., 1963, 238, 7, 2351
- 198 K. H. Magosch, *Synthesis*, 1972, **1**, 37
- 199 K. Kormendy, P. Sohar, J. Volford, *Ann. Univ. Budap. Rolando Eotvos Nominatae, Sect. Chim.*, 1962, **4**, 61
- 200 D. H. Williams and I. Fleming, *Spectroscopic methods in organic chemistry*, McGraw-Hill International UK Ltd, 1989, Fourth Edition (revised), p. 20
- 201 B. M. Trost and I. Fleming, *Comprehensive Organic Synthesis*, Oxford, 1991, Pergamon Press, Volume 2, 341-394
- 202 E. Knoevenagel, Chem. Ber., 1894, 27, 2345
- 203 E. Knoevenagel, *Chem. Ber.*, 1896, **29**, 172
- 204 P. S. Rao and R. V. Venkataratnam, Tetrahedron Lett., 1991, 23, 5821
- 205 W. Lehnert, *Tetrahedron Lett.*, 1970, 6, 4723
- 206 T. I. Reddy and R. S. Varma, Tetrahedron Lett., 1997, 38, 1721
- 207 K. R. Kloetstra and H. van Bekkum, *J. Chem. Soc., Chem. Commun.*, 1995, 1005
- 208 F. Texier-Boullet and A. Foucaud, Tetrahedron Lett., 1982, 23, 4927
- 209 S. Chalais, P. Laszlo and A. Mathy, Tetrahedron Lett., 1985, 26, 4453
- 200 G. G. Yakobson and N. E. Akhmetora, Synthesis, 1983, 169
- 211 E. Knoevenagel, Chem. Ber., 1898, 31, 2585
- 212 A. C. O. Hann and A. Lapworth, J. Chem. Soc., 1904, 85, 46
- A. Gazit, P. Yaish, C. Gilon and A. Levitzki, *J. Med. Chem.*, 1989, 32, 2344
- 214 M. Tandon, P. D. Thomas, M. Shokravi, S. Singh, S. Samra, D. Chang and K. Jimbow, *Biochem. Pharmacol.*, 1998, **55**, 2023
- 215 Y. Minamitsuji, S. Sugiyama, S. Singh and K. Jimbow, *J. Invest. Dermatol.*, 1998, **110**, 568
- 216 R. M. MacKie, Department of Dermatology, University of Glsagow
- 217 T. R. Burke, B. Lim, V. E. Marquez, Z-H. Li, J. B. Bolen, I. Stefanova and I. D. Horak, *J. Med. Chem.*, 1993, **36**, 425
- C. N. O'Callaghan and M. L. Conalty, *Proc. R. Ir. Acad., Sect. B*, 1979, 79, 87
- 219 A. Sakurai, Y. Motomura and H. Midorikawa, *J. Org. Chem.*, 1972, **37**, 1523
- 220 D. A. F. Gillespie, Beatson Institute for Cancer Research, Cancer Research Campaign Beatson Laboratories, Glasgow

- 221 G. H. May, K. E. Allen, W. Clark, M. Funk and D. A. F. Gillespie, *J. Biol. Chem.*, 1998, **273**, 33429
- E. Obrador, M. Castro, J. Tamariz, G. Zepeda, R. Miranda and F. Delgado, *Synthetic Commun.*, 1998, **28**, 4649
- 223 H. Cho, T. Iwashita, M. Hamaguchi and Y. Oyama, *Chem. Pharm. Bull.*, 1991, **39**, 3341
- 224 A. Vogel, Gilbert's Ann. Phys., 1820, 64, 161
- 225 W. H. Perkin, J. Chem. Soc., 1868, 21, 53
- 226 Ref. 183, p. 271-274
- 227 Ref. 183, p. 282-288
- 228 R. Raue, H. Harnisch and K. H. Drexhage, Heterocycles, 1984, 21, 167
- 229 R. W. Thomas and N. J. Leonard, Heterocycles, 1976, 5, 839
- 230 T. Harayama, K. Katsuno, H. Nishioki, M. Fujii, Y. Nishita, H. Ishii and Y. Kaneko, *Heterocycles*, 1996, **39**, 613
- 231 S. M. Sehna and N. S. Shah, Chem. Rev., 1945, 36, 1
- 232 R. L. Shrinker, Org. React., 1942, 1, 1
  - G. Brufola, F. Fringuelli, O. Piermatti and F. Pizzo, *Heterocycles*, 1996, **43**, 1257
- 233 F. Bigi, L. Chesini, R. Maggi and G. Sartori, *J. Org. Chem.*, 1999, **64**, 1033
- B. T. Watsona and G. E. Christiansen, *Tetrahedron Lett.*, 1998, 39, 6087
- 235 S. Gronowitz, *The Chemistry of Heterocyclic Compounds*, New York, 1991, John Wiley and Sons, Volume 44, part 4, p 397-502
- 236 E. Arribas and S. Vega, J. Heterocycl. Chem., 1984, 21, 167
- F. J. Gommers, J. Bakker and H. Wynberg, *Photochem. Photobiol.*, 1982, **35**, 615
- 238 S. Gronowitz, Ark. Kemi., 1954, 7, 361
- 239 S. Gronowitz, Ark. Kemi., 1958, 12, 239
- H. J. Jakobsen and S-O Lawesson, Tetrahedron, 1965, 21, 3331
- 241 G. Henrio, J. Morel and P. Pastour, Bull. Soc. Chim. Fr., 1976, 265
- 242 J. Leonard, B. Lygo and G. Procter, *Advanced Practical Organic Chemistry*, Chapman and Hall, Oxford, 1995, Second Edition
- 243 E. P. Papadopoulos and B. George, J. Org. Chem., 1977, 42, 2530
- 244 G. S. Poindexter, J. Heterocycl. Chem., 1983, 20, 1431
- 245 H. Witte and W. Seeliger, *Liebigs Ann. Chem.*, 1974, 996
- K. C. Nicolaou, T. K. Chakraborty, Y. Ogawa, R. A. Daines,
  N. S. Simpkins and G. T. Furst, *J. Am. Chem. Soc.*, 1988, 110, 4660
- 247 J. H. Boyer and J. Hamer, J. Am. Chem. Soc., 1955, 77, 951

- 248 Y. Iwakura, A. Nabeya, T. Nishiguchi and Y. Ichikawa, *J. Org. Chem.*, 1965, **30**, 3410
- 249 V. G. Brunton, M. J. Lear, D. J. Robins, S. Williamson and P. Workman, Anti-Cancer Drug Des., 1994, **9**, 291
- 250 B. B. Corson and R. W. Stroughton, J. Am. Chem. Soc., 1928, 50, 2825
- 251 A. Lapworth and F. H. Wykes, J. Chem. Soc., 1917, 111, 790
- R. H. Curtis, J. N. E. Day and L. G. Kimmins, *J. Chem. Soc.*, 1923, 123, 3131
- 253 G. E. H. Elgemeie and A. H. H. Elghandour, *Bull. Chem. Soc. Jpn.*, 1990, **63**, 1230
- 254 C. N. O'Callaghan, T. B. H. McMurray and J. E. O'Brien, *J. Chem. Soc., Perkin Trans. 2*, 1998, 425
- 255 G. Sunagawa and H. Nakao, *Chem. Pharm. Bull.*, 1965, **13**, 442
- 256 M. A. Weinberger and H. L. Holmes, Can. J. Chem., 1965, 43, 2585
- 257 K. Y. Chu, J. Griffiths and D. Ward, J. Chem. Res., (M), 1981, 10, 3701
- I. Cavaco, J. C. Pessoa, M. T. Duarte, R. D. Gillard and P. Matias,*J. Chem. Soc., Chem. Commun.*, 1996, **11**, 1365
- 259 H. Yasuda and H. Midorikawa, Bull. Chem. Soc. Jpn., 1966, 39, 1754
- 260 D. V. Rao, H. Ulrich, F. A. Suber and A. A. R. Sayigh, *Chem. Ber.*, 1973, 106, 388
- 261 S. Gronowitz and B. Jagersten, Arki. Kemi., 1961, 18, 213
- 262 S. Gronowitz and A. Bugge, Acta. Chem. Scand., 1966, 20, 261

